UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): November 6, 2009

ARMOUR Residential REIT, Inc.

(Exact Name of Registrant as Specified in Its Charter)

	Maryland	001-33/36	20-1908/63				
	(State or Other Jurisdiction of Incorporation)	(Commission File Number)	(I.R.S. Employer Identification No.)				
		l Parkway, Suite 200	22062				
		on, Florida al Executive Offices)	32963 (Zip Code)				
		<u>(772) 388-4758</u>					
	(Registrar	nt's Telephone Number, Including A	rea Code)				
	·	ammock Way, Vero Beach, Florida of Former Address, if Changed Since					
	appropriate box below if the Form 8-K of the following provisions (see Genera		satisfy the filing obligation of the registrar				
_] V	Vritten communications pursuant to Rul	le 425 under the Securities Act (17 C	CFR 230.425)				
_] S	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17CFR 240.14a-12)						
_] P	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))						
_] P	re-commencement communications pu	rsuant to Rule 13e-4(c) under the Ex	change Act (17 CFR 240.13e-4(c))				

Item 1.01. Entry into a Material Definitive Agreement.

Background

On November 5, 2009, the stockholders of Enterprise Acquisition Corp. ("Enterprise") approved certain proposals to: (i) amend Enterprise's amended and restated certificate of incorporation to allow for a business combination with ARMOUR Residential REIT, Inc., a Maryland corporation ("ARMOUR" or the "Company"), and (ii) adopt the Agreement and Plan of Merger, dated as of July 29, 2009 (the "Merger Agreement"), by and among Enterprise, ARMOUR and ARMOUR Merger Sub Corp., a Delaware corporation and a wholly-owned subsidiary of ARMOUR ("Merger Sub Corp."), and approve the merger of Merger Sub Corp. with and into Enterprise. Also on November 5, 2009, in connection with the merger, Enterprise warrantholders approved a proposal to amend the Warrant Agreement, dated as of November 7, 2007, by and between Enterprise and Continental Stock Transfer & Trust Co. (the "Warrant Agreement"). Additionally, in connection with the merger, a management agreement and sub-management agreement were entered into, as discussed further below. On November 6, 2009, Enterprise issued a press release announcing the completion of the merger, a copy of which is attached hereto as Exhibit 99.1 and is incorporated herein by reference.

Merger

On November 6, 2009, Merger Sub Corp. merged with and into Enterprise pursuant to the Merger Agreement. The Merger Agreement provided for two primary transactions: (i) the merger of Merger Sub Corp. with and into Enterprise with Enterprise surviving the merger and becoming a wholly-owned subsidiary of ARMOUR, and (ii) ARMOUR becoming the new publicly-traded corporation of which the holders of Enterprise securities will be security holders. The ARMOUR securities have the same terms as the Enterprise securities for which they were exchanged. However, the rights of stockholders under Maryland law may differ from those under Delaware law. A comparison of the rights of stockholders under Delaware and Maryland law is included in the Proxy Statement/Prospectus under the section titled "Comparison of Rights of Enterprise and ARMOUR," which section is incorporated herein by reference.

Amendment to Warrant Agreement

In connection with the consummation of the transactions contemplated by the Merger Agreement, the Warrant Agreement was amended to (i) increase the exercise price of Enterprise's warrants from \$7.50 to \$11.00 per share and (ii) extend the expiration date of the warrants from November 7, 2011 to November 7, 2013.

Management Agreement

On November 6, 2009, ARMOUR entered into a management agreement with ARMOUR Residential Management LLC, a Delaware limited liability company ("ARRM"). Pursuant to the management agreement, ARRM will provide the day-to-day management of ARMOUR's operations and will perform services and activities relating to ARMOUR's assets and operations in accordance with the terms of the management agreement. The management agreement requires ARRM to manage ARMOUR's business affairs in conformity with certain restrictions contained in the management agreement, including any material operating policies adopted by ARMOUR. The initial term of the management agreement will expire five years after its effective date, November 6, 2009, unless earlier terminated by either ARMOUR or ARRM pursuant to the terms of the management agreement. Following the initial term, the management agreement will automatically renew for successive 1-year renewal terms unless either ARM OUR or ARRM gives advance notice to the other of its intent not to renew the agreement prior to the expiration of the initial term or any renewal term, as applicable, subject to the terms and conditions for, and the restrictions on, the giving of such notice contained in the management agreement. ARRM is entitled to receive from ARMOUR an annual management fee, payable monthly in arrears, equal to the sum of (a) 1.5% of the total of all "gross equity raised" (as defined in the management agreement) up to \$1 billion and (b) 0.75% of the total of all gross equity raised in excess of \$1 billion divided by twelve (12) (which monthly management fee shall never be less than 1/12th of \$900,000). On November 6, 2009, ARMOUR and ARRM entered into an amended and restated management agreement for the sole purpose of reducing the management fee to 1/12th of 1% of the gross equity raised until the gross equity raised is \$50.0 million or greater. ARMOUR is also obligated to reimburse certain expenses incurred by ARRM and its affiliates. ARRM is further entitled to receive a termination fee from ARMOUR under certain circumstances. A copy of the management agreement and the amended and restated management agreement are attached hereto as Exhibits 10.5 and 10.8, respectively, and are incorporated herein by reference.

Sub-Management Agreement

On November 6, 2009, ARRM entered into a sub-management agreement with Staton Bell Blank Check LLC (the "Sub-Manager"), an entity affiliated with Daniel C. Staton and Marc H. Bell. Pursuant to the sub-management agreement, Sub-Manager has agreed to provide certain services to ARRM upon consummation of the merger. In exchange for such services, Sub-Manager will receive a sub-management fee of 25% of the net management fee earned by ARRM under its management agreement with ARMOUR. The sub-management agreement will continue in effect until it is terminated in accordance with its terms. If the sub-management is terminated upon expiration of the initial 5-year term of the management agreement at the election of Sub-Manager, ARMOUR will pay Sub-Manager a final payment equal to 6.16 times the annualized rate of the last three (3) monthly payments to Sub-Manager. The sub-management agreement provides that if, during its term, ARRM or its affiliates manage certa in other investment vehicles, including other REITs, ARRM will negotiate in good faith to provide Sub-Manager the right to enter into a sub-management agreement on substantially the same terms as the sub-management agreement or an alternative arrangement reasonably acceptable to ARRM and Sub-Manager. A copy of the sub-management agreement is attached hereto as Exhibit 10.7 and is incorporated herein by reference.

Item 2.01. Completion of Acquisition or Disposition of Assets.

As described above, on November 6, 2009, Enterprise completed the business combination, as a result of which, Enterprise became a wholly-owned subsidiary of ARMOUR. A description of the merger and the Merger Agreement is included in the Proxy Statement/Prospectus under the section titled "The Merger Agreement", which section is incorporated herein by reference.

At Enterprise's special meeting of stockholders held on November 5, 2009, holders of 11,890,903 shares of common stock of Enterprise elected to convert their shares into a pro rata portion of the trust account established in connection with Enterprise's initial public offering and thereafter maintained by Continental Transfer & Trust Co. (the "Trust Account"). The cash for the payment to the Enterprise stockholders electing conversion was funded with funds released from the Trust Account. The balance of the funds held in the Trust Account were released to ARMOUR at the closing of the business combination and will be used to pay transaction expenses and for working capital.

FORM 10 DISCLOSURES

As disclosed elsewhere in this report, on November 6, 2009, Merger Sub Corp. merged with and into Enterprise, with Enterprise surviving the merger and becoming a wholly-owned subsidiary of ARMOUR. Item 2.01(f) of Form 8-K states that if the registrant was a shell company, as Enterprise was immediately before the business combination disclosed under Item 2.01, then the registrant must disclose the information that would be required if the registrant were filing a general form for registration of securities on Form 10 under the Securities Exchange Act of 1934, as amended. Accordingly, below is the information that would be included in Form 10. Please note that the information provided below relates to the combined company after the business combination.

Further, please note that Item 2.01(f) of Form 8-K states that if any disclosure required by this item is previously reported the registrant may identify the filing in which that disclosure is included instead of including that disclosure in this report. Accordingly, references are made to the final proxy statement filed by Enterprise with the Securities and Exchange Commission on October 14, 2009 (the "Proxy Statement/Prospectus") and the Registration Statement filed by ARMOUR with the SEC on October 13, 2009 (the "Registration Statement"). A copy of the Proxy Statement/Prospectus is included as Exhibit 99.2 and is incorporated herein by reference.

Item 1. Business.

Information about ARMOUR's business is contained in the Proxy Statement/Prospectus under the section titled "Business of ARMOUR," which section is incorporated herein by reference.

Item 1A. Risk Factors.

The risks associated with ARMOUR are contained in the Proxy Statement/Prospectus under the section titled "Risk Factors," which section is incorporated herein by reference.

Item 2. Financial Information.

The Company's selected financial data is contained in the Proxy Statement/Prospectus under the section titled "Selected Summary Consolidated Financial Information," which section is incorporated herein by reference.

In addition, the following provides an update to the Proxy Statement/Prospectus section titled "ARMOUR's Management's Discussion and Analysis of Financial Condition and Results of Operation," which section is incorporated herein by reference.

Quantitative and Qualitative Disclosures About Market Risk

Market risk is the sensitivity of income to changes in interest rates, foreign exchanges, commodity prices, equity prices, and other market-driven rates or prices. ARMOUR is not presently engaged in any substantive commercial business. Accordingly, ARMOUR is not exposed to risks associated with foreign exchange rates, commodity prices, equity prices or other market-driven rates or prices. ARMOUR has not engaged in any hedging activities since its inception.

Item 3. Properties.

The following provides an update to the Proxy Statement/Prospectus under the section titled "Business of ARMOUR – Facilities." The new address of ARMOUR is: 6800 Broken Sound Parkway, Suite 200, Boca Raton, Florida 33487.

Item 4. Security Ownership of Certain Beneficial Owners and Management.

The following table sets forth information, as of the completion of the business combination on November 6, 2009, based on information obtained from the persons named below, with respect to the beneficial ownership of shares of Enterprise by (i) each person known by us to be the owner of more than 5% of outstanding shares of the Enterprise's ordinary shares, (ii) each director and executive officer and (iii) all directors and executive officers as a group. Except as indicated in the footnotes to the table, the persons named in the table have sole voting and investment power with respect to all shares of ordinary shares shown as beneficially owned by them. The total number of shares of common stock and warrants outstanding as of November 6, 2009 is 2,299,054 and 32,489,766, respectively.

	Amount and Nature of			
Name and Address of Beneficial Owner ⁽¹⁾⁽²⁾	Beneficial Ownership	Percent of Class		
Staton Bell Blank Check LLC	7,638,300 (3)	77.9 %		
Daniel C. Staton	3,819,150(3)	39.0 %		
Marc H. Bell	3,819,150(3)	39.0 %		
Stewart J. Paperin	0	0.0 %		
Jordan Zimmerman	0	0.0%		
Scott J. Ulm	0	0.0%		
Jeffrey J. Zimmer	0	0.0%		
Thomas K. Guba	0	0.0 %		
John P. Hollihan, III	0	0.0%		
Robert C. Hain	0	0.0 %		
All Post-Merger directors and executive officers as a group (9 individuals)	3,819,150	39.0 %		

^{*} less than 1%

Item 5. Directors and Executive Officers.

The disclosures set forth in Item 5.02 to this Current Report are incorporated into this item by reference.

⁽¹⁾ Includes shares of common stock which the person has the right to acquire within 60 days of November 10, 2009.

⁽²⁾ Unless otherwise noted, the business address of each of the following is 6800 Broken Sound Parkway, Suite 200, Boca Raton, Florida 33487.

⁽³⁾ Represents 138,300 shares of common stock and 7,500,000 warrants held by Staton Bell Blank Check LLC, 69,150 shares and 3,750,000 warrants which are beneficially owned by Daniel C. Staton and Maria Balodimas Staton, and 69,150 shares and 3,750,000 warrants which are beneficially owned by Marc H. Bell.

Item 6. Executive Compensation.

Information regarding the Company's executive compensation is set forth in the Proxy Statement/Prospectus under the heading "Management of ARMOUR Following the Merger – ARMOUR Executive Compensation," which section is incorporated herein by reference.

Item 7. Certain Relationships and Related Transactions, and Director Independence.

Information regarding the Company's related party transactions and director independence is described in the Proxy Statement/Prospectus under the sections section titled "Certain Relationships and Related Transactions" and "Management of ARMOUR Following the Merger – ARMOUR Directors and Executive Officers," which sections are incorporated herein by reference.

Item 8. Legal Proceedings.

Information regarding the Company's legal proceedings is contained in the Proxy Statement/Prospectus under the section titled "Business of ARMOUR – Legal Proceedings," which section is incorporated herein by reference.

Item 9. Market Price of and Dividends on the Registrant's Common Equity and Related Stockholder Matters.

Enterprise's units, common stock and warrants are listed on the NYSE Amex under the symbols EST.U, EST and EST.WS, respectively. The closing bid price for each share of units, common stock and warrant of Enterprise on July 29, 2009, the last trading day before announcement of the execution of the merger agreement, was \$10.13, \$9.87 and \$0.30, respectively. As of October 5, 2009, the record date for the Enterprise special meeting, the closing bid price for each unit, share of common stock and warrant of Enterprise was \$10.23, \$9.86 and \$0.32, respectively.

The following table sets forth the range of high and low closing bid prices for the units, common stock and warrants for the periods indicated since such units commenced public trading on November 14, 2007 and since such common stock and warrants commenced public trading on December 3, 2007.

		Units			Common Stock			Warrants				
	-	High		Low	-	High		Low	-	High		Low
2007	-											
Fourth Quarter	\$	10.25	\$	9.75	\$	9.15	\$	8.98	\$	0.95	\$	0.78
2008												
First Quarter	\$	9.95	\$	9.53	\$	9.22	\$	9.03	\$	0.92	\$	0.40
Second Quarter	\$	10.00	\$	9.40	\$	9.34	\$	9.10	\$	0.72	\$	0.21
Third Quarter	\$	9.95	\$	9.25	\$	9.54	\$	9.11	\$	0.90	\$	0.05
Fourth Quarter	\$	9.70	\$	8.60	\$	9.20	\$	8.46	\$	0.20	\$	0.01
2009												
First Quarter	\$	9.72	\$	8.95	\$	9.56	\$	9.15	\$	0.24	\$	0.01
Second Quarter	\$	9.90	\$	9.55	\$	9.78	\$	9.56	\$	0.15	\$	0.03
Third Quarter	\$	10.29	\$	9.85	\$	9.97	\$	9.76	\$	0.30	\$	0.04
Fourth Quarter (through November 10, 2009)	\$	9.90	\$	8.94	\$	9.93	\$	8.50	\$	0.38	\$	0.16

Following the business combination, ARMOUR became the successor entity to Enterprise, and Enterprise's common stock and warrants continued to trade on the NYSE Amex, LLC under the symbols ARR and ARR.W, respectively. It is unclear whether ARMOUR will meet the minimum listing requirements for continued listing on the NYSE Amex and as a result, NYSE Amex may delist ARMOUR's securities from quotation on its exchange, which could limit investors' ability to make transactions in ARMOUR's securities.

As of November 10, 2009, there were, of record, two holders of ARMOUR common stock and two holders of ARMOUR warrants.

Prior to the consummation of the merger, Enterprise declared a one-time cash distribution of \$0.13 per share to stockholders of record as of October 5, 2009, excluding Daniel C. Staton, Marc. H. Bell and Maria Balodimas Staton, who held an aggregate of 6,150,000 shares of common stock through Staton Bell Blank Check LLC, and Mr. Ezra Shashoua, Mr. Jordan Zimmerman, Mr. Richard Steiner and Mr. Stewart Paperin, each of whom held 25,000 shares of common stock. On November 6, 2009, in connection with the closing of the merger, Enterprise paid \$0.13 per share to holders of Enterprise common stock who either voted against the merger and elected to exercise their conversion rights or executed stock purchase agreements with Enterprise. This \$0.13 per share cash distribution was included as part of the pro rata portion of the Trust Account paid by Enterprise for each share of common stock owned by converting stockholders and the parties to the stock purchase agreements. ARMOUR intends to pay \$0.13 per share to each current stockholder who held shares of Enterprise common stock on October 5, 2009. Current holders of Enterprise common stock that were holders of record on October 5, 2009 who have not yet received a cash distribution of \$0.13 per share are encouraged to contact their bank or broker for instruction.

Item 10. Recent Sales of Unregistered Securities.

None.

Item 11. Description of Registrant's Securities to Be Registered.

The description of the Company's securities is contained in the Proxy Statement/Prospectus under the section titled "Description of Securities," which section is incorporated herein by reference.

Item 12. Indemnification of Directors and Officers.

A description of the indemnification provisions relating to the Company's officers and directors is contained in Part II, Item 20 of the Registration Statement, which section is incorporated herein by reference.

Item 13. Financial Statements and Supplementary Data

The disclosures set forth in Item 9.01 to this Current Report are incorporated into this item by reference.

Item 14. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure.

None.

Item 15. Financial Statements and Exhibits.

The disclosures set forth in Item 9.01 to this Current Report are incorporated into this item by reference.

Item 3.03. Material Modification to Rights of Security Holders.

In connection with the merger and related business combination, Enterprise adopted the Second Amended and Restated Certificate of Incorporation and the Supplement and Amendment to Warrant Agreement, which are attached hereto as Exhibits 3.2 and 4.4, respectively, and are incorporated herein by reference. The foregoing documents govern Enterprise's common stock and warrants, respectively. The provisions of the Second Amended and Restated Certificate of Incorporation are described in detail in the Proxy Statement/Prospectus under the sections titled "The Initial Charter Proposal" and "The Secondary Proposal," which sections are incorporated herein by reference. The provisions of the Supplement and Amendment to Warrant Agreement are described in detail in the Proxy Statement/Prospectus under the section titled "The Warrant Amendment Proposal," which section is incorporated herein by reference.

Item 5.01. Changes in Control of Registrant.

Reference is made to the disclosure in Item 2.01 of this Current Report on Form 8-K, which are incorporated herein by reference.

Item 5.02. Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangement of Certain Officers.

ARMOUR's board of directors consists of nine directors. Scott J. Ulm and Jeffrey J. Zimmer have remained on the ARMOUR board and Thomas K. Guba, John P. Hollihan, III, Robert C. Hain, Daniel C. Staton, Marc H. Bell, Stewart J. Paperin and Jordan Zimmerman were appointed to serve on ARMOUR's board of directors on November 6, 2009.

Also on November 6, 2009, Mr. Ulm was appointed Co-Chief Executive Officer, Chief Investment Officer, Head of Risk Management and Vice Chairman. Mr. Zimmer was appointed Co-Chief Executive Officer, President and Vice Chairman. Mr. Ezra Shashoua resigned as Chief Financial Officer of Enterprise.

The information set forth in Item 2.01 above under the heading "Certain Relationships and Related Transactions, and Director Independence" is incorporated herein by reference.

Item 5.03. Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

In connection with the merger and business combination, Enterprise adopted the form of Second Amended and Restated Certificate of Incorporation, attached hereto as Exhibit 3.2 and incorporated herein by reference. The provisions of the Second Amended and Restated Certificate of Incorporation are described in detail in the Proxy Statement/Prospectus under the sections titled "The Initial Charter Proposal" and "The Secondary Proposal," which sections are incorporated herein by reference.

Item 5.06. Change in Shell Company Status.

Upon the closing of the business combination, Enterprise ceased to be a shell company, and ARMOUR became the successor entity to Enterprise. The material terms of the transaction pursuant to which Merger Sub Corp. merged with and into Enterprise, with Enterprise surviving as a wholly-owned subsidiary of ARMOUR, are described in the Proxy Statement/Prospectus under the sections titled "The Merger Proposal" and "The Merger Agreement," which sections are incorporated herein by reference.

Item 9.01. Financial Statements and Exhibits.

(a) Financial Statements of businesses acquired

The following financial statements and notes are contained in the Proxy Statement/Prospectus under the section titled "Index to Financial Statements":

ARMOUR Residential REIT, Inc.

Report of Independent Registered Public Accounting Firm

Balance Sheet Dated as of June 30, 2009

Notes to Balance Sheet

Financial statements of ARMOUR Residential REIT, Inc. as of and for the period ended September 30, 2009 are filed herewith beginning on page F-1.

(b) Pro forma financial information

Pro forma financial information reflecting the period ended September 30, 2009 are filed herewith beginning on page PF-1.

(c) Exhibits

Exhibit No.	Description
2.1	Agreement and Plan of Merger, dated as of July 29, 2009, by and among Enterprise Acquisition Corp., ARMOUR Residential REIT, Inc., and ARMOUR Merger Sub Corp. (incorporated by reference to Exhibit 2.1 of ARMOUR Registration Statement on Form S-4 (Reg. No. 333-160870)).
3.1	Amended and Restated Certificate of Incorporation of Enterprise Acquisition Corp. (1)
3.2	Second Amended and Restated Certificate of Incorporation of Enterprise Acquisition Corp.
3.3	By-laws of Enterprise. ⁽¹⁾
3.4	Amendment and Restatement of the Articles of Incorporation of ARMOUR Residential REIT, Inc.
3.5	Amended Bylaws of ARMOUR Residential REIT, Inc.
4.1	Warrant Agreement, dated November 7, 2007, between Continental Stock Transfer & Trust Company and Enterprise Acquisition Corp. (2)
4.2	Specimen Common Stock Certificate of ARMOUR Residential REIT, Inc. (incorporated by reference to Exhibit 4.2 of ARMOUR's Registration Statement on Form S-4 (Reg. No. 333-160870))
4.3	Specimen Warrant Certificate of ARMOUR Residential REIT, Inc. (incorporated by reference to Exhibit 4.3 of ARMOUR's Registration Statement on Form S-4 (Reg. No. 333-160870))
4.4	Amendment to Warrant Agreement, dated November 6, 2009, between Continental Stock Transfer & Trust Company, Enterprise Acquisition Corp. and ARMOUR Residential REIT, Inc.
10.1	Investment Management Trust Agreement, dated November 7, 2007, between Enterprise Acquisition Corp. and Continental Stock Transfer & Trust Company. (1)
10.2	Stock Escrow Agreement, dated November 7, 2007, between Enterprise Acquisition Corp., Continental Stock Transfer & Trust Company and the initial stockholders. ⁽¹⁾
10.3	Stock Escrow Termination Agreement, dated November 6, 2009, between Enterprise Acquisition Corp., Continental Stock Transfer & Trust Company, Staton Bell Blank Check LLC, Stewart J. Paperin, Richard Steiner and Jordan Zimmerman.
10.5	Management Agreement, dated as of November 6, 2009, by and between ARMOUR and ARMOUR Residential Management, LLC
10.6	Sponsor's Voting and Support Agreement ⁽²⁾
10.7	Sub-Management Agreement, dated November 6, 2009, by and between Staton Bell Blank Check LLC and ARMOUR Residential Management, LLC
10.8	Amended and Restated Management Agreement, dated as of November 6, 2009, by and between ARMOUR and ARMOUR Residential Management, LLC

- 99.1 Press Release, dated November 6, 2009
- 99.2 Proxy Statement/Prospectus (incorporated by reference to the Proxy Statement/Prospectus on Form 424(b)(3) filed with the SEC on October 14, 2009).
- (1) Incorporated by reference to Enterprise Acquisition Corp.'s Registration Statement on Form S-1 filed with the SEC on August 6, 2007, as amended.
- (2) Incorporated by reference to Enterprise Acquisition Corp.'s Current Report on Form 8-K filed with the SEC on July 29, 2009.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated: November 12, 2009

ARMOUR RESIDENTIAL REIT, INC.

By: <u>/s/ Jeffrey J. Zimmer</u> Name: Jeffrey J. Zimmer

Title: Co-Chief Executive Officer, President and Vice Chairman

ARMOUR Residential REIT, Inc. Balance Sheet September 30, 2009

ASSETS

\$ 500
500
\$ 500
\$ -
-
0
500
500
\$ 500
s

ARMOUR Residential REIT, Inc. Unaudited Condensed Combined Pro Forma Financial Statements For the Nine Months Ended September 30, 2009

	Enterprise Acquisition Corp.	ARMOUR Residential REIT, Inc.	Combined Pro Forma
ASSETS			
Current Assets:			
Cash	113,699	500	20,881,366
Cash held in trust available for operations	20,084		-
Prepaid Expenses	20,176		20,176
Refundable federal and state income tax	415,525		415,525
Total current assets	569,484	500	21,317,067
Cash held in trust (restricted)	249,469,771		-
TOTAL ASSETS	\$ 250,039,255	\$ 500	\$ 21,317,067
LIABILITIES & STOCKHOLDERS' EQUITY Current Liabilities:			
Accounts payable	1,634,427	-	-
Accrued expenses	150,649		-
Franchise Tax payable	25,138		-
Deferred underwriters' fee	7,496,559		
Total current liabilities	9,306,773	-	-
Common Stock subject to possible redemption	74,249,990		-
Interest Income on CS subject to redemption	643,903		-
Stockholders' Equity:			
Preferred Stock	-		-
Common Stock - Enterprise	3,125		-
Common Stock - ARMOUR		0	2,299
Additional paid-in capital	165,904,776	500	23,344,659
Earnings (deficit) accumulated during development stage	(69,312)		(2,029,891)
Total Stockholders' Equity	165,838,589	500	21,317,067
TOTAL LIABILITIES & STOCKHOLDERS' EQUITY	\$ 250,039,255	\$ 500	\$ 21,317,067

Note to Unaudited Pro Forma Condensed Combined Balance Sheet

Refer to the Proxy Statement/Prospectus under "Unaudited Pro Forma Condensed Combined Financial Information" beginning on page 14 for details of the pro forma adjustments made to the historical activity of Enterprise Acquisition Corp. and ARMOUR Residential REIT, Inc. The above is presented in summary form to reflect the actual results of the Enterprise Acquisition Corp. stockholder vote and actual closing costs incurred.

SECOND AMENDED AND RESTATED CERTIFICATE OF INCORPORATION

OF

ENTERPRISE ACQUISITION CORP.

Pursuant to Section 245 of the Delaware General Corporation Law

ENTERPRISE ACQUISITION CORP., a corporation existing under the laws of the State of Delaware (the 'Corporation"), by its Chief Executive Officer, hereby certifies as follows:

- 1. The name of the Corporation is "Enterprise Acquisition Corp."
- 2. The Corporation's Certificate of Incorporation was filed in the office of the Secretary of State of the State of Delaware on July 9, 2007.
- 3. An Amended and Restated Certificate of Incorporation was filed in the office of the Secretary of State of Delaware on November 13, 2007.
- 4. This Second Amended and Restated Certificate of Incorporation restates, integrates and amends the Amended and Restated Certificate of Incorporation of the Corporation.
- 5. This Second Amended and Restated Certificate of Incorporation was duly adopted by joint written consent of the directors and stockholders of the Corporation in accordance with the applicable provisions of Sections 242 and 245 of the General Corporation Law of the State of Delaware ("DGCL").
- 6. The text of the Amended and Restated Certificate of Incorporation of the Corporation is hereby amended and restated to read in full as follows:
- FIRST: The name of the corporation is Enterprise Acquisition Corp. (hereinafter sometimes referred to as the "Corporation").
- SECOND: The registered office of the Corporation is to be located at the Corporation Trust Company, 1209 Orange Street, in the City of Wilmington, County of New Castle, zip code 19801. The name of its registered agent at that address is the Corporation Trust Company.
- THIRD: The purpose of the Corporation shall be to engage in any lawful act or activity for which corporations may be organized under the DGCL.
- FOURTH: The total number of shares of all classes of capital stock which the Corporation shall have authority to issue is 101,000,000 of which 100,000,000 shares shall be Common Stock of the par value of \$.0001 per share and 1,000,000 shares shall be Preferred Stock of the par value of \$.0001 per share.
- A. <u>Preferred Stock.</u> The Board of Directors is expressly granted authority to issue shares of the Preferred Stock, in one or more series, and to fix for each such series such voting powers, full or limited, and such designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions thereof as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issue of such series (a "<u>Preferred Stock Designation</u>") and as may be permitted by the DGCL. The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the voting power of all of the then outstanding shares of the capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, without a separate vote of the holders of the Preferred Stock, or any series thereof, unless a vote of any such holders is required pursuant to any Preferred Stock Designation.
- B. <u>Common Stock</u>. Except as otherwise required by law or as otherwise provided in any Preferred Stock Designation, the holders of the Common Stock shall exclusively possess all voting power and each share of Common Stock shall have one vote.

FIFTH: The name and mailing address of the sole incorporator of the Corporation are as follows:

Name Address

Mark Y. Liu Akerman Senterfitt
One SE Third Avenue
Suite 2800

Miami, Florida 33131

SIXTH: The Corporation's existence shall terminate on November 7, 2009 (the "Termination Date"). This provision may only be amended in connection with, and become effective upon, the consummation of a Business Combination (defined below). A proposal to so amend this section shall be submitted to stockholders in connection with any proposed Business Combination pursuant to Article Seventh (A) below.

SEVENTH: The following provisions (A) through (H) shall apply during the period commencing upon the filing of this Certificate of Incorporation and terminating upon the consummation of any "Business Combination." A "Business Combination" shall mean the acquisition by the Corporation or its stockholders, whether by merger, capital stock exchange, asset, stock purchase, reorganization or other similar business combination, of one or more entities or assets ("Target Businesse") and resulting in ownership by the Corporation or its stockholders of more than 50% of the voting securities of the Target Business or Businesses. Any acquisition of multiple Target Businesses shall occur simultaneously.

"IPO Shares" shall mean the shares of Common Stock issued in the IPO.

The "<u>Trust Fund</u>" shall mean the trust account established by the Corporation in connection with the consummation of its IPO and into which the Corporation will deposit a designated portion of the net proceeds from the IPO, including any amount that is or will be due and payable as deferred underwriting discounts and commissions (the "<u>Deferred Underwriting Compensation</u>") pursuant to the terms and conditions of the underwriting agreement (the "<u>Underwriting Agreement</u>") to be entered into with the underwriters of the IPO.

- A. Prior to the consummation of any Business Combination, the Corporation shall submit such Business Combination to its stockholders for approval regardless of whether the Business Combination is of a type which normally would require such stockholder approval under the DGCL. In the event that a majority of the IPO Shares present and entitled to vote at the meeting to approve the Business Combination are voted for the approval of such Business Combination, the Corporation shall be authorized to consummate the Business Combination; provided that the Corporation shall not consummate any Business Combination if the holders of 50% or more of the IPO Shares vote against the Business Combination and exercise their conversion rights described in paragraph C below.
- B. Upon consummation of the IPO, the Corporation delivered, or caused to be delivered, for deposit into the Trust Fund \$247,575,000 comprising (i) \$240,075,000 of the net proceeds of the IPO, including \$8,375,000 of deferred underwriting discounts and commissions and (ii) \$7,500,000 of the proceeds from the Corporation's sale of 7,500,000 warrants to its founding stockholder, Staton Bell Blank Check LLC.
- C. In the event that a Business Combination is approved in accordance with the above paragraph (A) and is consummated by the Corporation, any stockholder of the Corporation holding IPO Shares who voted against the Business Combination may, contemporaneously with such vote, demand that the Corporation convert its IPO Shares into cash. If so demanded, the Corporation shall, promptly after consummation of the Business Combination, convert such shares into cash at a per share conversion price equal to the quotient determined by dividing (i) the amount in the Trust Fund, inclusive of any interest thereon, calculated as of two business days prior to the consummation of the Business Combination, by (ii) the total number of IPO Shares.
- D. In the event that the Corporation does not consummate a Business Combination by the Termination Date, the officers of the Corporation shall take all such action necessary to dissolve and liquidate the Corporation as soon as reasonably practicable. In the event that the Corporation is so dissolved and liquidated, only the holders of IPO Shares shall be entitled to receive liquidating distributions and the Corporation shall pay no liquidating distributions with respect to any other shares of capital stock of the Corporation.
- E. A holder of IPO Shares shall be entitled to receive distributions from the Trust Fund only in the event of a liquidation of the Corporation or in the event such holder demands conversion of its shares in accordance with paragraph C above. In no other circumstances shall a holder of shares of Common Stock have any right or interest of any kind in or to the Trust Fund.

- F. Unless and until the Corporation has consummated a Business Combination as permitted under this Article Seventh, the Corporation may not consummate any other business combination, whether by merger, capital stock exchange, asset acquisition, stock purchase, reorganization or other similar business combination or transaction or otherwise. The Corporation will not enter into a Business Combination with a Target Business that is affiliated with any of the Corporation's officers, directors or stockholders, unless the Corporation obtains an opinion from an unaffiliated, independent investment banking firm that the Business Combination is fair to the Corporation's unaffiliated stockholders from a financial point of view.
- G. The Corporation shall not, and no employee of the Corporation shall, disburse or cause to be disbursed any of the proceeds held in the Trust Fund except (i) for the release of interest income to cover any tax obligation owed by the Corporation, (ii) for the release of interest income of up to \$2,450,000 to the Corporation to cover expenses related to investigating and selecting a Target Business and the Corporation's other working capital requirements, (iii) in connection with a Business Combination or thereafter, including the payment of any Deferred Underwriting Compensation in accordance with the terms of the Underwriting Agreement, (iv) upon the Corporation's liquidation or (v) as otherwise set forth herein.
- H. In no event will any of the Corporation's officers, directors, stockholders or special advisors, or any entity with which they are affiliated, be paid, from the Corporation or a Target Business, any finder's fee, consulting fee or other compensation prior to, or for any services they render in order to effectuate, the consummation of a Business Combination (regardless of the type of transaction that it is); provided that the Corporation's officers and directors and its and their affiliates shall be entitled to reimbursement of out-of-pocket expenses incurred by them in connection with certain activities on the Corporation's behalf, such as identifying and investigating possible Target Businesses and Business Combinations. Payments of an aggregate of \$7,500 per month for office space and related services to Bell & Staton, Inc. shall not be subject to the provisions of this paragraph H.

EIGHTH: The following provisions are inserted for the management of the business and for the conduct of the affairs of the Corporation, and for further definition, limitation and regulation of the powers of the Corporation and of its directors and stockholders:

- A. Election of directors need not be by ballot unless the by-laws of the Corporation so provide.
- B. The Board of Directors shall have the power, without the assent or vote of the stockholders, to make, alter, amend, change, add to or repeal the by-laws of the Corporation as provided in the by-laws of the Corporation.
- C. The directors in their discretion may submit any contract or act for approval or ratification at any annual meeting of the stockholders or at any meeting of the stockholders called for the purpose of considering any such act or contract, and any contract or act that shall be approved or be ratified by the vote of the holders of a majority of the stock of the Corporation which is represented in person or by proxy at such meeting and entitled to vote thereat (provided that a lawful quorum of stockholders be there represented in person or by proxy) shall be as valid and binding upon the Corporation and upon all the stockholders as though it had been approved or ratified by every stockholder of the Corporation, whether or not the contract or act would otherwise be open to legal attack because of directors' interests, or for any other reason.
- D. In addition to the powers and authorities hereinbefore or by statute expressly conferred upon them, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation; subject, nevertheless, to the provisions of the statutes of Delaware, of this Certificate of Incorporation, and to any bylaws from time to time made by the stockholders; provided, however, that no by-law so made shall invalidate any prior act of the directors which would have been valid if such by-law had not been made.
- E. The Board of Directors shall be divided into three classes: Class A, Class B and Class C. The number of directors in each class shall be as nearly equal as possible. At the first election of directors by the incorporator, the incorporator shall elect a Class C director for a term expiring at the Corporation's third Annual Meeting of Stockholders. The Class C director shall then appoint additional Class A, Class B and Class C directors, as necessary. The directors in Class A shall be elected for a term expiring at the first Annual Meeting of Stockholders, the directors in Class B shall be elected for a term expiring at the second Annual Meeting of Stockholders and the directors in Class C shall be elected for a term expiring at the third Annual Meeting of Stockholders.

 Commencing at the first Annual Meeting of Stockholders, and at each annual meeting thereafter, directors elected to succeed those directors whose terms expire shall be elected for a term of office to expire at the third succeeding annual meeting of stockholders after their election. Except as the DGCL may otherwise require, in the interim between annual meetings of stockholders or special meetings of stockholders called for the election of directors and/or the removal of one or more directors and the filling of any vacancy in that connection, newly created directorships and any vacancies in the Board of Directors, including unfilled vacancies resulting from the removal of directors for cause, may

be filled by the vote of a majority of the remaining directors then in office, although less than a quorum (as defined in the Corporation's by-laws), or by the sole remaining director. All directors shall hold office until the expiration of their respective terms of office and until their successors shall have been elected and qualified. A director elected to fill a vacancy resulting from the death, resignation or removal of a director shall serve for the remainder of the full term of the director whose death, resignation or removal shall have created such vacancy and until his successor shall have been elected and qualified.

- NINTH: A. A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of the DGCL, or (iv) for any transaction from which the director derived an improper personal benefit. If the DGCL is amended to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the DGCL, as so amended. Any repeal or modification of this paragraph A by the stockholders of the Corporation shall not adversely affect any right or protection of a director of the Corporation with respect to events occurring prior to the time of such repeal or modification.
- B. The Corporation, to the full extent permitted by Section 145 of the DGCL, as amended from time to time, shall indemnify all persons whom it may indemnify pursuant thereto. Expenses (including attorneys' fees) incurred by an officer or director in defending any civil, criminal, administrative, or investigative action, suit or proceeding for which such officer or director may be entitled to indemnification hereunder shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation as authorized hereby.

TENTH: Whenever a compromise or arrangement is proposed between this Corporation and its creditors or any class of them and/or between this Corporation and its stockholders or any class of them, any court of equitable jurisdiction within the State of Delaware may, on the application in a summary way of this Corporation or of any creditor or stockholder thereof or on the application of any receiver or receivers appointed for this Corporation under Section 291 of Title 8 of the Delaware Code or on the application of trustees in dissolution or of any receiver or receivers appointed for this Corporation under Section 279 of Title 8 of the Delaware Code order a meeting of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may be, to be summoned in such manner as the said court directs. If a majority in number representing three fourths in value of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may be, agree to any compromise or arrangement and to any reorganization of this Corporation as a consequence of such compromise or arrangement, the said compromise or arrangement and the said reorganization shall, if sanctioned by the court to which the said application has been made, be binding on all the creditors or class of creditors, and/or on all the stockholders or class of stockholders, of this Corporation, as the case may be, and also on this Corporation. This Article Tenth is subject to the requirements set forth in Article Seventh, and any conflict arising in respect of the terms set forth hereunder and thereunder shall be resolved by reference to the terms set forth in Article Seventh.

ELEVENTH: The Corporation hereby elects not to be governed by Section 203 of the DGCL.

IN WITNESS WHEREOF, the Corporation has caused this Second Amended and Restated Certificate of Incorporation to be signed by Daniel C. Staton, its Chief Executive Officer, as of the 5th day of November, 2009.

/s/ Daniel C. Staton
Daniel C. Staton, Chief Executive Officer

ARMOUR RESIDENTIAL REIT, INC.

ARTICLES OF AMENDMENT AND RESTATEMENT

FIRST: ARMOUR Residential REIT, Inc., a Maryland corporation (the "Corporation"), desires to amend and restate its charter as currently in effect and as hereinafter amended (such amended and restated charter, the "Charter").

SECOND: The following provisions are all the provisions of the charter currently in effect and as hereinafter amended:

ARTICLE I

INCORPORATION

The Corporation was formed under the general laws of the State of Maryland on February 5, 2008.

ARTICLE II

NAME

The name of the corporation is ARMOUR Residential REIT, Inc.

ARTICLE III

PURPOSE

The purposes for which the Corporation is formed are to engage in any lawful act or activity (including, without limitation or obligation, engaging in business as a real estate investment trust under the Internal Revenue Code of 1986, as amended, or any successor statute (the "Code")) for which corporations may be organized under the general laws of the State of Maryland as now or hereafter in force. For purposes of this Charter, "REIT" means a real estate investment trust under Sections 856 through 860 of the Code.

ARTICLE IV

PRINCIPAL OFFICE IN STATE AND RESIDENT AGENT

The address of the principal office of the Corporation in the State of Maryland is c/o CSC-Lawyers Incorporating Service Company, 7 St. Paul Street, Suite 1660, Baltimore, Maryland 21202. The name and address of the resident agent of the Corporation in the State of Maryland are CSC-Lawyers Incorporating Service Company, 7 St. Paul Street, Suite 1660, Baltimore, Maryland 21202. The resident agent is a Maryland corporation.

ARTICLE V

PROVISIONS FOR DEFINING, LIMITING AND REGULATING CERTAIN POWERS OF THE CORPORATION AND OF THE STOCKHOLDERS AND DIRECTORS

SECTION 5.1 <u>Number of Directors.</u> The business and affairs of the Corporation shall be managed under the direction of the Board of Directors. The number of directors of the Corporation initially shall be nine (which number shall be composed of four affiliated directors and five unaffiliated directors), which number may be increased or decreased only by the Board of Directors pursuant to the bylaws of the Corporation (the "Bylaws"), but shall never be less than the minimum number required by the Maryland General Corporation Law (the "MGCL") and the names of the directors who have been duly elected and qualified and who are currently in office are:

Scott J. Ulm and Jeffrey J. Zimmer.

The directors may increase the number of directors and may fill any vacancy, whether resulting from an increase in the number of directors or otherwise, on the Board of Directors occurring before the first annual meeting of stockholders in the manner provided in the Bylaws.

The Corporation elects, at such time as it becomes eligible to make the election provided for under Section 3-802(b) of the Maryland General Corporation Law, that, except as may be provided by the Board of Directors in setting the terms of any class or series of stock, any and all vacancies on the Board of Directors may be filled only by the affirmative vote of a majority of the remaining directors in office, even if the remaining directors do not constitute a quorum, and any director elected to fill a vacancy shall serve for the remainder of the full term of the directorship in which such vacancy occurred.

- SECTION 5.2 Extraordinary Actions. Except for amendments to Article VII and except as specifically provided in Section 5.8 (relating to removal of directors) and in the last sentence of Article VIII, notwithstanding any provision of law permitting or requiring any action to be taken or approved by the affirmative vote of the holders of shares entitled to cast a greater number of votes, any such action shall be effective and valid if declared advisable by the Board of Directors and taken or approved by the affirmative vote of holders of shares entitled to cast a majority of all the votes entitled to be cast on the matter.
- SECTION 5.3 <u>Authorization by Board of Stock Issuance</u>. The Board of Directors may authorize the issuance from time to time of shares of stock of the Corporation of any class or series, whether now or hereafter authorized, or securities or rights convertible into shares of its stock of any class or series, whether now or hereafter authorized, for such consideration as the Board of Directors may deem advisable (or without consideration in the case of a stock split or stock dividend), subject to such restrictions or limitations, if any, as may be set forth herein or in the Bylaws.
- SECTION 5.4 Preemptive and Appraisal Rights. Except as may be provided by the Board of Directors in setting the terms of classified or reclassified shares of stock pursuant to Section 6.4 or as may otherwise be provided by a contract approved by the Board of Directors, no holder of shares of stock of the Corporation shall, as such holder, have any preemptive right to purchase or subscribe for any additional shares of stock of the Corporation or any other security of the Corporation which it may issue or sell. Holders of shares of stock shall not be entitled to exercise any rights of an objecting stockholder provided for under Title 3, Subtitle 2 of the MGCL or any successor statute unless the Board of Directors, upon the affirmative vote of a majority of the Board of Directors, shall determine that such rights apply, with respect to all or any classes or series of stock, to one or more transactions occurring after the date of such det ermination in connection with which holders of such shares would otherwise be entitled to exercise such rights.
- SECTION 5.5 Indemnification. The Corporation shall have the power, to the maximum extent permitted by Maryland law in effect from time to time, to obligate itself to indemnify, and to pay or reimburse reasonable expenses in advance of final disposition of a proceeding to, (a) any individual who is a present or former director or officer of the Corporation or (b) any individual who, while a director or officer of the Corporation and at the request of the Corporation, serves or has served as a director, officer, partner or trustee of another corporation, real estate investment trust, partnership, joint venture, trust, employee benefit plan or any other enterprise from and against any claim or liability to which such person may become subject or which such person may incur by reason of his or her service in such capacity. The Corporation shall have the power, with the approval of the Board of Directors, to provide such indemnification and advancement of expenses to a person who served a predecessor of the Corporation in any of the capacities described in (a) or (b) above and to any employee or agent of the Corporation or a predecessor of the Corporation.
- SECTION 5.6 Determinations by Board. The determination as to any of the following matters, made in good faith by or pursuant to the direction of the Board of Directors consistent with the Charter, shall be final and conclusive and shall be binding upon the Corporation and every holder of shares of its stock: the amount of the net income of the Corporation for any period and the amount of assets at any time legally available for the payment of dividends, redemption of its stock or the payment of other distributions on its stock; the amount of paid-in surplus, net assets, other surplus, annual or other cash flow, funds from operations, net profit, net assets in excess of capital, undivided profits or excess of profits over losses on sales of assets; the amount, purpose, time of creation, increase or decrease, alteration or cancellation of any reserves or charges and the propriety thereof (whether or not any obligation or liability for which s uch reserves or charges shall have been created shall have been paid or discharged); any interpretation of the terms, preferences, conversion or other rights, voting powers or rights, restrictions, limitations as to dividends or distributions, qualifications or terms or conditions of redemption of any class or series of stock of the Corporation; the fair value, or any sale, bid or asked price to be applied in determining the fair value, of any asset owned or held by the Corporation or of any shares of stock of the Corporation; the number of shares of stock of any class of the Corporation; any matter relating to the acquisition, holding and disposition of any assets by the Corporation; or any other matter relating to the business and affairs of the Corporation or required or permitted by applicable law, the Charter or Bylaws or otherwise to be determined by the Board of Directors.
- SECTION 5.7 REIT Qualification. If the Corporation elects to qualify for U.S. federal income tax treatment as a REIT, the Board of Directors shall use its reasonable best efforts to take such actions as are necessary or appropriate to preserve the qualification of the Corporation as a REIT; however, if the Board of Directors determines that it is no longer in the best interests of the Corporation to continue to be qualified as a REIT, the Board of Directors may revoke or otherwise terminate the Corporation's REIT election pursuant to Section 856(g) of the Code. The Board of Directors also may determine that compliance with any restriction or limitation on stock ownership and transfers set forth in Article VII is no longer required for REIT qualification.

SECTION 5.8 Removal of Directors. Subject to the rights of holders of one or more classes or series of Preferred Stock (as hereinafter defined) to elect or remove one or more directors, any director, or the entire Board of Directors, may be removed from office at any time, but only by the affirmative vote of holders of shares entitled to cast at least two-thirds of all the votes entitled to be cast generally in the election of directors.

SECTION 5.9 Advisor Agreements. Subject to such approval of stockholders and other conditions, if any, as may be required by any applicable statute, rule or regulation, the Board of Directors may authorize the execution and performance by the Corporation of one or more agreements with any person, corporation, association, company, trust, partnership (limited or general) or other organization whereby, subject to the supervision and control of the Board of Directors, any such other person, corporation, association, company, trust, partnership (limited or general) or other organization shall render or make available to the Corporation managerial, investment, advisory and/or related services, office space and other services and facilities (including, if deemed advisable by the Board of Directors, the management or supervision of the investments of the Corporation) upon such terms and conditions as may be provided in such agreement or agreements (including, if deemed fair and equitable by the Board of Directors, the compensation payable thereunder by the Corporation).

SECTION 5.10 Investment Asset Class Restriction. The Corporation will be restricted to investing, on a leveraged basis, primarily in hybrid adjustable-rate, adjustable-rate and fixed-rate residential mortgage-backed securities issued or guaranteed by a U.S. Government-chartered entity, such as the Federal National Mortgage Association (more commonly known as Fannie Mae) and the Federal Home Loan Mortgage Corporation (more commonly known as Freddie Mac), or guaranteed by the Government National Mortgage Administration, a U.S. Government corporation (more commonly known as Ginnie Mae) (collectively, "Agency Securities"). A portion of the Corporation's portfolio may be invested in unsecured notes and bonds issued by U.S. Government-chartered entities (collectively, "Agency Debt"), U.S. Treasuries and money market instruments, or accounts at state or federal chartered financial institutions, subject to certain income tests the Corporation must satisfy for its qualification as a REIT. The Corporation may also invest in hedging and other derivative instruments related to the foregoing investments. In the case of any ambiguity in the application of the foregoing restrictions, the Board of Directors of the Corporation will determine such application.

ARTICLE VI

STOCK

SECTION 6.1 Authorized Shares. The Corporation has authority to issue 275,000,000 shares of stock, consisting of 250,000,000 shares of Common Stock, \$0.001 par value per share ("Common Stock"), and 25,000,000 shares of Preferred Stock, \$0.001 par value per share ("Preferred Stock"). The aggregate par value of all authorized shares of stock having par value is \$275,000. If shares of one class of stock are classified or reclassified into shares of another class of stock pursuant to Section 6.2, 6.3 or 6.4 of this Article VI, the number of authorized shares of the former class shall be automatically decreased and the number of shares of the latter class shall be automatically increased, in each case by the number of shares so classified or reclassified, so that the aggregate number of shares of stock of all classes that the Corporation has authority to issue shall not be more than the total number of shares of s tock set forth in the first sentence of this paragraph. The Board of Directors, with the approval of a majority of the entire Board and without any action by the stockholders of the Corporation, may amend the Charter from time to time to increase or decrease the aggregate number of shares of stock or the number of shares of stock of any class or series that the Corporation has authority to issue.

SECTION 6.2 <u>Common Stock</u>. Subject to the provisions of Article VII and except as may otherwise be specified in the terms of any class or series of Common Stock, each share of Common Stock shall entitle the holder thereof to one vote. The Board of Directors may reclassify any unissued shares of Common Stock from time to time in one or more classes or series of stock.

SECTION 6.3 <u>Preferred Stock.</u> The Board of Directors may classify any unissued shares of Preferred Stock and reclassify any previously classified but unissued shares of Preferred Stock of any series from time to time, in one or more classes or series of stock.

SECTION 6.4 Classified or Reclassified Shares. Prior to issuance of classified or reclassified shares of any class or series, the Board of Directors by resolution shall: (a) designate that class or series to distinguish it from all other classes and series of stock of the Corporation; (b) specify the number of shares to be included in the class or series; (c) set or change, subject to the provisions of Article VII and subject to the express terms of any class or series of stock of the Corporation outstanding at the time, the preferences, conversion or other rights, voting powers, restrictions, limitations as to dividends or other distributions, qualifications and terms and conditions of redemption for each class or series; and (d) cause the Corporation to file articles supplementary with the State Department of Assessments and Taxation of Maryland ("SDAT"). Any of the terms of any class or series of stock set or changed purs uant to clause (c) of this Section 6.4 may be made dependent upon facts or events ascertainable outside the Charter (including determinations by the Board of Directors or other

facts or events within the control of the Corporation) and may vary among holders thereof, provided that the manner in which such facts, events or variations shall operate upon the terms of such class or series of stock is clearly and expressly set forth in the articles supplementary or other Charter document.

SECTION 6.5 Stockholders' Consent in Lieu of Meeting. Any action required or permitted to be taken at any meeting of the stockholders may be taken without a meeting by consent, in writing or by electronic transmission, in any manner permitted by the MGCL and set forth in the Bylaws.

SECTION 6.6 <u>Charter and Bylaws</u>. The rights of all stockholders and the terms of all stock are subject to the provisions of the Charter and the Bylaws.

ARTICLE VII

RESTRICTION ON TRANSFER AND OWNERSHIP OF SHARES

SECTION 7.1 <u>Definitions</u>. For the purpose of this Article VII, the following terms shall have the following meanings:

Aggregate Stock Ownership Limit. The term "Aggregate Stock Ownership Limit" shall mean not more than 9.8 percent (in value or in number of shares, whichever is more restrictive) of the aggregate of the outstanding shares of Capital Stock.

Beneficial Ownership. The term "Beneficial Ownership" shall mean ownership of Capital Stock by a Person, whether the interest in the shares of Capital Stock is held directly or indirectly (including by a nominee), and shall include interests that would be treated as owned through the application of Section 544 of the Code, as modified by Section 856(h)(1)(B) of the Code. The terms "Beneficial Owner," "Beneficially Owns" and "Beneficially Owned" shall have the correlative meanings.

Business Day. The term "Business Day" shall mean any day, other than a Saturday or Sunday, that is neither a legal holiday nor a day on which banking institutions in New York City are authorized or required by law, regulation or executive order to close.

<u>Capital Stock</u>. The term "Capital Stock" shall mean all classes or series of stock of the Corporation, including, without limitation, Common Stock and Preferred Stock.

<u>Charitable Beneficiary</u>. The term "Charitable Beneficiary" shall mean one or more beneficiaries of the Trust as determined pursuant to Section 7.3.6, provided that each such organization must be described in Section 501(c)(3) of the Code and contributions to each such organization must be eligible for deduction under each of Sections 170(b)(1)(A), 2055 and 2522 of the Code.

Closing Price. The term "Closing Price" on any date shall mean the last sale price for such Capital Stock, regular way, or, in case no such sale takes place on such day, the average of the closing bid and asked prices, regular way, for such Capital Stock, in either case as reported in the principal consolidated transaction reporting system with respect to securities listed or admitted to trading on the NYSE AMEX, as reported on the principal consolidated transaction reporting system with respect to securities listed on the Principal aconsolidated transaction reporting system with respect to securities listed on the principal national securities exchange on which such Capital Stock is listed or admitted to trading or, if such Capital Stock is not listed or admitted to trading on any national securities exchange, the last quoted price, or, if not so quoted, the average of the high bid and low asked prices in the over-the-counter market, as reported by the principal automated quotation system that may then be in use or, if such Capital Stock is not quoted by any such system, the average of the closing bid and asked prices as furnished by a professional market maker making a market in such Capital Stock selected by the Board of Directors of the Corporation or, in the event that no trading price is available for such Capital Stock, the fair market value of the Capital Stock, as determined in good faith by the Board of Directors of the Corporation.

Common Stock Ownership Limit. The term "Common Stock Ownership Limit" shall mean not more than 9.8 percent (in value or in number of shares, whichever is more restrictive) of the aggregate of the outstanding shares of Common Stock of the Corporation.

Constructive Ownership. The term "Constructive Ownership" shall mean ownership of Capital Stock by a Person, whether the interest in the shares of Capital Stock is held directly or indirectly (including by a nominee), and shall include interests that would be treated as owned through the application of Section 318(a) of the Code, as modified by Section 856(d)(5) of the Code. The terms "Constructive Owner," "Constructively Owns" and "Constructively Owned" shall have the correlative meanings.

Excepted Holder. The term "Excepted Holder" shall mean a stockholder of the Corporation for whom an Excepted Holder Limit is created by this Charter or by the Board of Directors pursuant to Section 7.2.7.

Excepted Holder Limit. The term "Excepted Holder Limit" shall mean, provided that the affected Excepted Holder agrees to comply with the requirements established by the Board of Directors pursuant to Section 7.2.7 and subject to adjustment pursuant to Section 7.2.8, the percentage limit established by the Board of Directors pursuant to Section 7.2.7.

<u>Initial Date</u>. The term "Initial Date" shall mean the date upon which the Charter, as amended hereby, is accepted for record by the SDAT.

Market Price. The term "Market Price" on any date shall mean, with respect to any class or series of outstanding shares of Capital Stock, the Closing Price for such Capital Stock on such date.

NYSE AMEX. The term "NYSE AMEX" shall mean the NYSE Amex LLC.

<u>Person.</u> The term "Person" shall mean an individual, corporation, partnership, estate, trust (including a trust qualified under Sections 401(a) or 501(c)(17) of the Code), a portion of a trust permanently set aside for or to be used exclusively for the purposes described in Section 642(c) of the Code, association, private foundation within the meaning of Section 509(a) of the Code, joint stock company or other entity and also includes a group as that term is used for purposes of Section 13(d)(3) of the Securities Exchange Act of 1934, as amended, and a group to which an Excepted Holder Limit applies.

<u>Prohibited Owner</u>. The term "Prohibited Owner" shall mean, with respect to any purported Transfer, any Person who, but for the provisions of Section 7.2.1, would Beneficially Own or Constructively Own shares of Capital Stock in excess of the Aggregate Stock Ownership Limit, and if appropriate in the context, shall also mean any Person who would have been the record owner of the shares that the Prohibited Owner would have so owned.

Restriction Termination Date. The term "Restriction Termination Date" shall mean the first day after the Initial Date on which the Corporation determines pursuant to Section 5.7 of the Charter that it is no longer in the best interests of the Corporation to attempt to, or continue to, qualify as a REIT or that compliance with the restrictions and limitations on Beneficial Ownership, Constructive Ownership and Transfers of shares of Capital Stock set forth herein is no longer required in order for the Corporation to qualify as a REIT.

Transfer. The term "Transfer" shall mean any issuance, sale, transfer, gift, assignment, devise or other disposition, as well as any other event that causes any Person to acquire Beneficial Ownership or Constructive Ownership, or any agreement to take any such actions or cause any such events, of Capital Stock or the right to vote or receive dividends on Capital Stock, including (a) the granting or exercise of any option (or any disposition of any option), (b) any disposition of any securities or rights convertible into or exchangeable for Capital Stock or any interest in Capital Stock or any exercise of any such conversion or exchange right and (c) Transfers of interests in other entities that result in changes in Beneficial Ownership or Constructive Ownership of Capital Stock; in each case, whether voluntary or involuntary, whether owned of record, Constructively Owned or Beneficially Owned and whether by operation of law or otherwise. The terms "Transferring" and "Transferred" shall have the correlative meanings.

<u>Trust.</u> The term "Trust" shall mean any trust provided for in Section 7.3.1.

<u>Trustee</u>. The term "Trustee" shall mean the Person unaffiliated with the Corporation and a Prohibited Owner, that is appointed by the Corporation to serve as trustee of the Trust.

SECTION 7.2 Capital Stock

Section 7.2.1 Ownership Limitations. During the period commencing on the Initial Date and prior to the Restriction Termination Date, but subject to Section 7.4:

(a) <u>Basic Restrictions</u>.

(i) (1) No Person, other than an Excepted Holder, shall Beneficially Own or Constructively Own shares of Capital Stock in excess of the Aggregate Stock Ownership Limit, (2) no Person, other than an Excepted Holder, shall Beneficially Own or Constructively Own shares of Common Stock in excess of the Common Stock Ownership Limit and (3) no Excepted Holder shall Beneficially Own or Constructively Own shares of Capital Stock in excess of the Excepted Holder Limit for such Excepted Holder.

(ii) (1) No Person shall Beneficially Own or Constructively Own shares of Capital Stock to the extent that such Beneficial Ownership or Constructive Ownership of Capital Stock would result in the Corporation being "closely held" within the meaning of Section 856(h) of the Code (without regard to whether the ownership

interest is held during the last half of a taxable year), or (2) otherwise failing to qualify as a REIT (including, but not limited to, Beneficial Ownership or Constructive Ownership that would result in the Corporation owning (actually or Constructively) an interest in a tenant that is described in Section 856(d)(2)(B) of the Code if the income derived by the Corporation from such tenant could cause the Corporation to fail to satisfy any of the gross income requirements of Section 856(c) of the Code).

- (iii) Any Transfer of shares of Capital Stock that, if effective, would result in the Capital Stock being beneficially owned by less than 100 Persons (determined under the principles of Section 856(a)(5) of the Code) shall be void ab initio, and the intended transferee shall acquire no rights in such shares of Capital Stock.
- (b) <u>Transfer in Trust.</u> If any Transfer of shares of Capital Stock occurs which, if effective, would result in any Person Beneficially Owning or Constructively Owning shares of Capital Stock in violation of Section 7.2.1(a)(i) or (ii),
- (i) then that number of shares of the Capital Stock the Beneficial Ownership or Constructive Ownership of which otherwise would cause such Person to violate Section 7.2.1(a)(i) or (ii) (rounded to the nearest whole share) shall be automatically transferred to a Trust for the benefit of a Charitable Beneficiary, as described in Section 7.3, effective as of the close of business on the Business Day prior to the date of such Transfer, and such Person shall acquire no rights in such shares; or
- (ii) if the transfer to the Trust described in clause (i) of this sentence would not be effective for any reason to prevent the violation of Section 7.2.1(a)(i) or (ii), then the Transfer of that number of shares of Capital Stock that otherwise would cause any Person to violate Section 7.2.1(a)(i) or (ii) shall be void ab initio, and the intended transferee shall acquire no rights in such shares of Capital Stock.
- Section 7.2.2 Remedies for Breach. If the Board of Directors of the Corporation or any duly authorized committee thereof shall at any time determine in good faith that a Transfer or other event has taken place that results in a violation of Section 7.2.1 or that a Person intends to acquire or has attempted to acquire Beneficial Ownership or Constructive Ownership of any shares of Capital Stock in violation of Section 7.2.1 (whether or not such violation is intended), the Board of Directors or a committee thereof shall take such action as it deems advisable to refuse to give effect to or to prevent such Transfer or other event, including, without limitation, causing the Corporation to redeem shares, refusing to give effect to such Transfer on the books of the Corporation or instituting proceedings to enjoin such Transfer or other event; provided, however, that any Transfer or attempted Transfer or other event in violation of Section 7.2.1 shall automatically result in the transfer to the Trust described above, and, where applicable, such Transfer (or other event) shall be void ab initio as provided above irrespective of any action (or non-action) by the Board of Directors or a committee thereof.
- Section 7.2.3 Notice of Restricted Transfer. Any Person who acquires or attempts or intends to acquire Beneficial Ownership or Constructive Ownership of shares of Capital Stock that will or may violate Section 7.2.1(a) or any Person who would have owned shares of Capital Stock that resulted in a transfer to the Trust pursuant to the provisions of Section 7.2.1(b) shall immediately give written notice to the Corporation of such event or, in the case of such a proposed or attempted transaction, give at least 15 days prior written notice, and shall provide to the Corporation such other information as the Corporation may request in order to determine the effect, if any, of such Transfer on the Corporation's qualification as a REIT.
- Section 7.2.4 <u>Owners Required To Provide Information</u> From the Initial Date and prior to the Restriction Termination Date:
- (a) every owner of five percent or more (or such lower percentage as required by the Code or the U.S. Treasury Department regulations promulgated thereunder) of the outstanding shares of Capital Stock, within 30 days after the end of each taxable year, shall give written notice to the Corporation stating the name and address of such owner, the number of shares of Capital Stock and other shares of the Capital Stock Beneficially Owned and a description of the manner in which such shares are held. Each such owner shall provide to the Corporation such additional information as the Corporation may request in order to determine the effect, if any, of such Beneficial Ownership on the Corporation's qualification as a REIT and to ensure compliance with the Aggregate Stock Ownership Limit; and
- (b) each Person who is a Beneficial Owner or Constructive Owner of Capital Stock and each Person (including the stockholder of record) who is holding Capital Stock for a Beneficial Owner or Constructive Owner shall provide to the Corporation such information as the Corporation may request, in good faith, in order to determine the Corporation's qualification as a REIT and to comply with requirements of any taxing authority or governmental authority or to determine such compliance.

Section 7.2.5 <u>Remedies Not Limited.</u> Subject to Section 5.7 of the Charter, nothing contained in this Section 7.2 shall limit the authority of the Board of Directors of the Corporation to take such other action as it deems necessary or advisable to protect the Corporation and the interests of its stockholders in preserving the Corporation's qualification as a REIT.

Section 7.2.6 Ambiguity. In the case of an ambiguity in the application of any of the provisions of this Section 7.2, Section 7.3, or any definition contained in Section 7.1, the Board of Directors of the Corporation shall have the power to determine the application of the provisions of this Section 7.2 or Section 7.3 or any such definition with respect to any situation based on the facts known to it. In the event Section 7.2 or 7.3 requires an action by the Board of Directors and the Charter fails to provide specific guidance with respect to such action, the Board of Directors shall have the power to determine the action to be taken so long as such action is not contrary to the provisions of Sections 7.1, 7.2 or 7.3. Absent a decision to the contrary by the Board of Directors (which the Board may make in its sole and absolute discretion), if a Person would have (but for the remedies set forth in Section 7.2.2) acquired Beneficial Ownership or Constructive Ownership of Capital Stock in violation of Section 7.2.1, such remedies (as applicable) shall apply first to the shares of Capital Stock which, but for such remedies, would have been Beneficially Owned or Constructively Owned (but not actually owned) by such Person, pro rata among the Persons who actually own such shares of Capital Stock based upon the relative number of the shares of Capital Stock held by each such Person.

Section 7.2.7 Exceptions

- (a) Subject to Section 7.2.1(a)(ii), the Board of Directors of the Corporation, in its sole discretion, may exempt (prospectively or retroactively) a Person from the Aggregate Stock Ownership Limit and the Common Stock Ownership Limit, as the case may be, and may establish or increase an Excepted Holder Limit for such Person if:
- (i) the Board of Directors obtains such representations and undertakings from such Person as are reasonably necessary to ascertain that no individual's Beneficial Ownership or Constructive Ownership of such shares of Capital Stock will violate Section 7.2.1(a)(ii);
- (ii) such Person does not and represents that it will not own, actually or Constructively, an interest in a tenant of the Corporation (or a tenant of any entity owned or controlled by the Corporation) that would cause the Corporation to own, actually or Constructively, more than a 9.9% interest (as set forth in Section 856(d)(2)(B) of the Code) in such tenant and the Board of Directors obtains such representations and undertakings from such Person as are reasonably necessary to ascertain this fact (for this purpose, a tenant from whom the Corporation (or an entity owned or controlled by the Corporation) derives (and is expected to continue to derive) a sufficiently small amount of revenue such that, in the opinion of the Board of Directors of the Corporation, rent from such tenant would not adversely affect the Corporation's ability to qualify as a REIT shall not be treated as a tenant of the Corporation): and
- (iii) such Person agrees that any violation or attempted violation of such representations or undertakings (or other action which is contrary to the restrictions contained in Sections 7.2.1 through 7.2.6) will result in such shares of Capital Stock being automatically transferred to a Trust in accordance with Sections 7.2.1(b) and 7.3.

Notwithstanding anything to the contrary contained herein, the Board of Directors may, in its discretion, exempt a Person from the Aggregate Stock Ownership Limit and the Common Stock Ownership Limit prior to June 30, 2010, without regard to the parenthetical clause of Section 7.2.1(a)(ii)(1).

- (b) Prior to granting any exception pursuant to Section 7.2.7(a), the Board of Directors of the Corporation may require a ruling from the Internal Revenue Service, or an opinion of counsel, in either case in form and substance satisfactory to the Board of Directors in its sole discretion, as it may deem necessary or advisable in order to determine or ensure the Corporation's status as a REIT. Notwithstanding the receipt of any ruling or opinion, the Board of Directors may impose such conditions or restrictions as it deems appropriate in connection with granting such exception.
- (c) Subject to Section 7.2.1(a)(ii), an underwriter which participates in a public offering or a private placement of Capital Stock (or securities convertible into or exchangeable for Capital Stock) may Beneficially Own or Constructively Own shares of Capital Stock (or securities convertible into or exchangeable for Capital Stock) in excess of the Aggregate Stock Ownership Limit, the Common Stock Ownership Limit, or both such limits, but only to the extent necessary to facilitate such public offering or private placement.
- (d) The Board of Directors may only reduce the Excepted Holder Limit for an Excepted Holder: (1) with the written consent of such Excepted Holder at any time, or (2) pursuant to the terms and conditions of the agreements and undertakings entered into with such Excepted Holder in connection with the establishment of the Excepted Holder Limit for that Excepted Holder. No Excepted Holder Limit shall be reduced to a percentage that is less than the Common Stock Ownership Limit.

Section 7.2.8 Increase in Aggregate Stock Ownership and Common Stock Ownership Limits Subject to Section 7.2.1 (a)(ii), the Board of Directors may from time to time increase the Common Stock Ownership Limit and the Aggregate Stock Ownership Limit for one or more Persons and decrease the Common Stock Ownership Limit and the Aggregate Stock Ownership Limit for all other Persons; provided, however, that the decreased Common Stock Ownership Limit and/or Aggregate Stock Ownership Limit will not be effective for any Person whose percentage ownership in Common Stock is in excess of such decreased Common Stock Ownership Limit and/or whose percentage ownership in Capital Stock is in excess of such decreased Aggregate Stock Ownership Limit, as applicable, until such time as such Person's percentage of Common Stock equals or falls below the decreased Common Stock Ownership Limit, as applicable, but any further acquisition of Capital Stock equals or falls below the decreased Aggregate Stock Ownership Limit, as applicable, but any further acquisition of Capital Stock in excess of such percentage ownership of Common Stock and/or Capital Stock will be in violation of the Common Stock Ownership Limit and/or Aggregate Stock Ownership Limit, as applicable, and, provided further, that the new Common Stock Ownership Limit and/or Aggregate Stock Ownership Limit would not allow five or fewer Persons to Beneficially Own more than 49.9% in value of the outstanding Capital Stock.

Section 7.2.9 <u>Legend</u>. Each certificate for shares of Capital Stock, if certificated, or the written statement of information in lieu of a certificate shall bear substantially the following legend:

The shares represented by this certificate are subject to restrictions on Beneficial Ownership and Constructive Ownership and Transfer for the purpose, among others, of the Corporation's maintenance of its qualification as a Real Estate Investment Trust under the Code. Subject to certain further restrictions and except as expressly provided in the Corporation's Charter, (i) no Person may Beneficially Own or Constructively Own shares of the Corporation's Common Stock in excess of 9.8 percent (in value or number of shares) of the outstanding shares of Common Stock of the Corporation unless such Person is an Excepted Holder (in which case the Excepted Holder Limit shall be applicable); (ii) no Person may Beneficially Own or Constructively Own shares of Capital Stock of the Corporation in excess of 9.8 percent (in value or number of shares) of the total outstanding shares of Capital Stock of the Corporation, unless such Person is an Excepted Holder (in which case the Excepted Holder Limit shall be applicable); (iii) no Person may Beneficially Own or Constructively Own Capital Stock that would result in the Corporation being "closely held" under Section 856(h) of the Code or otherwise cause the Corporation to fail to qualify as a REIT; and (iv) no Person may Transfer shares of Capital Stock if such Transfer would result in the Capital Stock of the Corporation being owned by fewer than 100 Persons. Any Person who Beneficially Owns or Constructively Owns or attempts to Beneficially Own or Constructively Own shares of Capital Stock which causes or will cause a Person to Beneficially Own or Constructively Own shares of Capital Stock in excess or in violation of the above limitations must immediately notify the Corporation. If the restrictions on transfer or ownership provided in (i), (ii) or (iii) above are violated, the shares of Capital Stock in excess or in violation of the above limitations will be aut omatically transferred to a Trustee of a Trust for the benefit of one or more Charitable Beneficiaries. In addition, the Corporation may redeem shares upon the terms and conditions specified by the Board of Directors in its sole discretion if the Board of Directors determines that ownership or a Transfer or other event may violate the restrictions described above. Furthermore, if the ownership restriction provided in (iv) above would be violated or upon the occurrence of certain events, attempted Transfers in violation of the restrictions described above may be void ab initio. All capitalized terms in this legend have the meanings defined in the Charter of the Corporation, as the same may be amended from time to time, a copy of which, including the restrictions on transfer and ownership, will be furnished to each holder of Capital Stock of the Corporation on request and without charge. Requests for such a copy may be directed to the Secretary of the Corporation at its principal office.

Instead of the foregoing legend, the certificate or written statement of information in lieu of a certificate may state that the Corporation will furnish a full statement about certain restrictions on transferability to a stockholder on request and without charge.

SECTION 7.3 Transfer of Capital Stock in Trust.

Section 7.3.1 Ownership in Trust. Upon any purported Transfer or other event described in Section 7.2.1(b) that would result in a transfer of shares of Capital Stock to a Trust, such shares of Capital Stock shall be deemed to have been transferred to the Trustee as trustee of a Trust for the exclusive benefit of one or more Charitable Beneficiaries. Such transfer to the Trustee shall be deemed to be effective as of the close of business on the Business Day prior to the purported Transfer or other event that results in the transfer to the Trust pursuant to Section 7.2.1(b). The Trustee shall be appointed by the Corporation and shall be a Person unaffiliated with the Corporation and any Prohibited Owner. Each Charitable Beneficiary shall be designated by the Corporation as provided in Section 7.3.6.

Section 7.3.2 <u>Status of Shares Held by the Trustee</u>. Shares of Capital Stock held by the Trustee shall be issued and outstanding shares of Capital Stock of the Corporation. The Prohibited Owner shall have no rights in the shares held by the Trustee. The Prohibited Owner shall not benefit economically from ownership of any shares held in trust by the Trustee, shall have no rights to dividends or other distributions and shall not possess any rights to vote or other rights attributable to the shares held in the Trust

Section 7.3.3 Dividend and Voting Rights. The Trustee shall have all voting rights and rights to dividends or other distributions with respect to shares of Capital Stock held in the Trust, which rights shall be exercised for the exclusive benefit of the Charitable Beneficiary. Any dividend or other distribution paid prior to the discovery by the Corporation that the shares of Capital Stock have been transferred to the Trustee shall be paid by the recipient of such dividend or distribution to the Trustee upon demand and any dividend or other distribution authorized but unpaid shall be paid when due to the Trustee. Any dividend or distribution so paid to the Trustee shall be held in trust for the Charitable Beneficiary. The Prohibited Owner shall have no voting rights with respect to shares held in the Trust and, subject to Maryland law, effective as of the date that the shares of Capital Stock have been transferred to the Trustee, the Trustee shall have the authority (at the Trustee's sole discretion) (i) to rescind as void any vote cast by a Prohibited Owner prior to the discovery by the Corporation that the shares of Capital Stock have been transferred to the Trustee and (ii) to recast such vote in accordance with the desires of the Trustee acting for the benefit of the Charitable Beneficiary; provided, however, that if the Corporation has already taken irreversible corporate action, then the Trustee shall not have the authority to rescind and recast such vote. Notwithstanding the provisions of this Article VII, until the Corporation has received notification that shares of Capital Stock have been transferred into a Trust, the Corporation shall be entitled to rely on its share transfer and other stockholder records for purposes of preparing lists of stockholders entitled to vote at meetings, determining the validity and authority of proxies and otherwise conducting votes of stockholders.

Sale of Shares by Trustee. Within 20 days of receiving notice from the Corporation that shares Section 7 3 4 of Capital Stock have been transferred to the Trust, the Trustee of the Trust shall sell the shares held in the Trust to a person, designated by the Trustee, whose ownership of the shares will not violate the ownership limitations set forth in Section 7.2.1(a). Upon such sale, the interest of the Charitable Beneficiary in the shares sold shall terminate and the Trustee shall distribute the net proceeds of the sale to the Prohibited Owner and to the Charitable Beneficiary as provided in this Section 7.3.4. The Prohibited Owner shall receive the lesser of (1) the price paid by the Prohibited Owner for the shares or, if the Prohibited Owner did not give value for the shares in connection with the event causing the shares to be held in the Trust (e.g., in the case of a gift, devise or other such transaction), the Market Price of the shares on the day of the event causing the shares to be held in the Trust and (2) the price per share received by the Trustee (net of any commissions and other expenses of sale) from the sale or other disposition of the shares held in the Trust. The Trustee may reduce the amount payable to the Prohibited Owner by the amount of dividends and distributions which have been paid to the Prohibited Owner and are owed by the Prohibited Owner to the Trustee pursuant to Section 7.3.3 of this Article VII. Any net sales proceeds in excess of the amount payable to the Prohibited Owner shall be immediately paid to the Charitable Beneficiary. If, prior to the discovery by the Corporation that shares of Capital Stock have been transferred to the Trustee, such shares are sold by a Prohibited Owner, then (i) such shares shall be deemed to have been sold on behalf of the Trust and (ii) to the extent that the Prohibited Owner received an amount for such shares that exceeds the amount that such Prohibite d Owner was entitled to receive pursuant to this Section 7.3.4, such excess shall be paid to the Trustee upon demand.

Section 7.3.5 Purchase Right in Stock Transferred to the Trustee. Shares of Capital Stock transferred to the Trustee shall be deemed to have been offered for sale to the Corporation, or its designee, at a price per share equal to the lesser of (i) the price per share in the transaction that resulted in such transfer to the Trust (or, in the case of a devise or gift, the Market Price at the time of such devise or gift) and (ii) the Market Price on the date the Corporation, or its designee, accepts such offer. The Corporation may reduce the amount payable to the Prohibited Owner by the amount of dividends and distributions which has been paid to the Prohibited Owner and are owed by the Prohibited Owner to the Trustee pursuant to Section 7.3.3 of this Article VII. The Corporation may pay the amount of such reduction to the Trustee for the benefit of the Charitable Beneficiary. The Corporation shall have the right to accept such offer until the Trustee has sold the shares held in the Trust pursuant to Section 7.3.4. Upon such a sale to the Corporation, the interest of the Charitable Beneficiary in the shares sold shall terminate and the Trustee shall distribute the net proceeds of the sale to the Prohibited Owner.

Section 7.3.6 <u>Designation of Charitable Beneficiaries</u>. By written notice to the Trustee, the Corporation shall designate one or more nonprofit organizations to be the Charitable Beneficiary of the interest in the Trust such that the shares of Capital Stock held in the Trust would not violate the restrictions set forth in Section 7.2.1(a) in the hands of such Charitable Beneficiary.

SECTION 7.4 NYSE AMEX Transactions. Nothing in this Article VII shall preclude the settlement of any transaction entered into through the facilities of the NYSE AMEX or any other national securities exchange or automated inter-dealer quotation system. The fact that the settlement of any transaction occurs shall not negate the effect of any other provision of this Article VII and any transferee in such a transaction shall be subject to all of the provisions and limitations set forth in this Article VII.

SECTION 7.5 <u>Enforcement.</u> The Corporation is authorized specifically to seek equitable relief, including injunctive relief, to enforce the provisions of this Article VII.

SECTION 7.6 Non-Waiver. No delay or failure on the part of the Corporation or the Board of Directors in exercising any right hereunder shall operate as a waiver of any right of the Corporation or the Board of Directors, as the case may be, except to the extent specifically waived in writing.

ARTICLE VIII

AMENDMENTS

The Corporation reserves the right from time to time to make any amendment to its Charter, now or hereafter authorized by law, including any amendment altering the terms or contract rights, as expressly set forth in the Charter, of any shares of outstanding stock. All rights and powers conferred by the Charter on stockholders, directors and officers are granted subject to this reservation. Except as set forth below and except for those amendments permitted to be made without stockholder approval under Maryland law or by specific provision in the Charter, any amendment to the Charter shall be valid only if declared advisable by the Board of Directors and approved by the affirmative vote of holders of shares entitled to cast a majority of all the votes entitled to be cast on the matter. Any amendment to Section 5.8, Article VII or to this sentence of the Charter shall be valid only if declared advisable by the Board of Directors and approved by the affirmative vote of holders of shares entitled to cast at least two-thirds of all the votes entitled to be cast on the matter.

ARTICLE IX

LIMITATION OF LIABILITY

To the maximum extent that Maryland law in effect from time to time permits limitation of the liability of directors and officers of a corporation, no present or former director or officer of the Corporation shall be liable to the Corporation or its stockholders for money damages. Neither the amendment nor repeal of this Article IX, nor the adoption or amendment of any other provision of the Charter or Bylaws inconsistent with this Article IX, shall apply to or affect in any respect the applicability of the preceding sentence with respect to any act or failure to act which occurred prior to such amendment, repeal or adoption.

[SIGNATURE PAGE TO FOLLOW]

The undersigned Co-Chief Executive Officer acknowledges these Articles of Amendment and Restatement to be the corporate act of the Corporation and as to all matters or facts required to be verified under oath, the undersigned Co-Chief Executive Officer acknowledges that, to the best of his knowledge, information and belief, these matters and facts are true in all material respects and that this statement is made under the penalties for perjury.

IN WITNESS WHEREOF, the Corporation has caused these Articles of Amendment and Restatement to be signed in its name and on its behalf by its Co-Chief Executive Officer and attested to by its Secretary on this 5th day of November, 2009.

ATTEST:	ARMOUR Residential REIT, Inc.				
By: /s/ Jeffrey J. Zimmer	By: /s/ Scott J. Ulm	(SEAL)			
Title: Secretary	Title:Co-Chief Executive Officer				

[Signature Page of ARMOUR Articles of Amendment and Restatement]

ARMOUR RESIDENTIAL REIT, INC.

BYLAWS

ARTICLE I

OFFICES

Section 1. <u>PRINCIPAL OFFICE</u>. The principal office of the Corporation in the State of Maryland shall be located at such place as the Board of Directors may designate.

Section 2. <u>ADDITIONAL OFFICES</u>. The Corporation may have additional offices, including a principal executive office, at such places as the Board of Directors may from time to time determine or the business of the Corporation may require.

ARTICLE II

MEETINGS OF STOCKHOLDERS

Section 1. <u>PLACE</u>. All meetings of stockholders shall be held at the principal executive office of the Corporation or at such other place as shall be set by the Board of Directors and stated in the notice of the meeting.

Section 2. <u>ANNUAL MEETING</u>. An annual meeting of the stockholders for the election of directors and the transaction of any business within the powers of the Corporation shall be held on the date and at the time set by the Board of Directors.

Section 3. SPECIAL MEETINGS.

(a) <u>General</u>. The chairman of the board, president, chief executive officer or Board of Directors may call a special meeting of the stockholders. Subject to <u>Section 3(b)</u>, a special meeting of stockholders shall also be called by the secretary of the Corporation to act on any matter that may properly be considered at a meeting of stockholders upon the written request of stockholders entitled to cast not less than a majority of all the votes entitled to be cast on such matter at such meeting.

(b) Stockholder-Requested Special Meetings.

(1) Any stockholder of record seeking to have stockholders request a special meeting shall, by sending written notice to the secretary (the "Record Date Request Notice") by registered mail, return receipt requested, request the Board of Directors to fix a record date to determine the stockholders entitled to request a special meeting (the "Request Record Date"). The Record Date Request Notice shall set forth the purpose of the meeting and the matters proposed to be acted on at it, shall be signed by one or more stockholders of record as of the date of signature (or their agents duly authorized in a writing accompanying the Record Date Request Notice), shall bear the date of signature of each such stockholder (or such agent) and shall set forth all information relating to each such stockholder and each matter proposed to be acted on at the meeting that would be required to be disclosed in connection with the solicitation of proxies for the election of directors in an election contest (even if an election contest is not involved), or would otherwise be required in connection with such a solicitation, in each case pursuant to Regulation 14A (or any successor provision) under the Securities Exchange Act of 1934, as amended (the "Exchange Act"). Upon receiving the Record Date Request Notice, the Board of Directors may fix a Request Record Date. The Request Record Date shall not precede and shall not be more than ten days after the close of business on the date on which the resolution fixing the Request Record Date is adopted by the Board of Directors. If the Board of Directors, within ten days after the date on which a valid Record Date Request Notice is received, fails to adopt a resolution fixing the Request Record Date, the Request Record Date shall be the close of business on the tenth day after the first date on which the Record Date Request Notice is received by the secretary.

(2) In order for any stockholder to request a special meeting to act on any matter that may properly be considered at a meeting of stockholders, one or more written requests for a special meeting (collectively, the "Special Meeting Request") signed by stockholders of record (or their agents duly authorized in a writing accompanying the request) as of the Request Record Date entitled to cast not less than a majority of all of the votes entitled to be cast on such matter at such meeting (the "Special Meeting Percentage") shall be delivered to the secretary. In addition, the Special Meeting Request shall (a) set forth the purpose of the meeting and the matters proposed to be acted on at it (which shall be limited to those lawful matters set forth in the Record Date Request Notice received by the secretary), (b) bear the date of signature of each such stockholder (or such agent) signing the Special Meeting Request, (c) set forth (i) the name and address, as they appear in the Corporation's books, of each stockholder signing such request (or on whose behalf the Special Meeting Request is

signed), (ii) the class, series and number of all shares of stock of the Corporation which are owned (beneficially or of record) by such stockholder and (iii) the nominee holder for, and number of, shares of stock of the Corporation owned beneficially but not of record by such stockholder, (d) be sent to the secretary by registered mail, return receipt requested, and (e) be received by the secretary within 60 days after the Request Record Date. Any requesting stockholder (or agent duly authorized in a writing accompanying the revocation or the Special Meeting Request) may revoke his, her or its request for a special meeting at any time by written revocation delivered to the secretary

- (3) The secretary shall inform the requesting stockholders of the reasonably estimated cost of preparing and delivering the notice of the meeting (including the Corporation's proxy materials). The secretary shall not be required to call a special meeting upon stockholder request and such meeting shall not be held unless, in addition to the documents required by Section 3(b)(2), the secretary receives payment of such reasonably estimated cost prior to the preparation and mailing or delivery of such notice of the meeting.
- (4) In the case of any special meeting called by the secretary upon the request of stockholders (a 'Stockholder-Requested Meeting"), such meeting shall be held at such place, date and time as may be designated by the Board of Directors; provided, however, that the date of any Stockholder-Requested Meeting shall be not more than 90 days after the record date for such meeting (the "Meeting Record Date"); and provided further that if the Board of Directors fails to designate, within ten days after the date that a valid Special Meeting Request is actually received by the secretary (the "Delivery Date"), a date and time for a Stockholder-Requested Meeting, then such meeting shall be held at 2:00 p.m. local time on the 90th day after the Meeting Record Date or, if such 90th day is not a Business Day (as defined below), on the first preceding Business Day; and provided further that in the event that the Board of Directors fails to designate a place for a Stockholder-Requested Meeting within ten days after the Delivery Date, then such meeting shall be held at the principal executive office of the Corporation. In fixing a date for any special meeting, the chairman of the board, chief executive officer, president or Board of Directors may consider such factors as he, she or it deems relevant, including, without limitation, the nature of the matters to be considered, the facts and circumstances surrounding any request for the meeting and any plan of the Board of Directors to call an annual meeting or a special meeting. In the case of any Stockholder-Requested Meeting, if the Board of Directors fails to fix a Meeting Record Date that is a date within 30 days after the Delivery Date, then the close of business on the 30th day after the Delivery Date shall be the Meeting Record Date. The Board of Directors may revoke the notice for any Stockholder-Requested Meeting in the event that the requesting stockholders fail to comply with the provisions of Section 3(b)(3).
- (5) If written revocations of the Special Meeting Request have been delivered to the secretary and the result is that stockholders of record (or their agents duly authorized in writing), as of the Request Record Date, entitled to cast less than the Special Meeting Percentage have delivered, and not revoked, requests for a special meeting to the secretary: (i) if the notice of meeting has not already been delivered, the secretary shall refrain from delivering the notice of the meeting and send to all requesting stockholders who have not revoked such requests written notice of any revocation of a request for a special meeting on the matter, or (ii) if the notice of meeting has been delivered and if the secretary first sends to all requesting stockholders who have not revoked requests for a special meeting on the matter written notice of any revocation of a request for the special meeting and written notice of the Corporation's intention to revoke the notice of the meeting or for the chairman of the meeting and written notice of the meeting or (B) the secretary may revoke the notice of the meeting at any time before ten days before the commencement of the meeting or (B) the chairman of the meeting may call the meeting to order and adjourn the meeting without acting on the matter. Any request for a special meeting received after a revocation by the secretary of a notice of a meeting shall be considered a request for a new special meeting.
- (6) The chairman of the board, chief executive officer, president or Board of Directors may appoint regionally or nationally recognized independent inspectors of elections to act as the agent of the Corporation for the purpose of promptly performing a ministerial review of the validity of any purported Special Meeting Request received by the secretary. For the purpose of permitting the inspectors to perform such review, no such purported Special Meeting Request shall be deemed to have been delivered to the secretary until the earlier of (i) five Business Days after receipt by the secretary of such purported request and (ii) such date as the independent inspectors certify to the Corporation that the valid requests received by the secretary represent, as of the Request Record Date, stockholders of record entitled to cast not less than the Special Meeting Percentage. Nothing contained in this paragraph (6) shall in any way be construed to suggest or imply that the Corporation or any stockholder shall not be entitled to contest the validity of any request, whether during or after such five Business Day period, or to take any other action (including, without limitation, the commencement, prosecution or defense of any litigation with respect thereto, and the seeking of injunctive relief in such litigation).
- (7) For purposes of these Bylaws, "Business Day" shall mean any day other than a Saturday, a Sunday or a day on which banking institutions in the State of New York are authorized or obligated by law or executive order to close.

Section 4. NOTICE. Not less than ten nor more than 90 days before each meeting of stockholders, the secretary shall give to each stockholder entitled to vote at such meeting and to each stockholder not entitled to vote who is entitled to notice of the meeting notice in writing or by electronic transmission stating the time and place of the meeting and, in the case of a special meeting or as otherwise may be required by any statute, the purpose for which the meeting is called, by mail, by presenting it to such stockholder personally, by leaving it at the stockholder's residence or usual place of business or by any other means permitted by Maryland law. If mailed, such notice shall be deemed to be given when deposited in the United States mail addressed to the stockholder at the stockholder's address as it appears on the records of the Corporation, with postage thereon prepaid. If transmitted electronically, such notice shall be deemed to be given when transmitted to the stockholder by an electronic transmission to any address or number of the stockholder at which the stockholder receives electronic transmissions. The Corporation may give a single notice to all stockholders who share an address, which single notice shall be effective as to any stockholder at such address, unless a stockholder objects to receiving such single notice or revokes a prior consent to receiving such single notice. Failure to give notice of any meeting to one or more stockholders, or any irregularity in such notice, shall not affect the validity of any meeting fixed in accordance with this Article II or the validity of any proceedings at any such meeting.

Subject to Section 11(a) of this Article II, any business of the Corporation may be transacted at an annual meeting of stockholders without being specifically designated in the notice, except such business as is required by any statute to be stated in such notice. No business shall be transacted at a special meeting of stockholders except as specifically designated in the notice. The Corporation may postpone or cancel a meeting of stockholders by making a public announcement (as defined in Section 11(c)(3) of this Article II) of such postponement or cancellation prior to the meeting. Notice of the date, time and place to which the meeting is postponed shall be given not less than ten days prior to such date and otherwise in the manner set forth in this section.

Section 5. ORGANIZATION AND CONDUCT. Every meeting of stockholders shall be conducted by an individual appointed by the Board of Directors to be chairman of the meeting or, in the absence of such appointment, by the chairman of the board or, in the case of a vacancy in the office or absence of the chairman of the board, by one of the following officers present at the meeting in the following order: the vice chairman of the board, if there is one, the president, the vice presidents in their order of rank and seniority, the secretary, or, in the absence of such officers, a chairman chosen by the stockholders by the vote of a majority of the votes cast by stockholders present in person or by proxy. The secretary, or, in the secretary's absence, an assistant secretary, or, in the absence of both the secretary and assistant secretaries, a person appointed by the Board of Directors or, in the absence of such appointment, a person appointed by the chairman of the meeting shall act as secretary. In the event that the secretary presides at a meeting of the stockholders, an assistant secretary, or, in the absence of assistant secretaries, an individual appointed by the Board of Directors or the chairman of the meeting, shall record the minutes of the meeting. The order of business and all other matters of procedure at any meeting of stockholders shall be determined by the chairman of the meeting. The chairman of the meeting may prescribe such rules, regulations and procedures and take such action as, in the discretion of the chairman and without any action by the stockholders, are appropriate for the proper conduct of the meeting, including, without limitation,: (a) restricting admission to the time set for the commencement of the meeting; (b) limiting attendance at the meeting to stockholders of record of the Corporation, their duly authorized proxies and other such individuals as the chairman of the meeting may determine; (c) limiting participation at the meeting on any matter to stockholders of record of the Corporation entitled to vote on such matter, their duly authorized proxies and other such individuals as the chairman of the meeting may determine; (d) limiting the time allotted to questions or comments; (e) determining when and for how long the polls should be opened and when the polls should be closed; (f) maintaining order and security at the meeting; (g) removing any stockholder or any other individual who refuses to comply with meeting procedures, rules or guidelines as set forth by the chairman of the meeting; (h) concluding a meeting or recessing or adjourning the meeting to a later date and time and at a place announced at the meeting; and (i) complying with any state and local laws and regulations concerning safety and security. Unless otherwise determined by the chairman of the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure.

Section 6. QUORUM. At any meeting of stockholders, the presence in person or by proxy of stockholders entitled to cast a majority of all the votes entitled to be cast at such meeting on any matter shall constitute a quorum; but this section shall not affect any requirement under any statute or the charter of the Corporation (the "Charter") for the vote necessary for the adoption of any measure. If such quorum is not established at any meeting of the stockholders, the chairman of the meeting may adjourn the meeting sine die or from time to time to a date not more than one hundred 120 days after the original record date without notice other than announcement at the meeting. At such adjourned meeting at which a quorum shall be present, any business may be transacted which might have been transacted at the meeting as originally notified.

The stockholders present either in person or by proxy, at a meeting which has been duly called and at which a quorum has been established, may continue to transact business until adjournment, notwithstanding the withdrawal from the meeting of enough stockholders to leave fewer than required to establish a quorum.

Section 7. <u>VOTING</u>. A plurality of all the votes cast at a meeting of stockholders duly called and at which a quorum is present shall be sufficient to elect a director. Each share may be voted for as many individuals as there are directors to be elected and for whose election the share is entitled to be voted. A majority of the votes cast at a meeting of stockholders duly called and at which a quorum is present shall be sufficient to approve any other matter which may properly come before the meeting, unless more than a majority of the votes cast is required by statute or by the Charter. Unless otherwise provided by statute or by the Charter, each outstanding share, regardless of class, shall be entitled to one vote on each matter submitted to a vote at a meeting of stockholders. Voting on any question or in any election may be *viva voce* unless the chairman of the meeting shall order that voting be by ballot or otherwise.

Section 8. <u>PROXIES</u>. A stockholder may cast the votes entitled to be cast by the holder of the shares of stock owned of record by the stockholder in person or by proxy executed by the stockholder or by the stockholder's duly authorized agent in any manner permitted by law. Such proxy or evidence of authorization of such proxy shall be filed with the secretary of the Corporation before or at the meeting. No proxy shall be valid more than eleven months after its date unless otherwise provided in the proxy.

Section 9. <u>VOTING OF STOCK BY CERTAIN HOLDERS</u>. Stock of the Corporation registered in the name of a corporation, partnership, trust or other entity, if entitled to be voted, may be voted by the president or a vice president, a general partner, trustee or managing member thereof, as the case may be, or a proxy appointed by any of the foregoing individuals, unless some other person who has been appointed to vote such stock pursuant to a bylaw or a resolution of the governing body of such corporation or other entity or agreement of the partners of a partnership presents a certified copy of such bylaw, resolution or agreement, in which case such person may vote such stock. Any director or fiduciary may vote stock registered in the name of such person in the capacity of such director or fiduciary, either in person or by proxy.

Shares of stock of the Corporation directly or indirectly owned by it shall not be voted at any meeting and shall not be counted in determining the total number of outstanding shares entitled to be voted at any given time, unless they are held by it in a fiduciary capacity, in which case they may be voted and shall be counted in determining the total number of outstanding shares at any given time.

The Board of Directors may adopt by resolution a procedure by which a stockholder may certify in writing to the Corporation that any shares of stock registered in the name of the stockholder are held for the account of a specified person other than the stockholder. The resolution shall set forth the class of stockholders who may make the certification, the purpose for which the certification may be made, the form of certification and the information to be contained in it; if the certification is with respect to a record date, the time after the record date within which the certification must be received by the Corporation; and any other provisions with respect to the procedure which the Board of Directors considers necessary or desirable. On receipt of such certification, the person specified in the certification shall be regarded as, for the purposes set forth in the certification, the holder of record of the specified stock in place of the stockholder who makes the certification.

Section 10. <u>INSPECTORS</u>. The Board of Directors or the chairman of the meeting may appoint, before or at the meeting, one or more inspectors for the meeting and any successor to the inspector. The inspectors, if any, shall: (a) determine the number of shares of stock represented at the meeting, in person or by proxy, and the validity and effect of proxies; (b) receive and tabulate all votes, ballots or consents; (c) report such tabulation to the chairman of the meeting; (d) hear and determine all challenges and questions arising in connection with the right to vote; and (e) do such acts as are proper to fairly conduct the election or vote. Each such report shall be in writing and signed by the inspector or by a majority of them if there is more than one inspector acting at such meeting. If there is more than one inspector, the report of a majority shall be the report of the inspectors. The report of the inspector or inspectors on the number of shares represented at the meeting and the results of the voting shall be *prima facie* evidence thereof.

Section 11. <u>ADVANCE NOTICE OF STOCKHOLDER NOMINEES FOR DIRECTOR AND OTHER STOCKHOLDER PROPOSALS</u>.

(a) Annual Meetings of Stockholders.

(1) Nominations of individuals for election to the Board of Directors and the proposal of other business to be considered by the stockholders may be made at an annual meeting of stockholders (i) pursuant to the Corporation's notice of meeting, (ii) by or at the direction of the Board of Directors or (iii) by any stockholder of the Corporation who was a stockholder of record both at the time of giving of notice by the stockholder as provided for in this Section 11(a) and at the time of the annual meeting, who is entitled to vote at the meeting in the election of each individual so nominated or on any such other business and who has complied with this Section 11(a).

(2) For any nomination or other business to be properly brought before an annual meeting by a stockholder pursuant to Section 11(a)(1)(iii), the stockholder must have given timely notice thereof in writing to the secretary

of the Corporation and such other business must otherwise be a proper matter for action by the stockholders. To be timely, a stockholder's notice shall set forth all information required under this Section 11 and shall be delivered to the secretary at the principal executive office of the Corporation not earlier than the 150th day nor later than 5:00 p.m., Eastern Time, on the 120th day prior to the first anniversary of the date of the proxy statement for the preceding year's annual meeting; provided, however, that in connection with the Corporation's first annual meeting or in the event that the date of the annual meeting is advanced or delayed by more than 30 days from the first anniversary of the date of the preceding year's annual meeting, notice by the stockholder to be timely must be so delivered not earlier than the 150th day prior to the date of such annual meeting and not later than 5:00 p.m., Eastern Time, on the later of the 120th day prior to the date of such annual meeting, as originally convened, or the tenth day following the day on which public announcement of the date of such meeting is first made. The public announcement of a postponement or adjournment of an annual meeting shall not commence a new time period for the giving of a stockholder's notice as described above.

- (3) Such stockholder's notice shall set forth:
- (i) as to each individual whom the stockholder proposes to nominate for election or reelection as a director (each, a "<u>Proposed Nominee</u>"), all information relating to the Proposed Nominee that would be required to be disclosed in connection with the solicitation of proxies for the election of the Proposed Nominee as a director in an election contest (even if an election contest is not involved), or would otherwise be required in connection with such solicitation, in each case pursuant to Regulation 14A (or any successor provision) under the Exchange Act and the rules thereunder;
- (ii) as to any business that the stockholder proposes to bring before the meeting, a description of such business, the stockholder's reasons for proposing such business at the meeting and any material interest in such business of such stockholder or any Stockholder Associated Person (as defined below), individually or in the aggregate, including any anticipated benefit to the stockholder or the Stockholder Associated Person therefrom;
 - (iii) as to the stockholder giving the notice, any Proposed Nominee and any Stockholder Associated Person,
- (A) the class, series and number of all shares of stock or other securities of the Corporation or any affiliate thereof (collectively, the "Company Securities"), if any, which are owned (beneficially or of record) by such stockholder, Proposed Nominee or Stockholder Associated Person, the date on which each such Company Security was acquired and the investment intent of such acquisition, and any short interest (including any opportunity to profit or share in any benefit from any decrease in the price of such stock or other security) in any Company Securities of any such person,
- (B) the nominee holder for, and number of, any Company Securities owned beneficially but not of record by such stockholder, Proposed Nominee or Stockholder Associated Person,
- (C) whether and the extent to which such stockholder, Proposed Nominee or Stockholder Associated Person, directly or indirectly (through brokers, nominees or otherwise), is subject to or during the last six months has engaged in any hedging, derivative or other transaction or series of transactions or entered into any other agreement, arrangement or understanding (including any short interest, any borrowing or lending of securities or any proxy or voting agreement), the effect or intent of which is to (I) manage risk or benefit of changes in the price of (x) Company Securities or (y) any security of any entity that was listed in the Peer Group in the Stock Performance Graph in the most recent annual report to security holders of the Corporation (a "Peer Group Company") for such stockholder, Proposed Nominee or Stockholder Associated Person or (II) increase or decrease the voting power of such stockholder, Proposed Nominee or Stockholder Associated Person in the Corporation or any affiliate thereof (or, as applicable, in any Peer Group Company) disproportionately to such person's economic interest in the Company Securities (or, as applicable, in any Peer Group Company), and
- (D) any substantial interest, direct or indirect (including, without limitation, any existing or prospective commercial, business or contractual relationship with the Corporation), by security holdings or otherwise, of such stockholder, Proposed Nominee or Stockholder Associated Person, in the Corporation or any affiliate thereof, other than an interest arising from the ownership of Company Securities where such stockholder, Proposed Nominee or Stockholder Associated Person receives no extra or special benefit not shared on a *pro rata* basis by all other holders of the same class or series;
- (iv) as to the stockholder giving the notice, any Stockholder Associated Person with an interest or ownership referred to Section 11(a)(3)(ii) or (iii) and any Proposed Nominee,
- (A) the name and address of such stockholder, as they appear on the Corporation's stock ledger, and the current name and business address, if different, of each such Stockholder Associated Person and any Proposed Nominee and

(B) the investment strategy or objective, if any, of such stockholder and each such Stockholder Associated Person who is not an individual and a copy of the prospectus, offering memorandum or similar document, if any, provided to investors or potential investors in such stockholder and each such Stockholder Associated Person; and

- (v) to the extent known by the stockholder giving the notice, the name and address of any other stockholder supporting the nominee for election or reelection as a director or the proposal of other business on the date of such stockholder's notice
- (4) Such stockholder's notice shall, with respect to any Proposed Nominee, be accompanied by a certificate executed by the Proposed Nominee (i) certifying that such Proposed Nominee (a) is not, and will not become a party to, any agreement, arrangement or understanding with any person or entity other than the Corporation in connection with service or action as a director that has not been disclosed to the Corporation and (b) will serve as a director of the Corporation if elected; and (ii) attaching a completed Proposed Nominee questionnaire (which questionnaire shall be provided by the Corporation, upon request, to the stockholder providing the notice and shall include all information relating to the Proposed Nominee that would be required to be disclosed in connection with the solicitation of proxies for the election of the Proposed Nominee as a director in an election contest (even if an election contest is not involved), or would otherwise be required in connection with such solicitation, in each case pursuant to Regulation 14A (or any successor provision) under the Exchange Act and the rules thereunder, or would be required pursuant to the rules of any national securities exchange or over-the-counter market).
- (5) Notwithstanding anything to the contrary in Section 11(a), in the event that the number of directors to be elected to the Board of Directors is increased, and there is no public announcement of such action at least 130 days prior to the first anniversary of the date of the proxy statement for the preceding year's annual meeting, a stockholder's notice required by this Section 11(a) shall also be considered timely, but only with respect to nominees for any new positions created by such increase, if it shall be delivered to the secretary at the principal executive office of the Corporation not later than 5:00 p.m., Eastern Time, on the tenth day following the day on which such public announcement is first made by the Corporation.
- (6) For purposes of this <u>Section 11</u>, "<u>Stockholder Associated Person</u>" of any stockholder means (i) any person acting in concert with such stockholder, (ii) any beneficial owner of shares of stock of the Corporation owned of record or beneficially by such stockholder (other than a stockholder that is a depositary) and (iii) any person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, such stockholder or such Stockholder Associated Person.
- (b) Special Meetings of Stockholders. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting pursuant to the Corporation's notice of meeting. Nominations of individuals for election to the Board of Directors may be made at a special meeting of stockholders at which directors are to be elected only (i) by or at the direction of the Board of Directors or (ii) provided that the special meeting has been called in accordance with Section 3 of this Article II for the purpose of electing directors, by any stockholder of the Corporation who is a stockholder of record both at the time of giving of notice provided for in this Section 11 and at the time of the special meeting, who is entitled to vote at the meeting in the election of each individual so nominated and who has complied with the notice procedures set forth in this Section 11. In the event the Corporation calls a special meeting of stockholders for the purpose of electing one or more individuals to the Board of Directors, any such stockholder may nominate an individual or individuals (as the case may be) for election as a director as specified in the Corporation's notice of meeting, if the stockholder's notice, containing the information required by Section 11(a)(3), shall be delivered to the secretary at the principal executive office of the Corporation not earlier than the 120th day prior to such special meeting and not later than 5:00 p.m., Eastern Time, on the later of the 90th day prior to such special meeting or the tenth day following the day on which public announcement is first made of the date of the special meeting and of the nominees proposed by the Board of Directors to be elected at such meeting. The public announcement of a postponement or adjournment of a special meeting shall not commence a new time period for the giving of a stockholder's notice as described above.
- (c) <u>General</u>. (1) If information submitted pursuant to this <u>Section 11</u> by any stockholder proposing a nominee for election as a director or any proposal for other business at a meeting of stockholders shall be inaccurate in any material respect, such information may be deemed not to have been provided in accordance with this <u>Section 11</u>. Any such stockholder shall notify the Corporation of any inaccuracy or change (within two Business Days of becoming aware of such inaccuracy or change) in any such information. Upon written request by the secretary of the Corporation or the Board of Directors, any such stockholder shall provide, within five Business Days of delivery of such request (or such other period as may be specified in such request), (A) written verification, satisfactory, in the discretion of the Board of Directors or any authorized officer of the Corporation, to demonstrate the accuracy of any information submitted by the stockholder pursuant

to this <u>Section 11</u>, and (B) a written update of any information submitted by the stockholder pursuant to this <u>Section 11</u> as of an earlier date. If a stockholder fails to provide such written verification or written update within such period, the information as to which written verification or a written update was requested may be deemed not to have been provided in accordance with this <u>Section 11</u>.

- (2) Only such individuals who are nominated in accordance with this Section 11 shall be eligible for election by stockholders as directors, and only such business shall be conducted at a meeting of stockholders as shall have been brought before the meeting in accordance with this Section 11. The chairman of the meeting shall have the power to determine whether a nomination or any other business proposed to be brought before the meeting was made or proposed, as the case may be, in accordance with this Section 11.
- (3) "Public announcement" shall mean disclosure (i) in a press release reported by the Dow Jones News Service, Associated Press, Business Wire, PR Newswire or other widely circulated news or wire service or (ii) in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to the Exchange Act.
- (4) Notwithstanding the foregoing provisions of this Section 11, a stockholder shall also comply with all applicable requirements of state law and of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Section 11. Nothing in this Section 11 shall be deemed to affect any right of a stockholder to request inclusion of a proposal in, or the right of the Corporation to omit a proposal from, the Corporation's proxy statement pursuant to Rule 14a-8 (or any successor provision) under the Exchange Act. Nothing in this Section 11 shall require disclosure of revocable proxies received by the stockholder or Stockholder Associated Person pursuant to a solicitation of proxies after the filing of an effective Schedule 14A by such stockholder Or Stockholder Associated Person under Section 14(a) of the Exchange Act.
- Section 12. <u>TELEPHONE MEETINGS</u>. The Board of Directors or chairman of the meeting may permit one or more stockholders to participate in meetings of the stockholders by means of a conference telephone or other communications equipment by which all persons participating in the meeting can hear each other at the same time. Participation in a meeting by these means constitutes presence in person at the meeting.
- Section 13. <u>CONTROL SHARE ACQUISITION ACT</u>. Notwithstanding any other provision of the Charter or these Bylaws, Title 3, Subtitle 7 of the Maryland General Corporation Law, or any successor statute (the "<u>MGCL</u>"), shall not apply to any acquisition by any person of shares of stock of the Corporation. This section may be repealed, in whole or in part, at any time, whether before or after an acquisition of control shares and, upon such repeal, may, to the extent provided by any successor bylaw, apply to any prior or subsequent control share acquisition.
- Section 14. STOCKHOLDERS' CONSENT IN LIEU OF MEETING. Any action required or permitted to be taken at any meeting of stockholders may be taken without a meeting (a) if a unanimous consent setting forth the action is given in writing or by electronic transmission by each stockholder entitled to vote on the matter and filed with the minutes of proceedings of the stockholders or (b) if the action is advised, and submitted to the stockholders for approval, by the Board of Directors and a consent in writing or by electronic transmission of stockholders entitled to cast not less than the minimum number of votes that would be necessary to authorize or take the action at a meeting of stockholders is delivered to the Corporation in accordance with the MGCL. The Corporation shall give notice of any action taken by less than unanimous consent to each stockholder not later than ten days after the effective time of such action.

ARTICLE III DIRECTORS

Section 1. <u>GENERAL POWERS</u>. The business and affairs of the Corporation shall be managed under the direction of its Board of Directors

Section 2. <u>NUMBER, TENURE AND QUALIFICATIONS</u>. At any regular meeting or at any special meeting called for that purpose, a majority of the entire Board of Directors may establish, increase or decrease the number of directors, provided that the number thereof shall never be less than the minimum number required by the MGCL, nor more than 15, and further provided that the tenure of office of a director shall not be affected by any decrease in the number of directors. Any director of the Corporation may resign at any time by delivering his or her resignation to the Board of Directors, the chairman of the board or the secretary. Any resignation shall take effect immediately upon its receipt or at such later time specified in the resignation. The acceptance of a resignation shall not be necessary to make it effective unless otherwise stated in the resignation.

Section 3. <u>ANNUAL AND REGULAR MEETINGS</u>. An annual meeting of the Board of Directors shall be held immediately after and at the same place as the annual meeting of stockholders, no notice other than this Bylaw being

necessary. In the event such meeting is not so held, the meeting may be held at such time and place as shall be specified in a notice given as hereinafter provided for special meetings of the Board of Directors. The Board of Directors may provide, by resolution, the time and place for the holding of regular meetings of the Board of Directors without other notice than such resolution.

Section 4. <u>SPECIAL MEETINGS</u>. Special meetings of the Board of Directors may be called by or at the request of the chairman of the board, the chief executive officer, the president or by a majority of the directors then in office. The person or persons authorized to call special meetings of the Board of Directors may fix any place as the place for holding any special meeting of the Board of Directors called by them. The Board of Directors may provide, by resolution, the time and place for the holding of special meetings of the Board of Directors without other notice than such resolution.

Section 5. Notice of any special meeting of the Board of Directors shall be delivered personally or by telephone, electronic mail, facsimile transmission, courier or United States mail to each director at his or her business or residence address. Notice by personal delivery, telephone, electronic mail or facsimile transmission shall be given at least 24 hours prior to the meeting. Notice by United States mail shall be given at least three days prior to the meeting. Notice by courier shall be given at least two days prior to the meeting. Telephone notice shall be deemed to be given when the director or his or her agent is personally given such notice in a telephone call to which the director or his or her agent is a party. Electronic mail notice shall be deemed to be given upon transmission of the message to the electronic mail address given to the Corporation by the director. Facsimile transmission notice shall be deemed to be given upon completion of the transmission of the message to the number given to the Corporation by the director and receipt of a completed answer-back indicating receipt. Notice by United States mail shall be deemed to be given when deposited in the United States mail properly addressed, with postage thereon prepaid. Notice by courier shall be deemed to be given when deposited with or delivered to a courier properly addressed. Neither the business to be transacted at, nor the purpose of, any annual, regular or special meeting of the Board of Directors need be stated in the notice, unless specifically required by statute or these Bylaws.

Section 6. <u>QUORUM</u>. A majority of the directors shall constitute a quorum for transaction of business at any meeting of the Board of Directors, provided that, if less than a majority of such directors is present at such meeting, a majority of the directors present may adjourn the meeting from time to time without further notice, and provided further that if, pursuant to applicable law, the Charter or these Bylaws, the vote of a majority or other percentage of a particular group of directors is required for action, a quorum must also include a majority of such group.

The directors present at a meeting which has been duly called and at which a quorum has been established may continue to transact business until adjournment, notwithstanding the withdrawal from the meeting of enough directors to leave fewer than required to establish a quorum.

Section 7. <u>VOTING</u>. The action of a majority of the directors present at a meeting at which a quorum is present shall be the action of the Board of Directors, unless the concurrence of a greater proportion is required for such action by applicable law, the Charter or these Bylaws. If enough directors have withdrawn from a meeting to leave fewer than required to establish a quorum, but the meeting is not adjourned, the action of the majority of that number of directors necessary to constitute a quorum at such meeting shall be the action of the Board of Directors, unless the concurrence of a greater proportion is required for such action by applicable law, the Charter or these Bylaws.

Section 8. <u>ORGANIZATION</u>. At each meeting of the Board of Directors, the chairman of the board or, in the absence of the chairman, the vice chairman of the board, if any, shall act as chairman of the meeting. In the absence of both the chairman and vice chairman of the board, the chief executive officer or, in the absence of the chief executive officer, the president or, in the absence of the president, a director chosen by a majority of the directors present, shall act as chairman of the meeting. The secretary or, in his or her absence, an assistant secretary of the Corporation, or, in the absence of the secretary and all assistant secretaries, a person appointed by the chairman of the meeting, shall act as secretary of the meeting.

Section 9. <u>TELEPHONE MEETINGS</u>. Directors may participate in a meeting by means of a conference telephone or other communications equipment if all persons participating in the meeting can hear each other at the same time. Participation in a meeting by these means shall constitute presence in person at the meeting.

Section 10. <u>CONSENT BY DIRECTORS WITHOUT A MEETING</u>. Any action required or permitted to be taken at any meeting of the Board of Directors may be taken without a meeting, if a consent in writing or by electronic transmission to such action is given by each director and is filed with the minutes of proceedings of the Board of Directors.

Section 11. <u>VACANCIES</u>. If for any reason any or all the directors cease to be directors, such event shall not terminate the Corporation or affect these Bylaws or the powers of the remaining directors hereunder. Except as may be provided by the Board of Directors in setting the terms of any class or series of preferred stock, any vacancy on the Board of

Directors may be filled only by a majority of the remaining directors, even if the remaining directors do not constitute a quorum. Any director elected to fill a vacancy shall serve for the remainder of the full term of the class in which the vacancy occurred and until a successor is elected and qualifies.

Section 12. <u>COMPENSATION</u>. Directors shall not receive any stated salary for their services as directors but, by resolution of the Board of Directors, may receive compensation per year and/or per meeting and/or per visit to real property or other facilities owned or leased by the Corporation and for any service or activity they performed or engaged in as directors. Directors may be reimbursed for expenses of attendance, if any, at each annual, regular or special meeting of the Board of Directors or of any committee thereof and for their expenses, if any, in connection with each property visit and any other service or activity they performed or engaged in as directors; but nothing herein contained shall be construed to preclude any directors from serving the Corporation in any other capacity and receiving compensation therefor.

Section 13. <u>RELIANCE</u>. Each director and officer of the Corporation shall, in the performance of his or her duties with respect to the Corporation, be entitled to rely on any information, opinion, report or statement, including any financial statement or other financial data, prepared or presented by an officer or employee of the Corporation whom the director or officer reasonably believes to be reliable and competent in the matters presented, by a lawyer, certified public accountant or other person, as to a matter which the director or officer reasonably believes to be within the person's professional or expert competence, or, with respect to a director, by a committee of the Board of Directors on which the director does not serve, as to a matter within its designated authority, if the director reasonably believes the committee to merit confidence.

Section 14. <u>RATIFICATION</u>. The Board of Directors or the stockholders may ratify and make binding on the Corporation any action or inaction by the Corporation or its officers to the extent that the Board of Directors or the stockholders could have originally authorized the matter. Moreover, any action or inaction questioned in any stockholders' derivative proceeding or any other proceeding on the ground of lack of authority, defective or irregular execution, adverse interest of a director, officer or stockholder, non-disclosure, miscomputation, the application of improper principles or practices of accounting, or otherwise, may be ratified, before or after judgment, by the Board of Directors or by the stockholders, and if so ratified, shall have the same force and effect as if the questioned action or inaction had been originally duly authorized, and such ratification shall be binding upon the Corporation and its stockholders and shall constitute a bar to any claim or execution of any judgment in respect of such questioned action or inaction.

Section 15. <u>CERTAIN RIGHTS OF DIRECTORS AND OFFICERS</u>. A director who is not also an officer of the Corporation shall have no responsibility to devote his or her full time to the affairs of the Corporation. Any director or officer, in his or her personal capacity or in a capacity as an affiliate, employee, or agent of any other person, or otherwise, may have business interests and engage in business activities similar to, in addition to or in competition with those of or relating to the Corporation.

Section 16. EMERGENCY PROVISIONS. Notwithstanding any other provision in the Charter or these Bylaws, this Section 16 shall apply during the existence of any catastrophe, or other similar emergency condition, as a result of which a quorum of the Board of Directors under Article III of these Bylaws cannot readily be obtained (an "Emergency"). During any Emergency, unless otherwise provided by the Board of Directors, (i) a meeting of the Board of Directors or a committee thereof may be called by any director or officer by any means feasible under the circumstances; (ii) notice of any meeting of the Board of Directors during such an Emergency may be given less than 24 hours prior to the meeting to as many directors and by such means as may be feasible at the time, including publication, television or radio, and (iii) the number of directors necessary to constitute a quorum shall be one-third of the entire Board of Directors.

ARTICLE IV COMMITTEES

Section 1. <u>NUMBER, TENURE AND QUALIFICATIONS</u>. The Board of Directors may appoint from among its members an Executive Committee, an Audit Committee, a Compensation Committee, a Nominating and Corporate Governance Committee and one or more other committees, composed of one or more directors, to serve at the pleasure of the Board of Directors.

Section 2. <u>POWERS</u>. The Board of Directors may delegate to committees appointed under<u>Section 1</u> of this Article any of the powers of the Board of Directors, except as prohibited by law.

Section 3. <u>MEETINGS</u>. Notice of committee meetings shall be given in the same manner as notice for special meetings of the Board of Directors. A majority of the members of the committee shall constitute a quorum for the transaction of business at any meeting of the committee. The act of a majority of the committee members present at a meeting shall be the act of such committee. The Board of Directors may designate a chairman of any committee, and such chairman or, in the absence of a chairman, any two members of any committee (if there are at least two members of the Committee) may fix the

time and place of its meeting unless the Board shall otherwise provide. In the absence of any member of any such committee, the members thereof present at any meeting, whether or not they constitute a quorum, may appoint another director to act in the place of such absent member

Section 4. <u>TELEPHONE MEETINGS</u>. Members of a committee of the Board of Directors may participate in a meeting by means of a conference telephone or other communications equipment if all persons participating in the meeting can hear each other at the same time. Participation in a meeting by these means shall constitute presence in person at the meeting.

Section 5. C<u>ONSENT BY COMMITTEES WITHOUT A MEETING</u> Any action required or permitted to be taken at any meeting of a committee of the Board of Directors may be taken without a meeting, if a consent in writing or by electronic transmission to such action is given by each member of the committee and is filed with the minutes of proceedings of such committee.

Section 6. <u>VACANCIES</u>. Subject to the provisions hereof, the Board of Directors shall have the power at any time to change the membership of any committee, to fill any vacancy, to designate an alternate member to replace any absent or disqualified member or to dissolve any such committee.

ARTICLE V OFFICERS

Section 1. GENERAL PROVISIONS. The officers of the Corporation shall include a president, a secretary and a treasurer and may include a chairman of the board, a vice chairman of the board, a chief executive officer, one or more vice presidents, a chief operating officer, a chief financial officer, a chief investment officer, one or more assistant secretaries and one or more assistant treasurers. In addition, the Board of Directors may from time to time elect such other officers with such powers and duties as they shall deem necessary or desirable. The officers of the Corporation shall be elected annually by the Board of Directors, except that the chief executive officer or president may from time to time appoint one or more vice presidents, assistant secretaries and assistant treasurers or other officers. Each officer shall serve until his or her successor is elected and qualifies or until his or her death, or his or her resignation or removal in the manner hereinafter provided. Any two or more offices except president and vice president may be held by the same person. Election of an officer or agent shall not of itself create contract rights between the Corporation and such officer or agent.

Section 2. <u>REMOVAL AND RESIGNATION</u>. Any officer or agent of the Corporation may be removed, with or without cause, by the Board of Directors if in its judgment the best interests of the Corporation would be served thereby, but such removal shall be without prejudice to the contract rights, if any, of the person so removed. Any officer of the Corporation may resign at any time by delivering his or her resignation to the Board of Directors, the chairman of the board, the president or the secretary. Any resignation shall take effect immediately upon its receipt or at such later time specified in the resignation. The acceptance of a resignation shall not be necessary to make it effective unless otherwise stated in the resignation. Such resignation shall be without prejudice to the contract rights, if any, of the Corporation.

Section 3. VACANCIES. A vacancy in any office may be filled by the Board of Directors for the balance of the term.

Section 4. <u>CHIEF EXECUTIVE OFFICER</u>. The Board of Directors may designate a chief executive officer. In the absence of such designation, the chairman of the board shall be the chief executive officer of the Corporation. The chief executive officer shall have general responsibility for implementation of the policies of the Corporation, as determined by the Board of Directors, and for the management of the business and affairs of the Corporation. He or she may execute any deed, mortgage, bond, contract or other instrument, except in cases where the execution thereof shall be expressly delegated by the Board of Directors or by these Bylaws to some other officer or agent of the Corporation or shall be required by law to be otherwise executed; and in general shall perform all duties incident to the office of chief executive officer and such other duties as may be prescribed by the Board of Directors from time to time.

Section 5. <u>CHIEF OPERATING OFFICER</u>. The Board of Directors may designate a chief operating officer. The chief operating officer shall have the responsibilities and duties as determined by the Board of Directors or the chief executive officer.

Section 6. <u>CHIEF FINANCIAL OFFICER</u>. The Board of Directors may designate a chief financial officer. The chief financial officer shall have the responsibilities and duties as determined by the Board of Directors or the chief executive officer.

Section 7. <u>CHIEF INVESTMENT OFFICER</u>. The Board of Directors may designate a chief investment officer. The chief investment officer shall have the responsibilities and duties as determined by the Board of Directors or the chief executive officer.

Section 8. <u>CHAIRMAN OF THE BOARD</u>. The Board of Directors shall designate a chairman of the board. The chairman of the board shall preside over the meetings of the Board of Directors and of the stockholders at which he or she shall be present. The chairman of the board shall perform such other duties as may be assigned to him or her by the Board of Directors.

Section 9. <u>PRESIDENT</u>. In the absence of a chief executive officer, the president shall in general supervise and control all of the business and affairs of the Corporation. In the absence of a designation of a chief operating officer by the Board of Directors, the president shall be the chief operating officer. He or she may execute any deed, mortgage, bond, contract or other instrument, except in cases where the execution thereof shall be expressly delegated by the Board of Directors or by these Bylaws to some other officer or agent of the Corporation or shall be required by law to be otherwise executed; and in general shall perform all duties incident to the office of president and such other duties as may be prescribed by the Board of Directors from time to time.

Section 10. <u>VICE PRESIDENTS</u>. In the absence of the president or in the event of a vacancy in such office, the vice president (or in the event there be more than one vice president, the vice presidents in the order designated at the time of their election or, in the absence of any designation, then in the order of their election) shall perform the duties of the president and when so acting shall have all the powers of and be subject to all the restrictions upon the president; and shall perform such other duties as from time to time may be assigned to such vice president by the president or by the Board of Directors. The Board of Directors may designate one or more vice presidents as executive vice president, senior vice president, or as vice president for particular areas of responsibility.

Section 11. <u>SECRETARY</u>. The secretary shall: (a) keep the minutes of the proceedings of the stockholders, the Board of Directors and committees of the Board of Directors in one or more books provided for that purpose; (b) see that all notices are duly given in accordance with the provisions of these Bylaws or as required by law; (c) be custodian of the corporate records and of the seal of the Corporation; (d) keep a register of the post office address of each stockholder which shall be furnished to the secretary by such stockholder; (e) have general charge of the stock transfer books of the Corporation; and (f) in general perform such other duties as from time to time may be assigned to him or her by the chief executive officer, the president or by the Board of Directors.

Section 12. <u>TREASURER</u>. The treasurer shall have the custody of the funds and securities of the Corporation and shall keep full and accurate accounts of receipts and disbursements in books belonging to the Corporation and shall deposit all moneys and other valuable effects in the name and to the credit of the Corporation in such depositories as may be designated by the Board of Directors. In the absence of a designation of a chief financial officer by the Board of Directors, the treasurer shall be the chief financial officer of the Corporation.

The treasurer shall disburse the funds of the Corporation as may be ordered by the Board of Directors, taking proper vouchers for such disbursements, and shall render to the president and Board of Directors, at the regular meetings of the Board of Directors or whenever it may so require, an account of all his or her transactions as treasurer and of the financial condition of the Corporation.

Section 13. <u>ASSISTANT SECRETARIES AND ASSISTANT TREASURERS</u>. The assistant secretaries and assistant treasurers, in general, shall perform such duties as shall be assigned to them by the secretary or treasurer, respectively, or by the president or the Board of Directors.

Section 14. <u>COMPENSATION</u>. The compensation of the officers shall be fixed from time to time by or under the authority of the Board of Directors and no officer shall be prevented from receiving such compensation by reason of the fact that he or she is also a director.

ARTICLE VI

CONTRACTS, LOANS, CHECKS AND DEPOSITS

Section 1. <u>CONTRACTS</u>. The Board of Directors or any manager of the Corporation approved by the Board of Directors and acting within the scope of its authority pursuant to a management agreement with the Corporation may authorize any officer or agent to enter into any contract or to execute and deliver any instrument in the name of and on behalf of the Corporation and such authority may be general or confined to specific instances. Any agreement, deed, mortgage, lease or other document shall be valid and binding upon the Corporation when executed by an authorized person and duly authorized or ratified by action of the Board of Directors or a manager acting within the scope of its authority pursuant to a management agreement.

Section 2. <u>CHECKS AND DRAFTS</u>. All checks, drafts or other orders for the payment of money, notes or other evidences of indebtedness issued in the name of the Corporation shall be signed by such officer or agent of the Corporation in such manner as shall from time to time be determined by the Board of Directors.

Section 3. <u>DEPOSITS</u>. All funds of the Corporation not otherwise employed shall be deposited or invested from time to time to the credit of the Corporation as the Board of Directors, the chief executive officer, the chief financial officer, or any other officer designated by the Board of Directors may determine.

ARTICLE VII

STOCK

Section 1. <u>CERTIFICATES</u>. Except as may be otherwise provided by the Board of Directors, stockholders of the Corporation are not entitled to certificates representing the shares of stock held by them. In the event that the Corporation issues shares of stock represented by certificates, such certificates shall be in such form as prescribed by the Board of Directors or a duly authorized officer, shall contain the statements and information required by the MGCL and shall be signed by the officers of the Corporation in the manner permitted by the MGCL. In the event that the Corporation issues shares of stock without certificates, to the extent then required by the MGCL, the Corporation shall provide to the record holders of such shares a written statement of the information required by the MGCL to be included on stock certificates. There shall be no differences in the rights and obligations of stockholders based on whether or not their shares are represented by certificates.

Section 2. <u>TRANSFERS</u>. All transfers of shares of stock shall be made on the books of the Corporation, by the holder of the shares, in person or by his or her attorney, in such manner as the Board of Directors or any officer of the Corporation may prescribe and, if such shares are certificated, upon surrender of certificates duly endorsed. The issuance of a new certificate upon the transfer of certificated shares is subject to the determination of the Board of Directors that such shares shall no longer be represented by certificates. Upon the transfer of uncertificated shares, to the extent then required by the MGCL, the Corporation shall provide to record holders of such shares a written statement of the information required by the MGCL to be included on stock certificates.

The Corporation shall be entitled to treat the holder of record of any share of stock as the holder in fact thereof and, accordingly, shall not be bound to recognize any equitable or other claim to or interest in such share or on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise expressly provided by the laws of the State of Maryland.

Notwithstanding the foregoing, transfers of shares of any class or series of stock will be subject in all respects to the Charter and all of the terms and conditions contained therein.

Section 3. <u>REPLACEMENT CERTIFICATE</u>. Any officer of the Corporation may direct a new certificate or certificates to be issued in place of any certificate or certificates theretofore issued by the Corporation alleged to have been lost, destroyed, stolen or mutilated, upon the making of an affidavit of that fact by the person claiming the certificate to be lost, destroyed, stolen or mutilated; provided, however, if such shares have ceased to be certificated, no new certificate shall be issued unless requested in writing by such stockholder and the Board of Directors has determined that such certificates may be issued. Unless otherwise determined by an officer of the Corporation, the owner of such lost, destroyed, stolen or mutilated certificate or certificates, or his or her legal representative, shall be required, as a condition precedent to the issuance of a new certificate or certificates, to give the Corporation a bond in such sums as it may direct as indemnity against any claim that may be made against the Corporation.

Section 4. <u>FIXING OF RECORD DATE</u>. The Board of Directors may set, in advance, a record date for the purpose of determining stockholders entitled to notice of or to vote at any meeting of stockholders or determining stockholders entitled to receive payment of any dividend or the allotment of any other rights, or in order to make a determination of stockholders for any other proper purpose. Such date, in any case, shall not be prior to the close of business on the day the record date is fixed and shall be not more than 90 days and, in the case of a meeting of stockholders, not less than ten days, before the date on which the meeting or particular action requiring such determination of stockholders of record is to be held or taken.

When a record date for the determination of stockholders entitled to notice of and to vote at any meeting of stockholders has been set as provided in this section, such record date shall continue to apply to the meeting if adjourned or postponed, except if the meeting is adjourned to a date more than 120 days or postponed to a date more than 90 days after the record date originally fixed for the meeting, in which case a new record date for such meeting may be determined as set forth herein.

Section 5. <u>STOCK LEDGER</u>. The Corporation shall maintain at its principal office or at the office of its counsel, accountants or transfer agent, an original or duplicate stock ledger containing the name and address of each stockholder and the number of shares of each class held by such stockholder.

Section 6. FRACTIONAL STOCK; ISSUANCE OF UNITS. The Board of Directors may issue fractional stock or provide for the issuance of scrip, all on such terms and under such conditions as they may determine. Notwithstanding any other provision of the Charter or these Bylaws, the Board of Directors may issue units consisting of different securities of the Corporation. Any security issued in a unit shall have the same characteristics as any identical securities issued by the Corporation, except that the Board of Directors may provide that for a specified period securities of the Corporation issued in such unit may be transferred on the books of the Corporation only in such unit.

ARTICLE VIII

ACCOUNTING YEAR

The Board of Directors shall have the power, from time to time, to fix the fiscal year of the Corporation by a duly adopted resolution.

ARTICLE IX

DISTRIBUTIONS

Section 1. <u>AUTHORIZATION</u>. Dividends and other distributions upon the stock of the Corporation may be authorized by the Board of Directors, subject to the provisions of law and the Charter. Dividends and other distributions may be paid in cash, property or stock of the Corporation, subject to the provisions of law and the Charter.

Section 2. <u>CONTINGENCIES</u>. Before payment of any dividends or other distributions, there may be set aside out of any assets of the Corporation available for dividends or other distributions such sum or sums as the Board of Directors may from time to time, in its absolute discretion, think proper as a reserve fund for contingencies, for equalizing dividends, for repairing or maintaining any property of the Corporation or for such other purpose as the Board of Directors shall determine, and the Board of Directors may modify or abolish any such reserve.

ARTICLE X

INVESTMENT POLICY

Subject to the provisions of the Charter, the Board of Directors may from time to time adopt, amend, revise or terminate any policy or policies with respect to investments by the Corporation as it shall deem appropriate in its sole discretion.

ARTICLE XI

SEAL

Section 1. <u>SEAL</u>. The Board of Directors may authorize the adoption of a seal by the Corporation. The seal shall contain the name of the Corporation and the year of its incorporation and the words "Incorporated Maryland." The Board of Directors may authorize one or more duplicate seals and provide for the custody thereof.

Section 2. <u>AFFIXING SEAL</u>. Whenever the Corporation is permitted or required to affix its seal to a document, it shall be sufficient to meet the requirements of any law, rule or regulation relating to a seal to place the word "(SEAL)" adjacent to the signature of the person authorized to execute the document on behalf of the Corporation.

ARTICLE XII

INDEMNIFICATION AND ADVANCE OF EXPENSES

To the maximum extent permitted by Maryland law in effect from time to time, the Corporation shall indemnify and, without requiring a preliminary determination of the ultimate entitlement to indemnification, shall pay or reimburse reasonable expenses in advance of final disposition of a proceeding to (a) any individual who is a present or former director or officer of the Corporation and who is made or threatened to be made a party to the proceeding by reason of his or her service in that capacity or (b) any individual who, while a director or officer of the Corporation and at the request of the Corporation, serves or has served as a director, officer, partner or trustee of another corporation, real estate investment trust, partnership, joint venture, trust, employee benefit plan or other enterprise and who is made or threatened to be made a party to the proceeding by reason of his or her service in that capacity. The rights to indemnification and advance of expenses provided by the Charter and these Bylaws shall vest immediately upon election of a director or officer. The Corporation may,

with the approval of its Board of Directors, provide such indemnification and advance for expenses to an individual who served a predecessor of the Corporation in any of the capacities described in (a) or (b) above and to any employee or agent of the Corporation or a predecessor of the Corporation. The indemnification and payment or reimbursement of expenses provided in these Bylaws shall not be deemed exclusive of or limit in any way other rights to which any person seeking indemnification or payment or reimbursement of expenses may be or may become entitled under any bylaw, regulation, insurance, agreement or otherwise.

Neither the amendment nor repeal of this Article, nor the adoption or amendment of any other provision of the Bylaws or the Charter inconsistent with this Article, shall apply to or affect in any respect the applicability of the preceding paragraph with respect to any act or failure to act which occurred prior to such amendment, repeal or adoption.

ARTICLE XIII

WAIVER OF NOTICE

Whenever any notice of a meeting is required to be given pursuant to the Charter or these Bylaws or pursuant to applicable law, a waiver thereof in writing, signed by the person or persons entitled to such notice, whether before or after the time stated therein, shall be deemed equivalent to the giving of such notice. Neither the business to be transacted at nor the purpose of any meeting need be set forth in the waiver of notice, unless specifically required by statute. The attendance of any person at any meeting shall constitute a waiver of notice of such meeting, except where such person attends a meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened.

ARTICLE XIV

AMENDMENT OF BYLAWS

The Board of Directors shall have the exclusive power to adopt, alter or repeal any provision of these Bylaws and to make new Bylaws.

SUPPLEMENT & AMENDMENT TO WARRANT AGREEMENT

This Supplement and Amendment to the Warrant Agreement, dated as of November 6, 2009 (the "<u>Amendment</u>"), is executed by Enterprise Acquisition Corp., a Delaware corporation (the "<u>Company</u>"), ARMOUR Residential REIT, Inc. ("<u>ARMOUR</u>") and Continental Stock Transfer & Trust Company, a New York corporation (the "<u>Warrant Agent</u>").

WHEREAS, the Company and Warrant Agent are parties to that certain Warrant Agreement dated as of November 7, 2007 (the "Warrant Agreement"); and

WHEREAS, the parties desire to supplement and amend the Warrant Agreement upon the terms and conditions provided herein.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- 1. <u>Definitions.</u> Capitalized terms use herein and not otherwise herein shall have the meanings ascribed to them in the Warrant Agreement.
 - 2. <u>Amendment to Warrant Agreement.</u>
 - (a) Section 3.1 of the Warrant Agreement is hereby amended and restated in its entirety as follows:
 - "3.1. Warrant Price. Each Warrant shall, when countersigned by the Warrant Agent, entitle the registered holder thereof, subject to the provisions of such Warrant and of this Warrant Agreement, to purchase from the Company the number of shares of Common Stock stated therein at a Warrant Price of \$11.00, subject to the adjustments provided in Section 4 hereof and in the last sentence of this Section 3.1. The term "Warrant Price" as used in this Warrant Agreement refers to the price per share at which Common Stock may be purchased at the time a Warrant is exercised. The Company in its sole discretion may lower the Warrant Price at any time prior to the Expiration Date for a period of not less than ten business days; provided, however, that any such reduction shall be identical in percentage terms among all of the Warrants."
 - (b) Section 3.2 of the Warrant Agreement is hereby amended and restated in its entirety as follows:
 - "3.2. <u>Duration of Warrants</u>. A Warrant may be exercised only during the period ("Exercise Period") commencing on the consummation by the Company of a Business Combination (as such term is defined in the Company's Second Amended and Restated Certificate of Incorporation) and terminating at 5:00 p.m., New York City time, on the earlier to occur of (i) November 7, 2013 or (ii) the date fixed for redemption of the Warrants as provided in Section 6 of this Agreement ("Expiration Date"). Except with respect to the right to receive the Redemption Price (as set forth in Section 6 hereunder), each Warrant not exercised on or before the Expiration Date shall become void, and all rights thereunder and all rights in respect thereof under this Agreement shall cease at the close of business on the Expiration Date. The Company in its sole discretion may extend the duration of the Warrants by delaying the Expiration Da te; provided, however, that the Company will provide no less than twenty days' notice to registered holders of the Warrants of such extension."
 - (c) Section 3.3.5 of the Warrant Agreement is hereby amended and restated in its entirety as follows:
 - "3.3.5. <u>Limitations on Exercise</u>. Notwithstanding anything to the contrary contained herein, no Warrant may be exercised if it would cause the holder to beneficially own, within the meaning of ARMOUR's Articles of Amendment and Restatement, greater than 9.8% of the outstanding Common Stock."
 - (d) A new Section 4.4 of the Warrant Agreement is hereby amended and restated in its entirety as follows:
 - "4.4. Reorganization of Company. If the Company consolidates or merges with or into, or transfers or leases all or substantially all of its assets to any person, upon consummation of such transaction the Warrants shall automatically become exercisable for the kind and amount of securities, cash or other assets which the holder of a

Warrant would have owned immediately after the consolidation, merger, transfer or lease if such holder had exercised the Warrant immediately before the effective date of the transaction; provided that (i) if the holders of Common Stock were entitled to exercise a right of election as to the kind or amount of securities, cash or other assets receivable upon such consolidation or merger, then the kind and amount of securities, cash or other assets for which each Warrant shall become exercisable shall be deemed to be the weighted average of the kind and amount received per share by the holders of Common Stock in such consolidation or merger that affirmatively make such election or (ii) if a tender or exchange offer shall have been made to and accepted by the holders of Common Stock under circumstances in which, upon completion of such tender or exchange offer, the maker thereof, together with members of any gro up (within the meaning of Rule 13d-5(b) (1) under the Exchange Act) of which such maker is a part, and together with any affiliate or associate of such maker (within the meaning of Rule 12b-2 under the Exchange Act) and any members of any such group of which any such affiliate or associate is a part, own beneficially (within the meaning of Rule 13d-3 under the Exchange Act) more than 50% of the outstanding shares of Common Stock, the holder of a Warrant shall be entitled to receive the highest amount of cash, securities or other property to which such holder would actually have been entitled as a shareholder if such Warrant holder had exercised the Warrant prior to the expiration of such tender or exchange offer, accepted such offer and all of the Common Stock held by such holder had been purchased pursuant to such tender or exchange offer, subject to adjustments (from and after the consummation of such tender or exchange offer) as nearly equivalent as possible to the adjustments provided for in this Section 4. Immediately upon the consummation of a business combination between the Company and ARMOUR Residential REIT, Inc. ("ARMOUR"), (i) each holder of a Warrant shall be entitled to receive a new Warrant representing the right to purchase one share of ARMOUR's common stock, (ii) all references to the "Company" in this Agreement shall mean ARMOUR and (iii) ARMOUR shall assume all of the rights and all of the obligations of the Company under this Agreement. If ARMOUR subsequently consolidates or merges with or into, or transfers or leases all or substantially all its assets to, any person, upon consummation of such transaction, concurrently with the consummation of such transaction, the corporation or other entity formed by or surviving any such consolidation or merger if other than the Company, or the person to which such sale or conveyance shall have been made, shall enter into a supplemental Warrant Agreement so providing and further providing for adjustments which shall be as nearly equiva lent as may be practical to the adjustments provided for in this Section 4. The successor Company shall mail to Warrant holders a notice describing the supplemental Warrant Agreement.

If the issuer of securities deliverable upon exercise of Warrants under the supplemental Warrant Agreement is an affiliate of the formed, surviving, transferee or lessee corporation, that issuer shall join in the supplemental Warrant Agreement."

- 3. <u>Amendment.</u> All references in the Warrant Agreement (and in the other agreements, documents and instruments entered into in connection therewith) to the "Warrant Agreement" shall be deemed for all purposes to refer to the Warrant Agreement, as amended by this Amendment.
- 4. <u>Remaining Provisions of Warrant Agreement</u>. Except as expressly provided herein, the provisions of the Warrant Agreement shall remain in full force and effect in accordance with their terms and shall be unaffected by this Amendment.
- 5. <u>Counterparts</u>. This Amendment may be executed in counterparts, each of which when executed shall be deemed an original and both of which when executed shall be deemed one and the same instrument.
- 6. <u>Headings</u>. The headings to this Amendment are for ease of reference only and shall not limit or otherwise affect the meaning hereof.
- 7. <u>Governing Law.</u> This Amendment shall be governed by, and construed in accordance with, the laws of the State of New York, without regard to the principles of conflicts of law of any jurisdiction.
- 8. <u>Effective Time</u>. This Amendment shall be effective immediately prior to the consummation of a business combination between the Company and ARMOUR.

[SIGNATURE PAGE TO FOLLOW]

IN WITNESS WHEREOF, this Amendment has been duly executed and delivered by the authorized officers of each of the undersigned as of the date first above written.

ENTERPRISE ACQUISITION CORP.

By: /s/ Ezra Shashoua Name: Ezra Shashoua Title: Chief Financial Officer

ARMOUR RESIDENTIAL REIT, INC.

By: <u>/s/ Jeffrey J. Zimmer</u> Name: Jeffrey J. Zimmer

Title: Co-Chief Executive Officer

CONTINENTAL STOCK TRANSFER & TRUST CO.

By: <u>/s/ Steve Nelson</u> Name: Steve Nelson Title: Chairman

[Signature page to Supplement and Amendment to Warrant Agreement]

ESCROW TERMINATION AGREEMENT

Enterprise Acquisition Corporation 6800 Broken Sound Parkway Suite 200 Boca Raton, Florida 33487 July 29, 2009

Continental Stock Transfer & Trust Company 17 Battery Place New York New York 10004 Attn: Steven Nelson

Re: Trust Account No. Termination Letter

Gentlemen:

Pursuant to paragraph 1(i) of the Investment Management Trust Agreement between Enterprise Acquisition Corp. ("Company") and Continental Stock Transfer & Trust Company ("Trustee"), dated as of November 7, 2007 ("Trust Agreement"), this is to advise you that the Company has entered into an agreement ("Business Agreement") with ARMOUR Residential REIT, Inc. ("Target Business") to consummate a business combination with Target Business ("Business Combination") on or prior to November 7, 2009. The Company shall notify you at least 48 hours in advance of the actual date of the consummation of the Business Combination ("Consummation Date").

In accordance with the terms of the Trust Agreement, we hereby authorize you to commence liquidation of the Trust Account to the effect that, on the Consummation Date, all of funds held in the Trust Account will be immediately available for transfer to the account or accounts that the Company shall direct on the Consummation Date.

On the Consummation Date (i) counsel for the Company shall deliver to you written notification that the Business Combination has been consummated ("Counsel's Letter"), (ii) the Company shall deliver to you (a) an affiliate or certificate of its Corporate Secretary which verifies the vote of the Company's stockholders in connection with the Business Combination and (b) written instructions (the "Instruction Letter") with respect to the transfer of the funds held in the Trust Account other than the Deferred Discount, in an amount to be mutually agreed upon by the Company and the Representatives and so directed by them (the "Adjusted Deferred Discount") and (iii) the Representatives shall deliver to you written instructions for delivery of the Adjusted Deferred Discount. You are hereby directed and authorized to transfer the funds held in the Trust Account immediately upon your receipt of the Counsel's Letter and the Instruction Letter, (a) to the Representatives, the Adjusted Deferred Discount and (b) the remainder in accordance with the terms of the Instruction Letter. In the event that certain deposits held in the Trust Account may not be liquidated by the Consummation Date without penalty, you will notify the Company of the same and the Company shall direct you as to whether such funds should remain in the Trust Account and be distributed after the Consummation Date to the Company. Upon the distribution of all the funds in the Trust Account pursuant to the terms hereof, the Trust Agreement shall be terminated and the Trust Account closed

In the event that the Business Combination is not consummated on the Consummation Date described in the notice thereof and we have not notified you on or before the original Consummation Date of a new Consummation Date, then upon the Trustee's receipt of a written request from the Company, the funds held in the Trust Account shall be reinvested as provided in the Trust Agreement on the business day immediately following the original Consummation Date as set forth in the notice.

Very truly yours,

ENTERPRISE ACQUISITION CORP.

By: /s/ Marc H. Bell

Marc H. Bell, Chairman of the Board

By: /s/ Maria Balodimas Staton

Maria Balodimas Staton, Corporate Secretary

cc: UBS Investment Bank Ladenburg Thalmann & Co. Inc.

[Signature page to Escrow Termination Agreement]

MANAGEMENT AGREEMENT

This **MANAGEMENT AGREEMENT** is entered into as of November 6, 2009 by and between (i) ARMOUR RESIDENTIAL REIT, INC., a Maryland corporation (the "<u>REIT</u>"), and (ii) ARMOUR RESIDENTIAL MANAGEMENT LLC, a Delaware limited liability company (the "<u>Manager</u>").

RECITALS

WHEREAS, the REIT intends to use the net proceeds of borrowings and securities offerings and the net returns on its investments which are not otherwise distributed to stockholders (i) in Mortgage Assets (as defined below), and (ii) in any such other assets, in a manner which allows the REIT to qualify as a "real estate investment trust" under the Code (as defined below); and

WHEREAS, the REIT desires that the Manager undertake, on the REIT's behalf, the duties and responsibilities as set forth in this Agreement, subject to the direction of the Manager or, only where applicable and only if and when any of the stock of the REIT becomes publicly traded, subject to the direction and oversight of the Board of Directors (as defined below), on the terms and conditions set forth in this Agreement; and

WHEREAS, the Manager desires to undertake, on the REIT's behalf, the duties and responsibilities as set forth in this Agreement on the terms and conditions set forth in this Agreement; and

WHEREAS, the REIT and the Manager desire to state in its entirety the management agreement by and between the REIT and the Manager;

NOW, **THEREFORE**, in consideration of the premises and of the mutual covenants and agreement contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

- 1. <u>Definitions.</u> Capitalized terms used but not otherwise defined in this Agreement shall have the respective meanings assigned to them below:
- 1.1 "Affiliate" means, with respect to any specified Person, any other Person that directly or indirectly controls, is controlled by, or is under common control with, that specified Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlling", "controlled by" and "under common control with"), with respect to any specified Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of that specified Person, whether by contract, through the ownership of voting securities or other equity interests (including partnership or membership interests), or otherwise.
 - 1.2 "Agreement" means this Management Agreement, as the same may be amended from time to time.
 - 1.3 "Annual Minimum Fee" means \$900,000 for each fiscal year of this Agreement.
 - 1.4 "Base Management Fee" shall have the meaning set forth in Section 6.1 of this Agreement.
- 1.5 "Board of Directors" means the member(s) of the Board of Directors of the REIT, applicable if and when any of the stock of the REIT becomes publicly traded.
- 1.6 "Business Day" means a day on which the banks are opened for business (Saturdays, Sundays, statutory and civic holidays excluded) in New York, New York, United States.
- 1.7 "Cause" means, for purposes of a termination of this Agreement by the REIT without penalty or payment of a Termination Fee, a final determination by a court of competent jurisdiction (a) that the Manager has materially breached this Agreement that has a material adverse effect on the REIT and such material breach has continued for a period of 30 days after receipt by the Manager of written notice thereof specifying such breach and requesting that the same be remedied in such 30-day period, (b) that an action taken or omitted to be taken by the Manager in connection with this Agreement constitutes willful misconduct or gross negligence that results in material harm to the REIT and such willful misconduct or gross negligence has not been cured within a period of 30 days after receipt by the Manager of written notice thereof specifying such willful misconduct or gross negligence and requesting that the same be remedied in such 30-day period, or (c) that an action taken or omitted to be taken by the Manager in connection with this Agreement constitutes fraud that results in material harm to the REIT.

- 1.8 "Code" means the Internal Revenue Code of 1986, as amended.
- 1.9 "Effective Date" means the date of the consummation of the Merger.
- 1.10 "Governing Instruments" means the articles of incorporation or charter, as the case may be, and the bylaws of the REIT and its subsidiaries, as those documents may be amended from time to time.
- 1.11 "Gross Equity Raised" means an amount in dollars calculated as of the date of determination that is equal to (a) the initial equity capital of the REIT following the consummation of the Merger, plus (b) equity capital raised in public or private issuances of the REIT's equity securities (calculated before underwriting fees and distribution expenses, if any), less (c) capital returned to the stockholders of the REIT, as adjusted to exclude (d) one-time charges pursuant to changes in GAAP and certain non-cash charges after discussion between the Manager and the Board of Directors and approved by a majority of the Board of Directors, if and when any of the stock of the REIT becomes publicly traded.
- 1.12 "Independent Directors" means the members of the Board of Directors who are not officers or employees of the Manager or any Person directly or indirectly controlling or controlled by the Manager, and who are otherwise "independent" in accordance with the REIT's Governing Instruments and policies and, if applicable, the rules of any national securities exchange on which the REIT's common stock is listed.
 - 1.13 "Initial Term" shall have the meaning set forth in Section 10.1 of this Agreement.
 - 1.14 "Investment Company Act" shall mean the Investment Company Act of 1940, as amended.
- 1.15 "Manager" shall have the meaning set forth in the Preamble of this Agreement and shall include any successor thereto (subject to the provisions of Section 13).
- 1.16 "Manager Obligations" shall have the meaning set forth in Section 2.4.2 of this Agreement and may be limited from time to time in the REIT's discretion.
 - 1.17 "Manager Shareholders" shall have the meaning set forth in Section 2.5 of this Agreement.
 - 1.18 "Merger" means the merger contemplated pursuant to the Merger Agreement.
- 1.19 "Merger Agreement" means that Agreement and Plan of Merger, dated as of July 29, 2009, among the REIT, ARMOUR Merger Sub Corp., a Delaware corporation, and Enterprise Acquisition Corp., a Delaware corporation.
- 1.20 "Mortgage Assets" means the following assets types of the REIT which the REIT may determine from time to time shall be solely managed by the Manager:
 - (i) mortgage securities (or interests therein), including (a) adjustable-rate, hybrid adjustable-rate and pass-through certificates (including GNMA certificates, FNMA certificates and FHLMC certificates), collateralized mortgage obligations, (c) securities representing interests in, or secured by, agency wrapped mortgages on real property other than pass-through certificates and CMOs, (d) agency mortgage derivative securities and other agency mortgage-backed and mortgage collateralized obligations, and (e) mortgage derivative securities;
 - (ii) U.S. government issued bills, notes and bonds including general obligations of the agencies of the U.S. government (including, but not limited to GNMA, FNMA and FHLMC); and
 - (iii) short-term investments, including short-term bank certificates of deposit, short-term U.S. Treasury securities, short-term U.S. government agency securities, commercial paper, repurchase agreements, short-term CMOs, short-term asset backed securities and other similar types of short-term investment instruments, all of which will have maturities or average lives of less than one (1) year.
 - 1.21 "Non-Renewal Notice" shall have the meaning set forth in Section 10.1 of this Agreement.
 - 1.22 "Notice of Proposal to Negotiate" shall have the meaning set Forth in Section 10.5 of this Agreement.
- 1.23 "Person" means any individual, corporation, partnership, joint venture, limited liability company, estate, trust, unincorporated association, any federal, state, county or municipal government or any bureau, department or agency thereof and any fiduciary acting in such capacity on behalf of any of the foregoing.
 - 1.24 "Real Estate Investment Trust" means a "real estate investment trust" as defined under the Code.

- 1.25 "REIT" shall have the meaning set forth in the Preamble of this Agreement and shall include any subsidiary and any successor thereto.
 - 1.26 "REIT Provisions of the Code" means Sections 856 through 860 of the Code.
 - 1.27 "Renewal Term" shall have the meaning set forth in Section 10.1 of this Agreement.
 - 1.28 "Staton Bell" shall have the meaning set forth in Section 2.5 of this Agreement.
 - 1.29 "Sub-Management Agreement" shall have the meaning set forth in Section 2.5 of this Agreement.
- 1.30 "Termination Fee" means an amount equal to three (3) times the Base Management Fee paid to the Manager in the preceding full twelve (12) months, calculated as of the effective date of the termination of this Agreement pursuant to Section 10.2.

General Duties of the Manager.

- 2.1 Services. Until any of the stock of the REIT becomes publicly traded, all services performed by the Manager under this Agreement shall be under the direction of the Manager. If and when any of the stock of the REIT becomes publicly traded, all services performed by the Manager under this Agreement shall be subject to the direction and oversight of the Board of Directors. As may be limited from time to time by the REIT in its discretion, the Manager shall (i) manage the day-to-day operations of the REIT and perform the services and other activities described below, and (ii) to the extent directed by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), perform similar management and services for any subsidiary of the REIT; provided, however, that nothing herein shall give the Manager the right (or obligate the Manager) to supervise any other manager engaged by the REIT (each such other manager, an "Other Manager"), or to manage or otherwise participate in any way in any securitization transaction undertaken by the REIT on any joint venture formed by the REIT. Subject to the REIT's right to retain Other Managers and the REIT's right to limit the following duties in its discretion from time to time to the Mortgage Assets which the REIT determines from time to time shall be solely managed by the Manager, the Manager shall perform the following services from time to time as may be required for the management of the REIT and its assets (other than any such assets solely being managed by an Other Manager):
 - 2.1.1 serving as a consultant to the REIT with respect to the formulation of investment criteria for assets managed by the Manager and the preparation of policy guidelines by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) for such assets;
 - 2.1.2 assisting the REIT in developing criteria for Mortgage Asset purchase commitments that are consistent with the REIT's long-term investment objectives and making available to the REIT its knowledge and experience with respect to Mortgage Assets managed by the Manager;
 - 2.1.3 representing the REIT in connection with certain of the REIT's purchases, sales and commitments to purchase or sell Mortgage Assets managed by the Manager that meet in all material respects the REIT's investment criteria, including without limitation by providing repurchase agreement and similar portfolio management expertise as appropriate in connection therewith;
 - 2.1.4 managing the REIT's Mortgage Assets (other than any Mortgage Assets managed solely by Other Managers);
 - 2.1.5 advising the REIT and negotiating the REIT's agreements with third-party lenders for borrowings by the REIT;
 - 2.1.6 making available to the REIT statistical and economic research and analysis regarding the REIT's activities managed by the Manager and the services performed for the REIT by the Manager;
 - 2.1.7 monitoring and providing to the Board of Director, if and when any of the stock of the REIT becomes publicly traded, from time to time price information and other data obtained from certain nationally-recognized dealers that maintain markets in mortgage assets identified by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) from time to time, and providing data and advice to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) in connection with the identification of such dealers, in each case with respect to assets managed by the Manager;
 - 2.1.8 investing or reinvesting money of the REIT, which the REIT determines from time to time shall be solely managed by the Manager, in accordance with the REIT's policies and procedures;

- 2.1.9 providing executive and administrative personnel, office space and other appropriate services required in rendering services to the REIT, in accordance with and subject to the terms of this Agreement;
- 2.1.10 administering the day-to-day operations of the REIT and performing and supervising the performance of such other administrative functions necessary to the management of the REIT as may be agreed upon by the Manager and the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), including, without limitation, the collection of revenues and the payment of the REIT's debts and obligations from the REIT's accounts (in each case in respect of assets managed by the Manager), and the maintenance of appropriate computer systems and related information technology to perform such administrative and management functions;
- 2.1.11 advising the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) in connection with certain policy decisions (other than any such decisions solely relating to Other Managers);
- 2.1.12 evaluating and recommending hedging strategies to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) and, upon approval by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), engaging in hedging activities on behalf of the REIT consistent with the REIT's status as a Real Estate Investment Trust, in each case in respect of assets managed by the Manager;
- 2.1.13 supervising compliance by the REIT with the REIT Provisions of the Code and maintenance of its status as a Real Estate Investment Trust (other than in respect of any assets not managed by the Manager);
- 2.1.14 qualifying and causing the REIT to qualify to do business in all applicable jurisdictions and obtaining and maintaining all appropriate licenses (other than in respect of any activities not managed by the Manager);
- 2.1.15 assisting the REIT to retain qualified accountants and tax experts to assist in developing and monitoring appropriate accounting procedures and testing systems and to conduct quarterly compliance reviews as the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) may deem necessary or advisable (other than any such procedures or reviews relating solely to Other Managers);
- 2.1.16 assisting the REIT in its compliance with all federal (including, without limitation, the Sarbanes-Oxley Act of 2002), state and local regulatory requirements applicable to the REIT in respect of its business activities, including preparing or causing to be prepared all financial statements required under applicable regulations and contractual undertakings and all reports, documents and filings, if any, required under the Securities Exchange Act of 1934, as amended, or other federal or state laws;
- 2.1.17 assisting the REIT in its compliance with federal, state and local tax filings and reports, and generally enable the REIT to maintain its status as a Real Estate Investment Trust, including soliciting stockholders, as defined below, for required information to the extent provided in the REIT Provisions of the Code;
- 2.1.18 assisting the REIT in its maintenance of an exemption from the Investment Company Act and monitoring compliance with the requirements for maintaining an exemption from the Investment Company Act;
- 2.1.19 advising the REIT as to its capital structure and capital raising activities (other than in respect of capital not to be managed by the Manager);
- 2.1.20 handling and resolving all claims, disputes or controversies (including all litigation, arbitration, settlement or other proceedings or negotiations) in which the REIT may be involved or to which the REIT may be subject arising out of the REIT's day-to-day operations, subject to the approval of the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) and excluding any such proceedings or negotiations solely involving Other Managers;
- 2.1.21 engaging and supervising, on behalf of the REIT at the REIT's request and at the REIT's expense, the following, without limitation: independent contractors to provide investment banking services, leasing services, mortgage brokerage services, securities brokerage services, other financial services and such other services as may be deemed by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) to be necessary or advisable from time to time (other than Other Managers, or any of the foregoing to be utilized in connection with activities being solely conducted by Other Managers);

- 2.1.22 so long as the Manager does not incur additional costs or expenses, and the REIT does not incur additional costs or expenses which are not specifically approved in writing by the REIT, performing such other services as may be necessary or advisable from time to time for management and other activities relating to the assets of the REIT as the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) shall reasonably request or the Manager shall deem appropriate under the particular circumstances; and
- 2.1.23 assisting the REIT, upon the REIT's request therefor, in evaluating the advantages and disadvantages of the REIT internalizing the functions of the Manager or of any merger and acquisition transaction that the REIT may elect to pursue, which also may be subject to approval by the shareholders of the REIT.

2.2 Obligations of the Manager.

- 2.2.1 <u>Verify Conformity with Acquisition Criteria</u>. At all times (and, if and when any of the stock of the REIT becomes publicly traded, subject to the direction of the Board of Directors), the Manager shall use commercially reasonable efforts to provide that each Mortgage Asset acquired by the Manager for the REIT conforms in all material respects to the acquisition criteria of the REIT and shall seek to cause each seller or transferor of such Mortgage Assets to the REIT to make such representations and warranties regarding such Mortgage Assets as may, in the reasonable judgment of the Manager, be necessary and appropriate, subject to market custom. In addition, the Manager shall take such other action as it deems reasonably necessary or appropriate in seeking to protect the REIT's investments to the extent consistent with its duties under this Agreement.
- 2.2.2 Conduct Activities in Conformity with REIT Status and All Applicable Restrictions. At all times (and, if and when any of the stock of the REIT becomes publicly traded, subject to the direction of the Board of Directors) and with reasonable advance notice from the REIT of any pertinent information relating to any activities of the REIT as may then be conducted by Other Managers, the Manager shall refrain from any action which would adversely affect the status of the REIT or, if applicable, any subsidiary of the REIT as a Real Estate Investment Trust or (i) which would violate any material law, rule or regulation of any governmental body or agency having jurisdiction over the REIT or any such subsidiary or (ii) which would otherwise not be permitted by the REIT's or such subsidiary's Governing Instruments, any material operating policies adopted by the REIT, or any agreements actually known by the Manager, except in each of clauses (i) and (ii) as could not reasonably be expected to have a material adverse effect on the REIT. If the Manager is directed to take any such action by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), the Manager's judgment that such action would adversely affect such status or cause such violation or not be permitted as aforesaid.
- 2.2.3 Reports. If and when any of the stock of the REIT becomes publicly traded and upon the request of the Board of Directors and at the sole cost and expense of the REIT, the Manager shall cause an annual compliance report of the REIT to be prepared by a firm independent of the Manager and its Affiliates and having the proper expertise to determine compliance with the REIT Provisions of the Code and related matters. In addition, the Manager shall prepare regular reports for the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) that will review the REIT's acquisitions of Mortgage Assets, portfolio composition and characteristics, credit quality (if applicable), performance and compliance with the REIT's investment policies and policies that enable the REIT to maintain its qualification as a Real Estate Investment Trust and to maintain its exemption from being deemed an "investment company" under the Investment Company Act; provided that such reports shall only relate to assets the REIT has determined shall be managed by the Manager.
- 2.2.4 <u>Portfolio Transactions.</u> In placing portfolio transactions and selecting brokers or dealers, the Manager shall seek to obtain on behalf of the REIT commercially reasonable terms. In assessing commercially reasonable terms for any transaction, the Manager shall consider all factors it deems relevant, including, without limitation, the breadth of the market for the security, the price of the security, the financial condition and execution capability of the broker or dealer, and the reasonableness of the commission, if any, both for the specific transaction and on a continuing basis.
- 2.3 Cooperation of the REIT. The REIT (and, if and when any of the stock of the REIT becomes publicly traded, the Board of Directors) shall take such actions as may reasonably be required to permit and enable the Manager to carry out its duties and obligations under this Agreement, including, without limitation, the steps reasonably necessary to allow the Manager to file any registration statement on behalf of the REIT in a timely manner if the REIT requests that the Manager do so. The REIT further agrees to use commercially reasonable efforts to make available to the Manager reasonably available resources, information and materials reasonably requested by the Manager to enable the Manager to

satisfy its obligations hereunder, including its obligations to deliver financial statements and any other information or reports with respect to the REIT. If the Manager is not able to provide a service, or in the reasonable judgment of the Manager it is not prudent to provide a service, without the approval of the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), then the Manager shall be excused from providing such service (and shall not be in breach of this Agreement) until the applicable approval has been obtained; *provided*, *however*, that the Manager shall have promptly advised the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) in writing that the Manager is awaiting such approval.

2.4 Engagement of Third Parties.

- 2.4.1 Securities Dealers. Subject to the REIT's right to retain Other Managers and the REIT's right to limit the Manager's authorizations in the REIT's discretion from time to time, the Manager is authorized, for and on behalf, and at the sole cost and expense of the REIT, to employ such securities dealers (including Affiliates of the Manager) for the purchase and sale of the REIT's Mortgage Assets managed by the Manager as may, in the reasonable judgment of the Manager, be necessary to obtain the best commercially available net results taking into account such factors as the policies of the REIT, price, dealer spread, the size, type and difficulty of the transaction involved, the firm's general execution and operational facilities and the firm's risk in positioning the securities involved. Consistent with this policy, and subject to the foregoing caveats with respect to the REIT's rights, the Manager is authorized to direct the execution of the REIT's portfolio transactions to dealers and brokers furnishing statistical information or research deemed by the Manager to be reasonably necessary to the performance of its investment advisory functions for the REIT.
- 2.4.2 Other Third Parties. The Manager is authorized to retain, for and on behalf of the REIT, the services of third parties (including Affiliates of the Manager), including, without limitation, accountants, legal counsel, appraisers, insurers, brokers, dealers, transfer agents, registrars, developers, investment banks, financial advisors, banks and other lenders and others as the Manager deems reasonably necessary or advisable in connection with the management and operations of the REIT. The costs and expenses related to the retention of third parties shall be the sole cost and expense of the REIT except to the extent (i) the third party is retained to make decisions to invest in and dispose of Mortgage Assets, provide administrative, data processing or clerical services, prepare the financial records of the REIT or prepare a report summarizing the REIT's acquisitions of Mortgage Assets, portfolio compensation and characteristics, credit quality (if applicable) or performance of the portfolio, in each case with respect to assets the REIT has determined shall be managed by the Manager, in which case it shall be at the sole cost and expense of the Manager unless otherwise approved by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) or (ii) the costs and expenses are not reimbursable pursuant to Section 7.3 of this Agreement (collectively, the "Manager Obligations"). Notwithstanding anything in this Agreement to the contrary, in no event shall the Manager be responsible for any costs or expenses related to or incurred by any Other Manager.
- 2.4.3 Affiliates. Notwithstanding anything contained in this Agreement to the contrary, the Manager shall have the right to cause any of its services under this Agreement to be rendered by the Manager's employees or Affiliates of the Manager. The REIT shall pay or reimburse the Manager or its Affiliates (subject to the foregoing approval) for the reasonable and actually incurred cost and expense of performing such services by the Affiliate, including, without limitation, back office support services specifically requested by the REIT if the costs and expenses of such Affiliate would have been reimbursable under this Agreement if such Affiliate were an unaffiliated third party, or if such service had been performed by the Manager itself.
- 2.5 <u>Sub-Management Agreement</u>. The REIT and the Manager expressly acknowledge and agree that, concurrent with this Agreement, the Manager is entering into the Sub-Management Agreement, dated as of even date herewith, by and among the Manager, Staton Bell Blank Check LLC ("<u>Staton Bell</u>"), and Jeffrey J. Zimmer and Scott J. Ulm (Messrs. Zimmer and Ulm, together, the "<u>Manager Shareholders</u>") (such agreement, the "<u>Sub-Management Agreement</u>"), and nothing to the contrary contained in this Agreement shall limit the ability of the Manager, Staton Bell, or the Manager Shareholders to enter into and perform their respective obligations under such Sub-Management Agreement or otherwise limit the effectiveness of such Sub-Management Agreement. The REIT represents and warrants that the Sub-Management Agreement has been duly authorized and approved by all necessary action of the REIT.

3. Additional Activities.

3.1 Other Activities of the Manager. Nothing in this Agreement shall (i) prevent the Manager or its Affiliates, officers, directors or employees, from engaging in other businesses or from rendering services of any kind to any other person or entity, including, without limitation, investing in, or rendering advisory service to others investing in, any type of mortgage assets or other real estate investments (including, without limitation, investments that meet the principal investment objectives of the REIT), whether or not the investment objectives or policies of any such other person or entity are similar to

those of the REIT, or (ii) in any way bind or restrict the Manager or its Affiliates, officers, directors or employees from buying, selling or trading any securities or commodities for their own accounts or for the account of others for whom the Manager or its Affiliates, officers, directors or employees may be acting. The REIT acknowledges that the Manager will base allocation decisions on the procedures the Manager and the REIT reasonably and in good faith consider fair and equitable, including, without limitation, such considerations as investment objectives, restrictions and time horizon, availability of cash and the amount of existing holdings. While information and recommendations supplied to the REIT shall, in the Manager's reasonable and good faith judgment, be appropriate under the circumstances and in light of the investment objectives and policies of the REIT, they may be different from the information and recommendations supplied by the Manager or any Affiliate of the Manager to other investment companies, funds and advisory accounts. The REIT shall be entitled to equitable treatment under the circumstances in receiving information, recommendations and any other services. However, the REIT recognizes that it is not entitled to receive preferential treatment as compared with the treatment given by the Manager or any Affiliate of the Manager to any investment company, fund or advisory account other than any fund or advisory account which contains *only* funds invested by the Manager (and *not* of any of its clients or customers) or its officers and directors.

- 3.2 Other Activities of the REIT. Except to the extent expressly set forth in this Agreement or any other written agreement between the REIT and the Manager, neither this Agreement nor the relationship between the REIT and the Manager shall be deemed (i) to limit or restrict the activities of the REIT, its officers, its employees, or members of its Board of Directors (if and when any of the stock of the REIT becomes publicly traded), or (ii) impose a fee or other penalty on the REIT, its officers, its employees, or members of its Board of Directors (if and when any of the stock of the REIT becomes publicly traded) for pursuing any such other activities.
- 3.3 Service to the REIT; Execution of Documents Directors, officers, employees and agents of the Manager and its Affiliates may serve as trustees, directors, officers, employees, agents, nominees or signatories for the REIT or any subsidiary of the REIT, to the extent permitted by the Governing Instruments, as from time to time amended, or by any resolutions duly adopted by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) pursuant to the Governing Instruments. When executing documents or otherwise acting in such capacities for the REIT, such persons shall use their respective titles in the REIT.
- 4. <u>Bank Accounts</u>. The Manager may establish and maintain one or more bank accounts in the name of the REIT or any subsidiary of the REIT, and may collect and deposit into any such account or accounts, and disburse funds from any such account or accounts in a manner consistent with this Agreement, including, without limitation, the following: (a) the payment of the Base Management Fee, (b) the payment (or advance) of reimbursable costs and expenses, and (c) such other amounts. The Manager shall from time to time render appropriate accountings of such collections and payments to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) and, upon request (whether or not the REIT is publicly traded), to the auditors of the REIT or any subsidiary of the REIT. One or more of the obligations of the Manager hereunder may be revoked in whole or in part by the REIT from time to time in its sole discretion.
- 5. Records; Confidentiality. The Manager shall maintain appropriate and accurate books of account and records relating to services performed under this Agreement, and such books of account and records shall be accessible for inspection by representatives (including the auditors) of the REIT or any subsidiary of the REIT at any time during normal business hours. Except in the ordinary course of business of the REIT, the Manager shall, and shall use commercially reasonable efforts to cause each of its Affiliates to, keep confidential any and all information they (or such Affiliates) may obtain from time to time in connection with the services they (or such Affiliates) render under this Agreement.
 - 6. <u>Compensation of the Manager</u>.
- Base Management Fee. For services rendered under this Agreement, commencing after the end of the first month of business, the REIT shall pay to the Manager each month in arrears (by wire transfer of immediately available funds) compensation equal to 1/12th of the sum of (a) 1.5% of the Gross Equity Raised up to \$1 billion plus (b) 0.75% of the Gross Equity Raised in excess of \$1 billion (the "Base Management Fee") within one (1) Business Day after the end of such month; provided, however, that the Base Management Fee shall not ever be less than 1/12th of the Annual Minimum Fee. In the event of a termination of this Agreement during a calendar month, the Base Management Fee shall be pro-rated based upon the number of days elapsed in such calendar month prior to the effective date of such termination.
 - 6.2 <u>No Incentive Management Compensation.</u> The Manager shall not receive any incentive-based compensation.
 - 7. <u>Expenses of the Manager and the REIT.</u>
 - 7.1 <u>Expenses of the Manager</u>. The Manager shall be responsible for the following expenses:

- 7.1.1 employment expenses of the personnel employed by the Manager, including, without limitation, salaries (base and bonuses alike), wages, payroll taxes and the cost of employee benefit plans of such personnel (but excluding any stock of the REIT that the Board of Directors, if and when any of the stock of the REIT becomes publicly traded, may determine to grant to such personnel, which stock shall not reduce employment expenses otherwise payable by the Manager pursuant to this Section 7.1.1 or cause the Manager or the REIT to pay any payroll taxes in respect thereof); and
- 7.1.2 rent, telephone, utilities, office furniture, equipment, machinery and other office, internal and overhead expenses of the Manager required for the REIT's day-to-day operations, including, bookkeeping, clerical and back-office services provided by the Manager, *provided*, *however*, that the REIT shall pay for supplies applicable to operations (paper, software, presentation materials, etc.).
- 7.2 Expenses of the REIT. The REIT shall pay all of the costs and expenses of the REIT and the Manager incurred solely on behalf of the REIT or any subsidiary or in connection with this Agreement, other than (i) those expenses that are specifically the responsibility of the Manager pursuant to Section 7.1 of this Agreement, and (ii) any costs or expenses incurred by the Manager which the REIT is not required to reimburse pursuant to the provisions of Section 7.3 below. Without limiting the generality of the foregoing, it is specifically agreed that the following costs and expenses of the REIT or any subsidiary of the REIT shall be paid by the REIT and shall not be paid by the Manager and/or the Affiliates of the Manager (except to the extent of any costs or expenses which the REIT is not required to reimburse pursuant to the provisions of Section 7.3 below):
 - 7.2.1 all costs and expenses associated with the formation and capital raising activities of the REIT and its subsidiaries, including, without limitation, the costs and expenses of the preparation of the REIT's registration statements, and any and all costs and expenses of any public offering of the REIT, any subsequent offerings and any filing fees and costs of being a public company, including, without limitation, filings with the Securities and Exchange Commission, the Financial Industry Regulatory Authority, the New York Stock Exchange (and any other exchange or over-the-counter market), among other such entities;
 - 7.2.2 all costs and expenses of the REIT in connection with the acquisition, disposition, financing, hedging, administration and ownership of the REIT's or any subsidiary's investment assets (including, without limitation, the Mortgage Assets) and, including, without limitation, costs and expenses incurred in contracting with third parties, including Affiliates of the Manager (as may be approved by the REIT pursuant to the terms of this Agreement), to provide such services, such as legal fees, accounting fees, consulting fees, trustee fees, appraisal fees, insurance premiums, commitment fees, brokerage fees, guaranty fees, ad valorem taxes, costs of foreclosure, maintenance, repair and improvement of property and premiums for insurance on property owned by the REIT or any subsidiary of the REIT;
 - 7.2.3 all costs and expenses relating to the acquisition of, and maintenance and upgrades to, the REIT's portfolio analytics and accounting systems (including, but not limited to Bloomberg);
 - 7.2.4 all costs and expenses of money borrowed by the REIT or its subsidiaries, including, without limitation, principal, interest and the costs associated with the establishment and maintenance of any credit facilities, warehouse loans and other indebtedness of the REIT and its subsidiaries (including commitment fees, legal fees, closing and other costs);
 - 7.2.5 all taxes and license fees applicable to the REIT or any subsidiary of the REIT, including interest and penalties thereon;
 - 7.2.6 all legal, audit, accounting, underwriting, brokerage, listing, filing, rating agency, registration and other fees, printing, engraving, clerical, personnel and other expenses and taxes of the REIT incurred in connection with the issuance, distribution, transfer, registration and stock exchange listing of the REIT's or any subsidiary's equity securities or debt securities;
 - 7.2.7 other than for the Manager Obligations, all fees paid to and expenses of third-party advisors and independent contractors, consultants, managers and other agents (other than the Manager) engaged by the REIT or any subsidiary of the REIT or by the Manager for the account of the REIT or any subsidiary of the REIT (other than the Manager) and all employment expenses of the personnel employed by the REIT or any subsidiary of the REIT, including, without limitation, the salaries (base and bonuses alike), wages, equity based compensation of such personnel, and payroll taxes;

- 7.2.8 all insurance costs incurred by the REIT or any subsidiary of the REIT and including, but not limited to, insurance paid for by the REIT to insure the Manager for liabilities as a result of being the manager for the REIT;
 - 7.2.9 all custodian, transfer agent and registrar fees and charges incurred by the REIT;
- 7.2.10 all compensation and fees paid to directors of the REIT or any subsidiary of the REIT, all expenses of directors of the REIT or any subsidiary of the REIT (including those directors who are also employees of the Manager), the cost of directors and officers liability insurance and premiums for errors and omissions insurance, and any other insurance deemed necessary or advisable by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) for the benefit of the REIT and its directors and officers (including those directors who are also employees of the Manager), the cost of all meetings of the REIT's Board of Directors (if and when any of the stock of the REIT becomes publicly traded), and the cost of travel, hotel accommodations, food and entertainment for all participants in the meetings of the REIT's Board of Directors (if and when any of the stock of the REIT becomes publicly traded);
- 7.2.11 all third-party legal, accounting and auditing fees and expenses and other similar services relating to the REIT's or any subsidiary's operations (including, without limitation, all quarterly and annual audit or tax fees and expenses);
- 7.2.12 all legal, expert and other fees and expenses relating to any actions, proceedings, lawsuits, demands, causes of action and claims, whether actual or threatened, made by or against the REIT, or which the REIT is authorized or obligated to pay under applicable law or its Governing Instruments or by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded);
- 7.2.13 any judgment or settlement of pending or threatened proceedings (whether civil, criminal or otherwise) against the REIT or any subsidiary of the REIT, or against any trustee, director or officer of the REIT or any subsidiary of the REIT in his capacity as such for which the REIT or any subsidiary of the REIT is required to indemnify such trustee, director or officer by any court or governmental agency, or settlement of pending or threatened proceedings;
- 7.2.14 at all times all travel and related expenses of directors, officers and employees of the REIT and the Manager incurred in connection with meetings related to the business of the REIT, attending meetings of the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) or holders of securities of the REIT or any subsidiary of the REIT or performing other business activities that relate to the REIT or any subsidiary of the REIT, including, without limitation, travel and expenses incurred in connection with the purchase, financing, refinancing, sale or other disposition of Mortgage Assets or other investments of the REIT; provided, however, that the REIT shall only be responsible for a proportionate share of such expenses, as reasonably determined by the Manager in good faith after full disclosure to the REIT, in instances in which such expenses were not incurred solely for the benefit of the REIT;
- 7.2.15 all expenses of organizing, modifying or dissolving the REIT or any subsidiary of the REIT, costs preparatory to entering into a business or activity, and costs of winding up or disposing of a business or activity of the REIT or its subsidiaries:
- 7.2.16 all expenses relating to payments of dividends or interest or distributions in cash or any other form made or caused to be made by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) to or on account of holders of the securities of the REIT or any subsidiary of the REIT, including, without limitation, in connection with any dividend reinvestment plan;
- 7.2.17 all expenses of third parties relating to communications to holders of equity securities or debt securities issued by the REIT or any subsidiary of the REIT and the other bookkeeping and clerical work necessary in maintaining relations with holders of such securities and in complying with the continuous reporting and other requirements of governmental bodies or agencies, including any costs of computer services in connection with this function, the cost of printing and mailing certificates for such securities and proxy solicitation materials and reports to holders of the REIT's or any subsidiary's securities and reports to third parties required under any indenture to which the REIT or any subsidiary of the REIT is a party;
- 7.2.18 subject to Section 7.1, all expenses relating to any office or office facilities maintained by the REIT or any subsidiary of the REIT (exclusive of the office of the Manager and/or Affiliates of the Manager), including, without limitation, rent, telephone, utilities, office furniture, equipment, machinery and other office expenses for the REIT's chief financial officer and any other persons the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) authorizes the REIT to hire;

- 7.2.19 all costs and expenses related to the design and maintenance of the REIT's web site or sites and associated with any computer software or hardware that is used solely for the REIT:
- 7.2.20 other than for the Manager Obligations, all other costs and expenses relating to the REIT's business and investment operations, including, without limitation, the costs and expenses of acquiring, owning, protecting, maintaining, developing and disposing of Mortgage Assets, including, without limitation, appraisal, reporting, audit and legal fees;
- 7.2.21 other than for the Manager Obligations, and subject to a line item budget approved in advance by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), all other expenses actually incurred by the Manager, its Affiliates (as may be approved by the REIT pursuant to the terms of this Agreement) or their respective officers, employees, representatives or agents, or any Affiliates thereof (as may be approved by the REIT pursuant to the terms of this Agreement) which are reasonably necessary for the performance by the Manager of its duties and functions under this Agreement (including, without limitation, any fees or expenses relating to the REIT's compliance with all governmental and regulatory matters); and
- 7.2.22 all other expenses of the REIT or any subsidiary of the REIT that are not the responsibility of the Manager under Section 7.1 of this Agreement.
- 7.3 Expense Reimbursement to the Manager. Costs and expenses incurred by the Manager on behalf of the REIT or its subsidiaries shall be reimbursed in cash monthly to the Manager within five (5) Business Days of receipt by the REIT from the Manager of a statement of such costs and expenses. Cost and expense reimbursement to the Manager shall be subject to adjustment at the end of each calendar year in connection with the annual audit of the REIT.
 - Limits of Manager Responsibility: Indemnity.
- 8.1 <u>Limits of Manager Responsibility</u>. The Manager shall have the responsibility under this Agreement to render the services specifically called for under this Agreement and shall not be responsible for any action of the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) in following or declining to follow any advice or recommendations of the Manager, including, without limitation, as set forth in Section 2.2.2 of this Agreement. The Manager and its Affiliates, directors, officers, stockholders, equity holders, employees, representatives and agents, and any Affiliates thereof, shall not be liable to the REIT (including, without limitation, any stockholder thereof), any issuer of mortgage securities, any subsidiary of the REIT, its subsidiary's stockholders, the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), any credit-party, any counter-party under any agreement or any other person whatsoever for any acts or omissions, errors of judgment or mistakes of law by the Manager or its Affiliates, directors, officers, employees, representatives or agents, or any Affiliates thereof, under or in connection with this Agreement, except in the event that the Manager was grossly negligent, acted with reckless disregard or engaged in willful misconduct or fraud while discharging its duties under this Agreement.
- 8.2 Indemnification. The REIT and its subsidiaries shall reimburse, indemnify and hold harmless the Manager and its Affiliates, directors, officers, stockholders, equity holders, employees, representatives and agents, and any Affiliates thereof from and against any and all expenses, losses, costs, damages, liabilities, demands, charges and claims of any nature whatsoever, actual or threatened (including, without limitation, reasonable attorneys' fees), arising from or in respect of any acts or omissions, errors of judgment or mistakes of law (or any alleged acts or omissions, errors of judgment or mistakes of law) performed or made while acting in any capacity contemplated under this Agreement or pursuant to any underwriting agreement or similar agreement to which Manager is a party that is related to the REIT's activities. Notwithstanding the foregoing, the REIT shall have no indemnification obligation under this Section 8.2 in the event that the Manager was grossly negligent, acted with reckless disregard or engaged in willful misconduct or fraud while discharging its duties under this Agreement.
- 9. <u>No Joint Venture</u>. The REIT and the Manager are not partners or joint venturers with each other, and nothing in this Agreement shall be construed to make them such partners or joint venturers or impose any liability as such on any of them. The Manager is an independent contractor and, except as expressly provided or authorized in this Agreement, shall have no authority to act for or represent the REIT.
 - 10. <u>Effectiveness; Termination</u>.
- 10.1 <u>Effectiveness</u>. This Agreement shall commence on the Effective Date and shall continue in effect thereafter for an initial term of five (5) years (the "<u>Initial Term</u>"). Following the Initial Term, this Agreement shall automatically extend for successive one (1)-year terms (each, a "<u>Renewal Term</u>"), unless either party gives 180 days' written notice prior to the expiration of the Initial Term or any Renewal Term to the respective other party of such first party's intent

not to renew the then-current term (any such notice, a "Non-Renewal Notice"); provided, however, that if the REIT pays the Final Payment (as such term is defined in the Sub-Management Agreement) to Staton Bell pursuant to the terms of the Sub-Management Agreement, the then-current Renewal Term shall automatically be extended as necessary so that it expires one (1) year from the date on which such Final Payment was made; provided, further, that the REIT may give a Non-Renewal Notice to the Manager only if, if and when any of the stock of the REIT becomes publicly traded, at least two-thirds of all of the Independent Directors or the holders of a majority of the outstanding shares of common stock of the REIT (other than those shares held by the Manager or its Affiliates) agree that (i) there has been unsatisfactory performance by the Manager that is materially detrimental to the REIT and its subsidiaries or (ii) the compensation payable to the Manager hereunder is unfair; provided further, however, that in the event that the REIT gives a Non-Renewal Notice to the Manager under clause (ii) above, such Non-Renewal Notice, and its effectiveness, shall be subject to Section 10.5. This Agreement may be terminated during the Initial Term or any Renewal Term only in accordance with the provisions of Sections 10.2, 10.3 and 10.4 or 13.1 (as applicable).

10.2 <u>Early Termination without Cause</u>.

- 10.2.1 The REIT may not terminate the Agreement during the Initial Term, except for Cause. After the Initial Term, the REIT may terminate the agreement without Cause upon 180 days' prior written notice to the Manager and subject to payment of the Termination Fee pursuant to Section 10.4 (except as otherwise provided in Section 13.1).
- 10.2.2 The Manager may terminate the agreement at any time and for any reason upon 180 days' prior written notice to the REIT.
- 10.3 <u>Early Termination for Cause.</u> Notwithstanding the provisions of Section 10.2.1, or any other provision of this Agreement to the contrary, the REIT may terminate the agreement for Cause at any time and without paying any Termination Fee, effective immediately upon written notice.

10.4 Payments In Connection With Termination.

- 10.4.1 Payments By the REIT. Following any termination of this Agreement by the REIT or the Manager, the REIT shall pay the following amounts to the Manager (by wire transfer of immediately available funds to such bank account as is designated by the Manager to the REIT in writing) not later than five (5) Business Days after the effective date of such termination:
 - (i) all reimbursable costs and expenses permitted under the Agreement (to the extent not previously reimbursed to the Manager), if any, as of the date of the effectiveness of such termination of this Agreement; and
 - (ii) either (a) if this Agreement was terminated by the REIT for Cause pursuant to Section 10.3, any Base Management Fee due and not yet paid to the Manager, (as pro-rated pursuant to Section 6.1 through the date of the effectiveness of such termination of this Agreement) or (b) if this Agreement was terminated by the REIT without Cause pursuant to Section 10.2.1, and subject to the provisions of Section 13.1, the Termination Fee (as calculated through the effective date of such termination of the Agreement).
- 10.4.2 <u>Payments By the Manager</u>. For the avoidance of doubt, following any termination of this Agreement by the Manager, no fees or other payment shall be due from the Manager to the REIT except as otherwise expressly provided in this Agreement.
- 10.5 Renegotiation of Compensation. In the event that a Non-Renewal Notice is given by the REIT to the Manager in connection with a determination pursuant to clause (b)(ii) of Section 10.1 that the compensation payable to the Manager is unfair, the Manager shall have the right to renegotiate such compensation by delivering to the REIT, no fewer than 45 days prior to the prospective expiration of the Initial Term or Renewal Term then in effect, as applicable, written notice (any such notice, a "Notice of Proposal to Negotiate") of its intent to renegotiate its compensation under this Agreement. Thereupon, the REIT (represented by the Independent Directors, if and when any of the stock of the REIT becomes publicly traded) and the Manager shall endeavor to negotiate the revised compensation payable to the Manager under this Agreement. In the event that the Manager and the REIT, including, if and when any of the stock of the REIT becomes publicly traded, at least two-thirds of all of the Independent Directors, agree to the terms of the revised compensation to be payable to the Manager within 45 days following the receipt of the Notice of Proposal to Negotiate, the Non-Renewal Notice shall be deemed of no force and effect and this Agreement shall continue in full force and effect on the terms stated in this Agreement, except that the compensation payable to the Manager hereunder shall be the revised compensation then agreed upon by the parties to this Agreement. The REIT and the Manager agree to execute and deliver an amendment to this Agreement setting forth such revised compensation promptly upon reaching an agreement regarding same.

In the event that the REIT and the Manager are unable to agree to the terms of the revised compensation to be payable to the Manager during such 45-day period, this Agreement shall terminate, such termination to be effective on the expiration of the Initial Term or Renewal Term then in effect, as applicable.

- 11. <u>Action Upon Termination</u>. In connection with any termination of this Agreement, the Manager shall promptly:
- 11.1.1 pay over to the REIT or any subsidiary of the REIT all money collected and held for the account of the REIT or any subsidiary of the REIT by the Manager pursuant to this Agreement;
- 11.1.2 deliver to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) an accounting, including a statement showing all payments collected by it and a statement of all money held by it, covering the period following the date of the last accounting furnished to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) with respect to the REIT or any subsidiary of the REIT;
- 11.1.3 deliver to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) all property and documents of the REIT or any subsidiary of the REIT then in the custody of the Manager;
- 11.1.4 assign to the REIT any authorized agreements the Manager executed in its name on behalf of the REIT (and obtain the counter-parties' consent thereto); and
- 11.1.5 assign to the REIT all proprietary information with respect to the REIT, including, without limitation, software, models, intellectual property, licenses, tradenames and trademarks (but subject to the limitations set forth in Section 28 hereof).
- 12. <u>Survival of Obligations</u>. The REIT's obligation to make payments hereunder and the limitations set forth herein shall survive the termination of this Agreement. The covenants and agreements of the Manager contained herein (for expenses through the effective date of termination) shall survive the termination of this Agreement.

Assignments.

- 13.1 Assignment by the Manager. This Agreement shall terminate automatically in the event that the Manager assigns all or any part of this Agreement (including, without limitation, any transfer or assignment by operation of law), unless such assignment is consented to in advance in writing by the REIT, including, if and when any of the stock of the REIT becomes publicly traded, the Board of Directors. In the event an assignment by the Manager is consented to by the REIT, including, if and when any of the stock of the REIT becomes publicly traded, the Board of Directors in accordance with this Section 13.1, such assignment shall bind the assignee under this Agreement in the same manner as the Manager is bound, and the Manager shall be released from all of its obligations, duties and responsibilities under this Agreement and all liability therefore and in respect hereof accruing on or after that date. In addition, the assignee shall execute and deliver to the REIT a counterpart of this Agreement naming such assignee as Manager, and the REIT shall deliver to the assigning Manager a duly executed instrument evidencing the release of the assigning Manager from such obligations, duties and responsibilities as aforesaid. Notwithstanding the provisions of Section 10.2.1, or any other provision of this Agreement to the contrary, in the event that the REIT shall not have any payment obligations to such successor Manager other than to pay unpaid reimbursable costs and expenses pursuant to Section 10.4.1(i) and earned but unpaid Base Management Fee payments pursuant to Section 10.4.1(ii)(a).
- 14. Release of Money or Other Property Upon Written Request. The Manager agrees that any money or other property of the REIT or any subsidiary of the REIT held by the Manager under this Agreement shall be held by the Manager as custodian for the REIT or such subsidiary, and the Manager's records shall be appropriately marked clearly to reflect the ownership of such money or other property by the REIT or such subsidiary.
- 14.1 <u>Procedures.</u> Upon the receipt by the Manager of a written request signed by a duly authorized officer of the REIT or an authorized member of the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) requesting the Manager to release to the REIT or any subsidiary of the REIT any money or other property then held by the Manager for the account of the REIT or any subsidiary of the REIT under this Agreement, the Manager shall release such money or other property to the REIT or such subsidiary of the REIT within a reasonable period of time, but in no event later than the earlier to occur of (i) thirty (30) days following such request, or (ii) the date of the termination of this Agreement.
- 14.2 <u>Limitations</u>. The Manager and its Affiliates, directors, officers, stockholders, equity holders, employees, representatives and agents, and any Affiliates thereof, shall not be liable to the REIT, any subsidiaries of the REIT, the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) or the REIT's or its subsidiaries' stockholders for any acts performed or omissions to act by the REIT or any subsidiary of the REIT in connection with the

money or other property released to the REIT or any subsidiary of the REIT in accordance with this Section 14, except in the event that the Manager was grossly negligent, acted with reckless disregard or engaged in willful misconduct or fraud while discharging its duties under this Agreement.

- 14.3 <u>Indemnification</u>. The REIT and any subsidiary of the REIT shall indemnify the Manager and its Affiliates, directors, officers, stockholders, equity holders, employees, representatives and agents, and any Affiliates thereof, against any and all expenses, costs, losses, damages, liabilities, demands, charges and claims of any nature whatsoever, which arise in connection with the Manager's release of such money or other property to the REIT or any subsidiary of the REIT in accordance with the terms of this Section 14, except in the event that the Manager was grossly negligent, acted with reckless disregard or engaged in willful misconduct or fraud while discharging its duties under this Agreement. Indemnification pursuant to this provision shall be in addition to any right of the Manager and its Affiliates, directors, officers, stockholders, equity holders, employees, representatives and agents, and any Affiliates thereof, to indemnification under Section 8 of this Agreement.
 - Representations, Warranties and Covenants.
 - 15.1 <u>REIT in Favor of the Manager.</u> The REIT hereby represents and warrants to the Manager as follows:
 - 15.1.1 <u>Due Formation</u>. The REIT is duly organized, validly existing and in good standing under the laws of Maryland, has the power to own its assets and to transact the business in which it is now engaged and is duly qualified to do business and is in good standing under the laws of each jurisdiction where its ownership or lease of property or the conduct of its business requires such qualification, except for failures to be so qualified, authorized or licensed that could not in the aggregate have a material adverse effect on the business operations, assets or financial condition of the REIT and its subsidiaries, taken as a whole. The REIT does not do business under any fictitious business name.
 - 15.1.2 Power and Authority. The REIT has the power and authority to execute, deliver and perform this Agreement and all obligations required under this Agreement and has taken all necessary action to authorize this Agreement on the terms and conditions hereof and the execution, delivery and performance of this Agreement and all obligations required under this Agreement. Except as shall have been obtained, no consent of any other person, including, without limitation, stockholders and creditors of the REIT, and no license, permit, approval or authorization of, exemption by, notice or report to, or registration, filing or declaration with, any governmental authority is required by the REIT in connection with this Agreement or the execution, delivery, performance, validity or enforceability of this Agreement and all obligations required under this Agreement. This Agreement has been, and each instrument or document required under this Agreement will be, executed and delivered by a duly authorized officer of the REIT, and this Agreement constitutes, and each instrument or document required under this Agreement when executed and delivered under this Agreement will constitute, the legally valid and binding obligation of the REIT enforceable against the REIT in accordance with its terms.
 - 15.1.3 Execution, Delivery and Performance. The execution, delivery and performance of this Agreement and the documents or instruments required under this Agreement will not violate any provision of any existing law or regulation binding on the REIT, or any order, judgment, award or decree of any court, arbitrator or governmental authority binding on the REIT, or the Governing Instruments of, or any securities issued by, the REIT or of any mortgage, indenture, lease, contract or other agreement, instrument or undertaking to which the REIT is a party or by which the REIT or any of its assets may be bound, the violation of which would have a material adverse effect on the business operations, assets or financial condition of the REIT and its subsidiaries, taken as a whole, and will not result in, or require, the creation or imposition of any lien on any of its property, assets or revenues pursuant to the provisions of any such mortgage, indenture, lease, contract or other agreement, instrument or undertaking (other than the pledge of amounts payable to the Manager under this Agreement to secure the Manager's obligations to its lenders).
 - 15.2 <u>Manager in Favor of the REIT</u>. The Manager hereby represents and warrants to the REIT as follows:
 - 15.2.1 <u>Due Formation</u>. The Manager is duly organized, validly existing and in good standing under the laws of Delaware, has the power to own its assets and to transact the business in which it is now engaged and is duly qualified to do business and is in good standing under the laws of each jurisdiction where its ownership or lease of property or the conduct of its business requires such qualification, except for failures to be so qualified, authorized or licensed that could not in the aggregate have a material adverse effect on the business operations, assets or financial condition of the Manager and its subsidiaries, taken as a whole. The Manager does not do business under any fictitious business name.

- 15.2.2 Power and Authority. The Manager has the power and authority to execute, deliver and perform this Agreement and all obligations required under this Agreement and has taken all necessary corporate action to authorize this Agreement on the terms and conditions hereof and the execution, delivery and performance of this Agreement and all obligations required under this Agreement. Except as shall have been obtained, no consent of any other person including, without limitation, stockholders and creditors of the Manager, and no license, permit, approval or authorization of, exemption by, notice or report to, or registration, filing or declaration with, any governmental authority is required by the Manager in connection with this Agreement or the execution, delivery, performance, validity or enforceability of this Agreement and all obligations required under this Agreement. This Agreement has been and each instrument or document required under this Agreement will be executed and delivered by a duly authorized officer of the Manager, and this Agreement constitutes, and each instrument or document required under this Agreement when executed and delivered under this Agreement will constitute, the legally valid and binding obligation of the Manager enforceable against the Manager in accordance with its terms.
- 15.2.3 Execution, Delivery and Performance. The execution, delivery and performance of this Agreement and the documents or instruments required under this Agreement will not violate any provision of any existing law or regulation binding on the Manager, or any order, judgment, award or decree of any court, arbitrator or governmental authority binding on the Manager, or the governing instruments of, or any securities issued by, the Manager or of any mortgage, indenture, lease, contract or other agreement, instrument or undertaking to which the Manager is a party or by which the Manager or any of its assets may be bound, the violation of which would have a material adverse effect on the business operations, assets or financial condition of the Manager and its subsidiaries, taken as a whole, and will not result in, or require, the creation or imposition of any lien on any of its property, assets or revenues pursuant to the provisions of any such mortgage indenture, lease, contract or other agreement, instrument or undertaking.
- 15.2.4 <u>No Limitations</u>. The personnel of the Manager providing services to the REIT on the Manager's behalf pursuant to this Agreement will be free of legal and contractual impediments to their provision of services pursuant to the terms of this Agreement.
- 16. Notices. Unless expressly provided otherwise in this Agreement, all notices, requests, demands and other communications required or permitted under this Agreement shall be in writing and shall be deemed to have been duly given, made and received when (1) delivered by hand, (2) otherwise delivered by reputable overnight courier against receipt therefor, or (3) upon actual receipt of registered or certified mail, postage prepaid, return receipt requested. The parties may deliver to each other notice by electronically transmitted facsimile copies or electronically transmitted mail (i.e., e-mail), provided that such facsimile or e-mail notice is followed within 24 hours by any type of notice otherwise provided for in this Section 16. Any party may alter the address or other contact information to which communications or copies are to be sent by giving notice of such change of address or other contact information in conformity with the provisions of this Section 16 for the giving of notice. Any notice shall be duly addressed to the parties as follows:

16.1 If to the REIT:

Jeffrey Zimmer ARMOUR Residential REIT, Inc. 3005 Hammock Way Vero Beach, FL 32963 Telecopy: (772) 388-4758 E-mail: jz@armourreit.com

with a copy given in the manner prescribed above, to:

Akerman Senterfitt One Southeast Third Avenue, 25th Floor SunTrust International Center Miami, FL 33131 Telecopy: (305) 374-5095

Attn.: Martin Burkett, Esq., and Bradley Houser, Esq.

E-mail: martin.burkett@akerman.com and bradley.houser@akerman.com

16.2 If to the Manager:

Jeffrey Zimmer ARMOUR Residential Management, LLC 3005 Hammock Way Vero Beach, FL 32963 Telecopy: (772) 388-4758 E-mail: jz@armourreit.com

with a copy given in the manner prescribed above, to:

Cahill Wink LLP

5 Penn Plaza, 23rd Floor New York, NY 10001 Telecopy: (518) 584-1962 Attn: David G. Nichols, Jr. PLLC E-mail: david.nichols@cahillwink.com

- 17. <u>Binding Nature of Agreement: Successors and Assigns</u> This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, personal representatives, successors and assigns as provided in this Agreement.
- 18. Entire Agreement. This Agreement and the Sub-Management Agreement contain the entire agreement and understanding among the parties hereto with respect to the subject matter of this Agreement and the Sub-Management Agreement, and supersede all prior and contemporaneous agreements, understandings, inducements and conditions, express or implied, oral or written, of any nature whatsoever with respect to the subject matter of this Agreement and the Sub-Management Agreement. The express terms of this Agreement and the Sub-Management Agreement control and supersede any course of performance and/or usage of the trade inconsistent with any of the terms of this Agreement or the Sub-Management Agreement. This Agreement may not be modified or amended other than in accordance with Section 27.
- 19. <u>GOVERNING LAW</u>. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF FLORIDA WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES TO THE CONTRARY.
- 20. <u>Jurisdiction; Waiver of Jury Trial</u> Any proceeding or action based upon, arising out of or related to this Agreement or the transactions contemplated hereby shall be brought in any state court of the State of Florida or, in the case of claims to which the federal courts have subject matter jurisdiction, any federal court of the United States of America, in either case, located in the State of Florida, and each of the parties irrevocably submits to the exclusive jurisdiction of each such court in any such proceeding or action, waives any objection it may now or hereafter have to personal jurisdiction, venue or to convenience of forum, agrees that all claims in respect of the proceeding or action shall be heard and determined only in any such court, and agrees not to bring any proceeding or action arising out of or relating to this Agreement or the transactions contemplated hereby in any other court. Nothing herein contained shall be deemed to affect the right of any party to serve process in any manner permitted by law or to commence legal proceedings or otherwise proceed against any other party in any other jurisdiction, in each case, to enforce judgments obtained in any action, suit or proceeding brought pursuant to this Section 20. EACH PARTY HERETO HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT THAT SUCH PARTY MAY HAVE TO A TRIAL BY JURY OF ANY CLAIM OR CAUSE OF ACTION DIRECTLY OR INDIRECTLY BASED UPON OR ARISING OUT OF THIS AGREEMENT.
- 21. No Waiver; Cumulative Remedies. No failure to exercise and no delay in exercising, on the part of any party hereto, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. Except as otherwise provided in this Agreement, the rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law. No waiver of any provision hereunder shall be effective unless it is in writing and is signed by the party asserted to have granted such waiver.
- 22. <u>Headings</u>. The headings of the sections of this Agreement have been inserted for convenience of reference only and shall not be deemed part of this Agreement.

- 23. <u>Counterparts.</u> This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original as against any party whose signature appears thereon, and all of which shall together constitute one and the same instrument. This Agreement shall become binding when one or more counterparts of this Agreement, individually or taken together, shall bear the signatures of all of the parties reflected hereon as the signatories.
- 24. <u>Severability.</u> Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 25. <u>Gender.</u> Words used herein regardless of the number and gender specifically used shall be deemed and construed to include any other number, singular or plural, and any other gender, masculine, feminine or neuter, as the context requires.
- 26. <u>Attorneys' Fees.</u> Should any action or other proceeding be necessary to enforce any of the provisions of this Agreement or the various transactions contemplated hereby, the prevailing party will be entitled to recover its actual reasonable attorneys' fees and expenses from the non-prevailing party.
- 27. Amendments. This Agreement may not be amended, modified or changed (in whole or in part), except by a formal, definitive written agreement expressly referring to this Agreement, which agreement is executed by all of the parties and, in the case of the REIT, if and when any of the stock of the REIT becomes publicly traded, approved by the Board of Directors. The parties hereto expressly acknowledge that no consent or approval of the REIT's stockholders is required in connection with any amendment, modification or change to this Agreement.
- 28. <u>Authority</u>. Each signatory to this Agreement warrants and represents that such signatory is authorized to sign this Agreement on behalf of and to bind the party on whose behalf such signatory is signing this Agreement.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first written above.

"REIT"

ARMOUR RESIDENTIAL REIT, INC.,

a Maryland corporation

By: /s/ Jeffrey J. Zimmer

Name: Jeffrey J. Zimmer

Title: Chairman, CEO & President

"MANAGER"

ARMOUR RESIDENTIAL MANAGEMENT LLC,

a Delaware limited liability company

By: /s/ Jeffrey J. Zimmer

Name: Jeffrey J. Zimmer Title: Managing Partner

[Signature page to Management Agreement]

FORM OF SUB-MANAGEMENT AGREEMENT

This **FORM OF SUB-MANAGEMENT AGREEMENT** (this "<u>Agreement</u>"), is entered into as of November 6, 2009, by and among (i) ARMOUR RESIDENTIAL MANAGEMENT, LLC, a Delaware limited liability company (the "<u>Manager</u>"), (ii) STATON BELL BLANK CHECK LLC, a Delaware limited liability company (the '<u>Sub-Manager</u>"), (iii) ARMOUR RESIDENTIAL REIT, INC., a Maryland corporation, but solely with respect to Sections 1, 6(a), 9(b), 11(a), 11(b), 11(e), 14(a), 14(b), 15, and 18 through 32 (the "<u>REIT</u>"), and (iv) JEFFREY J. ZIMMER and SCOTT J. ULM, but solely with respect to Sections 1, 9, 11(a), 14(a), 15(a), 18 through 30, and 32 (Messrs. Zimmer and Ulm, together, the "<u>Members</u>" and, each, a "<u>Member</u>").

RECITALS

WHEREAS, on the date hereof, the Manager and the REIT are entering into that certain Management Agreement, dated as of the date hereof (as amended from time to time, the "Management Agreement"), pursuant to which the Manager will provide day-to-day operating and investment advisory services to the REIT on the terms and conditions set forth therein; and

WHEREAS, the Manager wishes to enter into this Agreement with the Sub-Manager in order to engage the Sub-Manager to serve as a sub-advisor to the Manager to support the performance of the Manager's services under the Management Agreement on the terms and conditions set forth herein; and

WHEREAS, in connection with the execution of this Agreement, the Members, as shareholders of the Manager, have agreed to provide the Sub-Manager with opportunities to serve as a sub-advisor to (or to enter into alternative arrangements with) the Manager on selected additional advisory arrangements, on the terms and subject to the conditions set forth herein;

NOW, **THEREFORE**, in consideration of the premises and of the mutual covenants and agreement contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

- 1. <u>Definitions.</u> Capitalized terms used but not otherwise defined in this Agreement shall have the respective meanings ascribed to them in the Management Agreement. In addition, the following terms shall have the respective meanings assigned to them below, unless otherwise indicated:
 - "Affiliate" has the meaning set forth in the Management Agreement.
 - "Agreement" means this Sub-Management Agreement, as the same may be amended from time to time.
 - "Base Management Fee" has the meaning set forth in the Management Agreement.
- "Bell Control Persons" means, collectively: (i) Marc H. Bell, (ii) any Affiliate of Marc H. Bell, (iii) any entity, directly or indirectly, owned or controlled by Marc H. Bell and (iv) upon Marc H. Bell's death or disability, the executors, attorneys in fact, administrators of Marc H. Bell or his estate, or any trustee (whether or not as a testamentary trustee or as a successor trustee under a non-testamentary trust), if any, with respect to Marc H. Bell.
 - "Board of Directors" has the meaning set forth in the Management Agreement.
 - "Business Day" has the meaning set forth in the Management Agreement.
- "Cause" means a final determination by a court of competent jurisdiction (a) that the Sub-Manager has materially breached this Agreement that has a material adverse effect on the Manager or the REIT and such material breach has continued for a period of 30 days after receipt by the Sub-Manager of written notice thereof specifying such breach and requesting that the same be remedied in such 30-day period, (b) that an action taken or omitted to be taken by the Sub-Manager in connection with this Agreement constitutes willful misconduct or gross negligence that results in material harm to the Manager and/or the REIT and such willful misconduct or gross negligence has not been cured within a period of 30 days after receipt by the Sub-Manager of written notice thereof specifying such willful misconduct or gross negligence and requesting that the same be remedied in such 30-day period, or (c) that an action taken or omitted to be taken by the Sub-Manager in connection with this Agreement constitutes fraud that results in material harm to the Manager and/or the REIT.
 - "Code Section 409A" has the meaning set forth in Section 32 of this Agreement.
 - "Effective Date" means the date of the consummation of the Merger.
 - "Final Payment" has the meaning set forth in Section 11(b) of this Agreement.
 - "Initial Term" has the meaning set forth in the Management Agreement.

- "Guidelines" means the REIT's the investment guidelines and other parameters for the Investments, financing activities and operations, any modifications to which shall be approved by a majority of the Independent Directors, as the same may be modified with such approval.
 - "Grant" has the meaning set forth in Section 9(a) of this Agreement.
 - "Gross Equity Raised" has the meaning set forth in the Management Agreement.
 - "Independent Directors" has the meaning set forth in the Management Agreement.
- "Independent Revenue Sharer" means any Person who, in connection with a Member Manager's entry into a management agreement with a Related Vehicle, is offered a Grant so long as (i) such Person is not an Affiliate of any Member, the Manager or any of their respective Affiliates, and (ii) none of any of the Members, the Manager or any of their respective Affiliates has any pecuniary interest (through contract, equity or debt interests or otherwise) in such Person or the Grant offered to such Person; provided, that "Independent Revenue Sharer" shall not include any Person who is offered a Grant in connection with the financing or recapitalization of the Manager.
- "Interest Rate" means the current (as of the Termination Date) London Interbank Offered Rate as quoted by Citibank, N.A. (or any successor entity thereto) for interest periods of one year, plus 200 basis points per annum, compounding quarterly.
 - "Investment Company Act" has the meaning set forth in the Management Agreement.
 - "Investments" has the meaning set forth in Section 3(a)(iii) of this Agreement.
 - "Management Agreement" has the meaning set forth in the Recitals to this Agreement.
 - "Manager" has the meaning set forth in the Preamble to this Agreement.
 - "Member" and "Members" have the meaning set forth in the Preamble to this Agreement.
- "Member Entities" means, separately and collectively, the Members and their direct and indirect subsidiaries; provided, however, that the term "Member Entities" shall not include the REIT or any of its subsidiaries.
 - "Member Manager" has the meaning set forth in Section 9(a) of this Agreement.
 - "Merger" has the meaning set forth in the Management Agreement.
 - "Merger Agreement" has the meaning set forth in the Management Agreement.
 - "Notice Date" has the meaning set forth in Section 9(c) of this Agreement.
 - "Parties" has the meaning set forth in Section 32(a) of this Agreement.
 - "Person" has the meaning set forth in the Management Agreement.
 - "Real Estate Investment Trust" has the meaning set forth in the Management Agreement.
 - "REIT" has the meaning set forth in the Preamble to this Agreement.
- "Related Vehicle" means an entity that is involved in the Relevant Business, including, without limitation, a Real Estate Investment Trust, private investment fund, or closed-end fund; provided that "Related Vehicle" shall not include (i) any bank, savings and loan company, other thrift or insurance company or other similar operating financial institution, (ii) any portfolio under management by the Manager as of the date of this Agreement (whether or not such portfolio involves the Relevant Business), or (iii) the REIT.
- "Relevant Business" means the business of managing a portfolio of financial assets primarily comprised of real estate related securities, including, without limitation, hybrid adjustable-rate, adjustable-rate and fixed rate residential mortgage-backed securities issued or guaranteed by FNMA, FHLMC or GNMA, unsecured notes and bonds issued by U.S. Government-chartered entities including FNMA, FHLMC or GNMA, and commercial mortgage-backed securities, distressed mortgage debt and similar assets.
 - "Services" has the meaning set forth in Section 3(a) of this Agreement.
- "Staton Control Persons" means, collectively: (i) Daniel C. Staton, (ii) any Affiliate of Daniel C. Staton, (iii) any entity, directly or indirectly, owned or controlled by Daniel C. Staton and (iv) upon Daniel C. Staton's death or disability, the executors, attorneys in fact, administrators of Daniel C. Staton or his respective estate, or trustee (whether or not as a testamentary trustee or as a successor trustee under a non-testamentary trust), if any, with respect to Daniel C. Staton.

"Staton/Bell Related Persons" means, either (A) collectively (i) Daniel C. Staton, (ii) any Affiliate of Daniel C. Staton, (iii) any trusts (or trustees thereof), family limited partnerships or other estate planning vehicles over which one or more Staton Control Persons exercise control, (iv) upon Daniel C. Staton's death or disability, any executors, attorneys in fact, administrators, testamentary trustees, legatees or beneficiaries of Daniel C. Staton or the respective estate, or any trust formed by either, and (v) to the extent any Staton Control Person retains voting control over the applicable interest, any charitable trust, organization or entity, or (B) collectively (i) Marc H. Bell, (ii) any Affiliate of Marc H. Bell, (iii) any trusts (or trustees thereof), family limited partnerships or other estate planning vehicles over which one or more Bell Control Persons exerci se control, (iv) upon Marc H. Bell's death or disability, any executors, attorneys in fact, administrators, testamentary trustees, legatees or beneficiaries of Marc H. Bell or the respective estate, or any trust formed by either, and (v) to the extent any Bell Control Person retains voting control over the applicable interest, any charitable trust, organization or entity.

"Sub-Manager" has the meaning set forth in the Preamble to this Agreement.

"Sub-Manager Base Management Fee" means a base management fee, calculated and paid (by wire transfer of immediately available funds) monthly in arrears, equal to 25% of the Base Management Fee earned by the Manager or its assignee under the Management Agreement during such month, net of any expenses that shall be the responsibility of the Manager pursuant to Section 7.1 of the Management Agreement or shall not otherwise be reimbursed to the Manager pursuant to the Management Agreement.

"Sub-Manager Termination Fee" means a termination fee equal to 25% of the Termination Fee due payable to the Manager or its assignee under the Management Agreement.

"Termination Date" has the meaning set forth in Section 11(a) of this Agreement.

2. <u>Appointment.</u> The Manager hereby appoints the Sub-Manager to serve as sub-advisor on the terms and conditions set forth in this Agreement, and the Sub-Manager hereby accepts such appointment.

Duties of the Sub-Manager.

- (a) The Sub-Manager shall provide the following services (the "Services") to support the Manager's performance of services to the REIT under the Management Agreement, in each case upon reasonable request by the Manager:
 - (i) serving as a consultant to the Manager with respect to the periodic review of the Guidelines;
 - (ii) identifying for the Manager potential new lines of business and investment opportunities for the

REIT;

- (iii) identifying for and advising the Manager with respect to selection of independent contractors that provide investment banking, securities brokerage, mortgage brokerage and other financial services, due diligence services, underwriting review services, legal and accounting services, and all other services as may be required relating to the investments of the REIT and its subsidiaries (the "Investments");
 - (iv) advising the Manager with respect to the REIT's stockholder and public relations matters;
 - $(v) \qquad \text{advising and assisting the Manager with respect to the REIT's capital structure and capital} \\$

raising; and

government.

(vi) advising the Manager on negotiating agreements relating to programs established by the U.S.

Notwithstanding anything in this Agreement to the contrary, the Manager shall remain primarily and directly responsible for the provision of all services provided to the REIT under the Management Agreement. Without limiting the foregoing, the Manager shall be solely responsible for (i) identifying and consummating all Investments to be made by the REIT and its subsidiaries, (ii) any and all portfolio monitoring or reporting services to be provided to the REIT and (iii) any matters relating to the REIT's Real Estate Investment Trust qualification for U.S. federal income tax purposes or the status of the REIT and its Affiliates under the Investment Company Act.

- (b) The Sub-Manager shall, and shall cause its officers and employees to, devote such portion of its and their time to the provision of the Services to the Manager as necessary for the proper performance of all of the Services hereunder.
- 4. <u>Authority of the Sub-Manager</u>. The Sub-Manager is not authorized to advise or bind the REIT or to enter into any agreements relative to the REIT, and, with respect to the Manager, is to act only as an advisor to the Manager, upon reasonable request. The Sub-Manager shall have no obligation or authority under the Management Agreement.

5. <u>Confidentiality.</u> The Sub-Manager and the Manager shall each keep confidential any nonpublic information obtained in connection with the Services rendered under this Agreement and shall not disclose any such information (or use the same except in furtherance of its duties under this Agreement), except: (i) to the other party hereto and its respective employees, officers, directors, consultants or advisors; (ii) with the prior written consent of the other party hereto; or (iii) as required by law or legal process. The foregoing shall not apply to information which has previously become available through the actions of a Person not resulting a violation this Section 5 or to any information within the public domain or, for the avoidance of doubt, with respect to Sub-Manager's or its Affiliates' business enterprises unrelated to the Services or the REIT. The provisions of this Section 5 shall survive the expiration or earlier termination of this Agreement for a period of one year.

6. Fees

- (a) As compensation for all Services performed by the Sub-Manager under this Agreement, the Sub-Manager shall be paid (i) the Sub-Manager Base Management Fee by the Manager and (ii) in recognition of the level of the upfront effort and commitment of resources required by the Sub-Manager in connection with this Agreement, either the Sub-Manager Termination Fee by the Manager or the Final Payment by the REIT, in each case pursuant to the terms set forth herein. Payment of the Sub-Manager Base Management Fee by the Manager to the Sub-Manager for any month shall be contingent upon the receipt by the Manager of the Base Management Fee under the Management Agreement for such month. The Sub-Manager Base Management Fee shall be payable by the Manager to the Sub-Manager within five (5) Business Days of receipt by the Manager of the Base Management Fee; provided, however, that all the Sub-Manager Base Management Fees with resp ect to the services provided during a given calendar year shall be paid no later than sixty (60) days following the end of such calendar year. Payment of the Sub-Manager Termination Fee by the Manager to the Sub-Manager shall be contingent upon the receipt by the Manager of the Termination Fee under the Management Agreement. Subject to the provisions of Section 11, the Sub-Manager Termination Fee; provided, however, that the Sub-Manager Termination Fee shall be paid no later than sixty (60) days following the end of the calendar year in which the Management Agreement is terminated entitling the Manager to the Termination Fee (which event will, in turn, terminate this Agreement). Payment of the Final Payment shall be made by the REIT to the Sub-Manager in accordance with the provisions of Section 11.
- (b) The Manager agrees that, without the approval of the Sub-Manager (which approval will not be unreasonably withheld, delayed or conditioned), it shall not agree to any modification of the Management Agreement that would both (i) amend or waive (A) the terms of payments due to the Manager under the Management Agreement or (B) the indemnification or expense reimbursement provisions of the Management Agreement and (ii) have either (A) a disproportionately adverse effect on the Sub-Manager or (B) a disproportionately positive result for the Manager.
- (c) The Manager agrees to use reasonable best efforts to collect the Base Management Fee and the Termination Fee on a prompt and timely basis.
- Expenses. The Manager shall promptly submit any expenses incurred by the Sub-Manager on behalf and at the request of the Manager that are eligible for reimbursement by the REIT pursuant to the terms of the Management Agreement to the REIT for reimbursement. The Sub-Manager shall prepare a statement documenting such eligible expenses of the Sub-Manager during each month, and shall deliver such statement to the Manager within five (5) Business Days after the end of each month. The Manager shall reimburse the Sub-Manager for such eligible expenses within five (5) Business Days of receipt by the Manager of reimbursement from the REIT for such eligible expenses (provided, that a failure to deliver such statement within such period shall not limit the rights of the Sub-Manager hereunder except to the extent it prevents the Manager from being reimbursed by the REIT). Reimbursement by the Manager to the Sub-Manager for any expenses shall be contingent upon the receipt by the Manager of reimbursement from the REIT under the Management Agreement for such expenses, and the Manager will pursue such reimbursement with substantially the same level of effort that it pursues requests for its own reimbursement of expenses. Notwithstanding anything herein to the contrary or otherwise: (a) the amount of expenses eligible for reimbursement provided to the Sub-Manager during any calendar year will not affect the amount of expenses eligible for reimbursements provided to the Sub-Manager in any other calendar year, (b) the reimbursements for expenses for which the Sub-Manager is entitled to be reimbursed shall be made on or before the last day of the calendar year following the calendar year in which the applicable expense is incurred, (c) the right to payment or reimbursement hereunder may not be liquidated or exchanged for any other benefit and (d) the reimbursements shall be made pursuant to objectively determinable and nondiscretionary policies and procedures regarding such reimbursement of expenses.

Restrictions on Transfer.

(a) The Sub-Manager shall not directly transfer all or any portion of its right to receive the Sub-Manager Base Management Fee, the Sub-Manager Termination Fee or any of the amounts payable to the Sub-Manager under this Agreement (but excluding proceeds thereof) to any other Person, without the prior written consent of the Manager (which consent shall not be unreasonably withheld, delayed or conditioned).

(b) The consent of the Manager (which shall not be unreasonably be withheld, delayed or conditioned) shall be required prior to a transfer of membership interests in the Sub-Manager that would result in Staton/Bell Related Persons, in the aggregate, holding less than a majority in interest of the Sub-Manager. The term transfer as used in this Section 8(b) shall include any actions that may result in any Staton/Bell Related Person ceasing to be a Staton/Bell Related Person.

Related Vehicles.

- During the period from the Effective Date through the Termination Date, each Member represents and warrants that it has used, and agrees that it shall use, as the case may be, commercially reasonable efforts to ensure that the Manager is designated, and remains designated, as the investment manager with respect to the REIT and any Related Vehicle to which the Member or any of its direct or indirect majority-owned subsidiaries serve as external investment manager. Subject in all cases to Section 9(b) hereof, in the event that the Manager or any direct or indirect majority-owned subsidiary of a Member is designated as the investment manager (a "Member Manager") with respect to any Related Vehicle, the Manager and the Sub-Manager shall enter into good faith negotiations with regard to (a) an agreement pursuant to which the Sub-Manager will be engaged as sub-manager with regard to the management of such Related V ehicle on substantially the terms set forth herein or (b) alternative arrangements that are reasonably acceptable to both the Manager and the Sub-Manager and that provide for substantially the same proportionate compensation to the Sub-Manager as set forth herein and the Member shall cause the Member Manager to provide the Sub-Manager with the right to enter into such an agreement; provided, however, that if the Member Manager grants to an Independent Revenue Sharer irrevocable and fully vested contractual rights (a "Grant") to a percentage of the total fees payable under the Member Manager's management agreement with such Related Vehicle, then the fees to which the Sub-Manager is entitled under its applicable agreement with the Member Manager with respect to such Related Vehicle will be reduced by a percentage equal to the percentage interest provided to such Independent Revenue Sharer in such Grant during the life of such Grant. If the Manager proposes any Grant in connection with this Section 9(a), the Members will provide the Sub-Manager with all information and certifications reasonably requested by the Sub-Manager, including with respect to (x) whether the Person receiving the applicable Grant is an Independent Revenue Sharer and (y) the terms and conditions of the Grant. In connection with any such Grant, to the extent that one or more of the Sub-Manager and Members parties believe in good faith that the adjustment to the Sub-Manager's fee under this Section 9(a) does not fairly allocate the dilution between the parties, the parties will negotiate in good faith to adjust the terms of such adjustment. Notwithstanding any delay in executing such agreement or arrangement, the Sub-Manager shall be entitled to the accrual for payment of fees commencing upon the receipt of management fees by the Manager or such Member Manager with regard to such Related Vehicle. Neither the Members nor the Manager shall have any obligations pursuant to the foregoing sentences of this Section 9(a) if the Sub-Manager remains in material breach of any provision of this Agreement after written notice of such material breach delivered by the Manager to the Sub-Manager; provided, however, upon cure by the Sub-Manager of any such breach, the Members shall cause the applicable Member Manager to negotiate in good faith with the Sub-Manager and provide the Sub-Manager with the right to enter into an agreement described in this paragraph to the extent that the Sub-Manager was, due to such breach, previously not entitled to enter into such an agreement.
- (b) Notwithstanding any other provision of this Agreement, the Sub-Manager shall only be entitled to receive (A) 25% of the net aggregate management or similar fees earned by the Manager and any Member Manager with respect to the REIT and all Related Vehicles cumulatively managed by the Manager and any Member Manager; (B) 25% of the aggregate termination or similar fees earned by the Manager and any Member Manager with respect to the REIT and all Related Vehicles cumulatively managed by the Manager and any Member Manager; and (C) the Final Payment, if any.
- any management entity at the time such management entity is acquired by the Members or their Affiliates after the date of this Agreement, provided, however, that in connection with any such acquisition, the Members shall, or shall cause their applicable Affiliate(s) to, offer to the Sub-Manager (which may assign the right to accept this offer to any if its Affiliates in whole or in part) the right to invest, on the same terms and conditions as the Members or their applicable Affiliate(s) (except that the Sub-Manager and/or any designated Affiliate will be entitled to customary and reasonable minority investor protections), in up to 25% of each class of securities of such management entity acquired by the Members and their respective Affiliates, collectively, provided, further, that the Sub-Manager and/or its desi gnated Affiliates will have 45 days to accept such offer after receiving written notice from the Members describing in reasonable detail all the material terms of the offer (the "Notice Date"), and will have 60 days to fund the purchase price if such offer is accepted after the Notice Date. Subject to conflict of interest policies to be agreed upon between the parties, nothing will limit the ability of the Members and their respective Affiliates, or the Sub-Manager and its Affiliates, to enter into other transactions, including (x) trading in the securities of special purpose acquisition companies other than Enterprise Acquisition Corp. through one or more funds managed by the Members or their respective Affiliates, (y) investing or trading in securities, or pursing

investment strategies, that are the same or similar to those of the REIT, or (z) subject to applicable law, pursuing any other investment or opportunity that may be of interest to the REIT.

10. <u>Relationship of Sub-Manager, Manager, Members and REIT</u>. The Manager and the Members, on the one hand, and the Sub-Manager, on the other hand, are not partners, members or joint venturers with each other nor with the REIT, and nothing in this Agreement shall be construed to make them such partners, members or joint venturers or impose any liability as such on any of them. The relationship of the parties is intended to be contractual and not fiduciary in nature.

11. <u>Term and Termination</u>.

- Subject to Section 12(b), this Agreement shall terminate on the earliest to occur of (i) the election of the Sub-Manager, upon the expiration of the Initial Term of the Management Agreement, to terminate this Agreement, (ii) the termination of the Management Agreement by the REIT, or (iii) the effective date of the removal of the Sub-Manager for Cause (the "Termination Date"); provided that all rights and obligations with respect to any earned but unpaid Sub-Manager Base Management Fee and any other amounts payable under this Agreement with respect to periods prior to, on or in connection with the Termination Date shall survive the termination of this Agreement; provided, further, that, subject to the foregoing proviso, in the event of termination pursuant to clause (i) or (iii) above, there shall be no Sub-Manager Termination Fee paid to the Sub-Manager and, in the event of termination pu rsuant to clause (ii) or (iii) above, there shall be no Final Payment paid to the Sub-Manager. In the event of a termination pursuant to clause (ii) above, if, during the Initial Term, the REIT or any of its Affiliates, on the one hand, and the Manager or any Member Manager, on the other hand, enter into a new management agreement effective within six months of such termination, this Agreement will be deemed to apply with respect to such new management agreement, and, without limiting the foregoing, for purposes of Section 9(a), the Termination Date shall be deemed not to have occurred; provided, however, that the Sub-Manager shall not be entitled to receive any fees during any period in which neither the Manager nor the Managing Member receives fees from the REIT or any of its Affiliates. The applicable Member, or the Members, as may be the case, shall cause the applicable Member Manager, if it is not the Manager, to assume the Manager's obligations under this Agreem ent. In the event one or more of the Sub-Manager and the applicable Member Manager believes in good faith that this Agreement should be amended to reflect differences between the new management agreement and the Management Agreement, the Sub-Manager and the applicable Member Manager shall enter into good faith negotiations with regard to any such appropriate amendments and the applicable Member, or the Members, as may be the case, shall cause the Member Manager to provide the Sub-Manager with the right to enter into any such amendments. In any such event the applicable Member, or the Members, as the case may be, will provide the Sub-Manager with all information and certifications reasonably requested by the Sub-Manager. Notwithstanding any delay in executing any such amendment, the Sub-Manager shall be entitled to the accrual for payment of fees (on the terms as so amended) commencing upon the receipt of management fees by the Manager or such Member Manager with regard to such new ag reement.
- (b) If the Termination Date occurs under Section 11(a)(i), subject to Section 14(b), the REIT shall pay to the Sub-Manager a final payment (the "Final Payment") of 6.16 times the annualized rate of *(i.e.*, 24.64 times) the last three (3) monthly payments of the Sub-Manager Base Management Fee; *provided* that, (i) the Final Payment shall be calculated and determined in accordance with Section 11(e). The Final Payment shall be paid on the date that is 60 days after the Termination Date (or, if such date is not a Business Day, the next Business Day).
- (c) Upon the termination of this Agreement (or, in the case of a termination pursuant to Section 11(a)(iii), the determination of termination in accordance with Section 14(b)), except to the extent inconsistent with applicable law, the Sub-Manager shall as promptly as reasonably practicable (A) deliver to the Manager one copy of all expense statements generated pursuant to Section 7 hereof covering the period following the date of the last provision of such expense statements to the Manager through the Termination Date; and (B) deliver to the Manager all property and documents of the REIT provided to or obtained by the Sub-Manager pursuant to or in connection with this Agreement, including all copies and extracts thereof in whatever form, then in the Sub-Manager's possession or under its control (provided that the Sub-Manager's outside counsel may retain one copy to be kept confidential and used solely for arch ival purposes).
- (d) Subject to other provisions of this Agreement, if the Sub-Manager is removed for Cause, the effective date of a removal for Cause shall be the date upon which the Manager shall have delivered to Sub-Manager both (i) written notice that the Sub-Manager is being removed for Cause in accordance with the Sub-Management Agreement, and (ii) a copy of the applicable final, non-appealable order evidencing the required final determination of the court of competent jurisdiction.
- (e) The Manager shall calculate the Final Payment (and other related amounts or numbers on which the Final Payment is based, directly or indirectly, including, without limitation the Sub-Manager Base Management Fee, Base Management Fee, and Gross Equity Raised) (A) in accordance with (1) the books and records of the REIT and (2) the definitions of Base Management Fee and Gross Equity Raised in the Management Agreement and (B) using the same accounting principles, policies, practices and methodologies used in, and applied consistently with, the prior calculations of

Base Management Fee and Gross Equity Raised under the Management Agreement, and (ii) shall otherwise follow the past practices of the REIT and the Manager with respect to calculating Base Management Fee and Gross Equity Raised, in each case subject to the adjustments expressly provided for in Section 11(b). The REIT agrees to provide such access to its books and records as the Manager may reasonably require to perform the calculations required under this Section 11(e). Upon the Manager's calculation of the Final Payment, the Manager shall promptly deliver to the Sub-Manager a written statement setting forth in reasonable detail its calculation of the Final Payment (including the applicable calculations of Sub-Manager Base Management Fee, Base Management Fee, and Gross Equity Raised) and both the Manager and the REIT shall provide the Sub-Manager reasonable access, during normal business hours and upon reason able notice, to all work papers, schedules, memoranda and other documents prepared or reviewed by the Manager and the REIT, respectively, or by any of their respective representatives that are relevant to the Manager's calculation of the Final Payment (or the applicable calculation of Sub-Manager Base Management Fee, Base Management Fee, and Gross Equity Raised), and such access shall be provided promptly after request by the Sub-Manager. The Manager shall request that its accountants communicate with the Sub-Manager and its representatives; provided, that the Sub-Manager may be required to sign an "indemnification letter" in the form generally used by the Manager's accountant prior to receiving access to any materials prepared by such accountant. Each of the Manager and the REIT shall cause its representatives to be available, during normal business hours and upon reasonable notice, to the Sub-Manager to review the calculation of the Final Payment (including the applicable Sub-Manager Base Management Fee, Base Management Fee, and Gross Equity Raised). In the event that the Sub-Manager disputes the calculation of the Final Payment (including the calculation of the applicable Sub-Manager Base Management, Fee Base Management Fee, and Gross Equity Raised), the Sub-Manager and the Manager will use commercially reasonable efforts, and negotiate in good faith, to promptly reach agreement as to the correct calculation of the Final Payment.

- 12. <u>Assignment</u>. Except as set forth in this Section 12, this Agreement shall not be assigned by any party hereto without the prior written consent of the other parties.
- (a) The Manager may not assign any of its rights and obligations under this Agreement, except that the Manager may assign, in their entirety, its rights and obligations under this Agreement to any third party upon the assignment of its rights and obligations under the Management Agreement to such third party; provided that both the assignee agrees in writing to assume all obligations hereunder and the Manager shall continue to remain liable for its obligations to the Sub-Manager hereunder.
- (b) Upon termination of this Agreement pursuant to Section 11(a)(i) and payment by the REIT of the Final Payment, the right of the Sub-Manager to receive the Sub-Manager Base Management Fee shall be deemed to have been assigned to the REIT, the Manager shall pay the Sub-Manager Base Management Fee to the REIT on a basis consistent with past practice, and the provisions of this Agreement with respect thereto shall survive solely for such purpose; provided, however, that the Manager may, on or prior to the first anniversary of the date on which the Final Payment is made, terminate its obligations and all rights of the REIT hereunder by paying or causing to be paid to the REIT an amount equal to the Final Payment. Any other transfer of fees and other amounts payable to the Sub-Manager under this Agreement shall be governed by Section 8(a).
- 13. <u>Indemnification</u>. The Manager will use commercially reasonable efforts to cause the Sub-Manager to be indemnified by the REIT in accordance with the Management Agreement.

14. Remedies; Limitation of Liability.

Notwithstanding any other provision of this Agreement, in no event shall the Members, the REIT or any of their Affiliates (including the Manager), on the one hand, or the Sub-Manager, on the other hand, be responsible or liable for special, indirect, punitive or consequential loss or damage of any kind whatsoever (including, but not limited to, loss of profit) irrespective of whether the Members, the REIT or their Affiliates, on the one hand, or the Sub-Manager, on the other hand, has been advised of the likelihood of such loss or damage and regardless of the form of action; provided, however, that in connection with any dispute between any Member Entities or the Manager, on the one hand, and the Sub-Manager, on the other hand, regarding the Sub-Manager's right to receive payments under this Agreement, (x) if the Sub-Manager is finally determined to have been entitled to receive any amounts (not paid when due) under t his Agreement, the Sub-Manager will be entitled to (1) reimbursement of reasonable costs and expenses (including attorneys' fees) incurred in connection with such dispute and collection of such amounts and (2) interest accruing at the Interest Rate on such unpaid amounts from the date payment was originally due until actually paid, and (y) if the Sub-Manager is finally determined not to have been entitled to receive any amounts (not paid when due) under this Agreement, the Manager will be entitled to reimbursement of reasonable costs and expenses (including attorneys' fees) incurred in connection with such dispute. Further, notwithstanding any other provision of this Agreement, neither the Manager nor either of the Members (or any of their respective Affiliates) shall be liable to the Sub-Manager for payment of a Sub-Manager Base Management Fee, Sub-Manager Termination Fee or any similar compensation except to the extent that the Manager, the applicable Members or such Affiliate (as the case may be) or its assignee has actually received a corresponding fee from the REIT or any Related Vehicles. This Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or

related to this Agreement or the transactions contemplated hereby may only be brought against, the entities that are expressly named as parties hereto and then only with respect to the specific obligations set forth herein with respect to such party. Except to the extent a named party to this Agreement (and then only to the extent of the specific obligations undertaken by such named party in this Agreement and not otherwise), no past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney, advisor or representative or Affiliate of any of the foregoing shall have any liability (whether in contract, tort, equity or otherwise) for any one or more of the representations, warranties, covenants, agreements or other obligations or liabilities of any one or more of the parties under this Agreement (whether for indemnification or otherwise) of or for any claim based on, arising out of, or related to this Agreement or the transactions contemplated hereby.

- (b) Notwithstanding anything to the contrary in this Agreement, the Manager will not be entitled to terminate this Agreement (except as expressly set forth in the termination provisions above), withhold or offset amounts owed hereunder or otherwise seek recourse against the Sub-Manager for any breach of this Agreement, except that the Manager may seek payment of monetary damages to the extent a court of competent jurisdiction finally determines such damages are awarded to the Manager against the Sub-Manager as a result of Cause (and such damages exceed any amounts released from escrow to the Manager in accordance with this Section 14(b)); provided that, in the event that the Manager has, in good faith and in accordance with this Agreement, initiated litigation with respect to Cause prior to the termination of this Agreement pursuant to Section 11(a)(i), the REIT may, in lieu of making payments to the Sub-Manager with respect to the F inal Payment, but at the time at which such payments would otherwise be required, deposit such payments into an interest-bearing escrow arrangement reasonably satisfactory to the Sub-Manager. If a court of competent jurisdiction finally determines that Cause occurred prior to the termination of this Agreement pursuant to Section 11(a)(i), the amounts then in escrow will be released to the REIT (and the REIT shall have no further obligation to make payments with respect to the Final Payment). If a court of competent jurisdiction finally determines that no Cause occurred prior to the termination of this Agreement pursuant to Section 11(a)(i), then the amounts then in escrow will be released to the Sub-Manager, and the REIT will be responsible for any additional amounts owing in respect of the Final Payment.
- 15. <u>Representations and Warranties of the REIT, Each Member and the Manager.</u> The REIT, the Manager and each of the Members each, severally and not jointly, hereby represents and warrants to the Sub-Manager, as follows:
- (a) It (a) in the case of the REIT, is a corporation and is duly organized, validly existing and in good standing under the laws of the State of Maryland, and is qualified to do business in every jurisdiction in which the failure to so qualify might reasonably be expected to have a material adverse effect on the financial condition, operating results, assets, operations or business prospects of it and its subsidiaries taken as a whole, (b) in the case of the Manager, is a limited liability company and is duly organized, validly existing and in good standing under the laws of the State of Delaware, and is qualified to do business in every jurisdiction in which the failure to so qualify might reasonably be expected to have a material adverse effect on the financial condition, operating results, assets, operations or business prospects of it and its subsidiaries taken as a whole, and (c) in the case of each Member, is a natural person. It has all requisite corporate or other power and authority and all material licenses, permits and authorizations necessary to own and operate its properties, to carry on its businesses as now conducted and to carry out the transactions contemplated by this Agreement.
- (b) The execution, delivery and performance of this Agreement has been duly authorized by it. This Agreement has been duly executed and delivered by it and, assuming due execution and delivery by the Sub-Manager, constitutes a valid and binding obligation of it, enforceable in accordance with its terms. The execution and delivery by it of this Agreement, and the fulfillment of and compliance with the terms hereof by it, do not and will not (i) conflict with or result in a breach of the terms, conditions or provisions of, (ii) constitute a default under, (iii) result in the creation of any lien, security interest, charge or encumbrance upon its capital stock or membership interests, as applicable, or assets pursuant to, (iv) give any third party the right to modify, terminate or accelerate any obligation under, (v) result in a violation of, or (vi) require any authorization, consent, approval, exemption or other acti on by or notice to any court or administrative or governmental body pursuant to, its organizational documents (if applicable), or any law, statute, rule or regulation to which it is subject, or any agreement, instrument, order, judgment or decree to which it is a party or by which it is bound.
- 16. <u>Representations and Warranties of the Sub-Manager</u>. The Sub-Manager hereby represents and warrants to the Members, the REIT, and the Manager as follows:
- (a) The Sub-Manager is a limited liability company duly organized, validly existing and in good standing under the laws of Delaware and is qualified to do business in every jurisdiction in which the failure to so qualify might reasonably be expected to have a material adverse effect on the financial condition, operating results, assets, operations or business prospects of the Sub-Manager taken as a whole. The Sub-Manager has all requisite power and authority and all material licenses, permits and authorizations necessary to own and operate its properties, to carry on its businesses as now conducted and to carry out the transactions contemplated by this Agreement.

- (b) The execution, delivery and performance of this Agreement has been duly authorized by the Sub-Manager. This Agreement has been duly executed and delivered by the Sub-Manager and, assuming due execution and delivery by the Members and the Manager, constitutes a valid and binding obligation of the Sub-Manager, enforceable in accordance with its terms. The execution and delivery by the Sub-Manager of this Agreement, and the fulfillment of and compliance with the terms hereof by the Sub-Manager, do not and will not (i) conflict with or result in a breach of the terms, conditions or provisions of, (ii) constitute a default under, (iii) result in the creation of any lien, security interest, charge or encumbrance upon the Sub-Manager's capital stock or assets pursuant to, (iv) give any third party the right to modify, terminate or accelerate any obligation under, (v) result in a violation of, or (vi) require any a uthorization, consent, approval, exemption or other action by or notice to any court or administrative or governmental body pursuant to, the Sub-Manager's certificate of formation or amended and restated limited liability company agreement, or any law, statute, rule or regulation to which the Sub-Manager is subject, or any agreement, instrument, order, judgment or decree to which the Sub-Manager is a party or by which it is bound.
- 17. <u>Further Assurances.</u> The Sub-Manager and the Manager will use commercially reasonable efforts to cooperate to give effect to the arrangements contemplated hereby.
- 18. Notices. Unless expressly provided otherwise in this Agreement, all notices, requests, demands and other communications required or permitted under this Agreement shall be in writing and shall be deemed to have been duly given, made and received when (1) delivered by hand, (2) otherwise delivered by reputable overnight courier against receipt therefor, or (3) upon actual receipt of registered or certified mail, postage prepaid, return receipt requested. The parties may deliver to each other notice by electronically transmitted facsimile copies or electronically transmitted mail (i.e., e-mail), provided that such facsimile or e-mail notice is followed within 24 hours by any type of notice otherwise provided for in this Section 18. Any party may alter the address or other contact information to which communications or copies are to be sent by giving notice of such change of address or other contact information in conformity with the provisions of this Section 18 for the giving of notice. Any notice shall be duly addressed to the parties as follows:

If to the Manager:

Jeffrey Zimmer ARMOUR Residential Management, LLC 3005 Hammock Way Vero Beach, FL 32963 Telecopy: (772) 388-4758 E-mail: [●]

with a copy given in the manner prescribed above, to:

Cahill Wink LLP 5 Penn Plaza, 23rd Floor New York, NY 10001 Telecopy: (518) 584-1962 Attn: David G. Nichols, Jr. PLLC E-mail: david.nichols@cahillwink.com

If to the Sub-Manager:

Staton Bell Blank Check LLC 6800 Broken Sound Parkway Boca Raton, Florida 33487 Attention: Daniel C. Staton E-mail: daniel@statoncapital.com

with a copy given in the manner prescribed above, to:

Akerman Senterfitt One Southeast Third Avenue, 25th Floor SunTrust International Center Miami, FL 33131

Telecopy: (305) 374-5095

Attn.: Martin Burkett, Esq., and Bradley Houser, Esq.

E-mail: martin.burkett@akerman.com and bradley.houser@akerman.com

[•]

with a copy given in the manner prescribed above, to:

Cahill Wink LLP 5 Penn Plaza, 23rd Floor New York, NY 10001 Telecopy: (518) 584-1962 Attn: David G. Nichols, Jr. PLLC E-mail: david.nichols@cahillwink.com

If to Scott J. Ulm:

[•]

with a copy given in the manner prescribed above, to:

Cahill Wink LLP 5 Penn Plaza, 23rd Floor New York, NY 10001 Telecopy: (518) 584-1962 Attn: David G. Nichols, Jr. PLLC E-mail: david.nichols@cahillwink.com

If to the REIT:

Jeffrey Zimmer ARMOUR Residential REIT, Inc. 3005 Hammock Way Vero Beach, FL 32963 Telecopy: (772) 388-4758 E-mail: jz@armourreit.com

with a copy given in the manner prescribed above, to:

Akerman Senterfitt One Southeast Third Avenue, 25th Floor SunTrust International Center Miami, FL 33131 Telecopy: (305) 374-5095

Attn.: Martin Burkett, Esq., and Bradley Houser, Esq.

E-mail: martin.burkett@akerman.com and bradley.houser@akerman.com

- 19. <u>Binding Nature of Agreement; Successors and Assigns</u> This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, personal representatives, successors and permitted assigns as provided in this Agreement.
- 20. Entire Agreement. This Agreement contains the entire agreement and understanding among the parties hereto with respect to the subject matter of this Agreement, and supersedes all prior and contemporaneous agreements, understandings, inducements and conditions, express or implied, oral or written, of any nature whatsoever with respect to the subject matter of this Agreement. The express terms of this Agreement control and supersede any course of performance and/or usage of the trade inconsistent with any of the terms of this Agreement. This Agreement may not be modified or amended other than in accordance with Section 29.
- 21. <u>GOVERNING LAW</u>. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF FLORIDA WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES TO THE CONTRARY.
- 22. <u>Jurisdiction; Waiver of Jury Trial</u> Any proceeding or action based upon, arising out of or related to this Agreement or the transactions contemplated hereby shall be brought in any state court of the State of Florida or, in the case of claims to which the federal courts have subject matter jurisdiction, any federal court of the United States of America, in either case, located in the State of Florida, and each of the parties irrevocably submits to the exclusive jurisdiction of each such court in any such proceeding or action, waives any objection it may now or hereafter have to personal jurisdiction, venue or

to convenience of forum, agrees that all claims in respect of the proceeding or action shall be heard and determined only in any such court, and agrees not to bring any proceeding or action arising out of or relating to this Agreement or the transactions contemplated hereby in any other court. Nothing herein contained shall be deemed to affect the right of any party to serve process in any manner permitted by law or to commence legal proceedings or otherwise proceed against any other party in any other jurisdiction, in each case, to enforce judgments obtained in any action, suit or proceeding brought pursuant to this Section 22. EACH PARTY HERETO HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT THAT SUCH PARTY MAY HAVE TO A TRIAL BY JURY OF ANY CLAIM OR CAUSE OF ACTION DIRECTLY OR INDIRECTLY BASED UPON OR ARISING OUT OF THIS AGREEMENT.

- 23. <u>No Waiver: Cumulative Remedies.</u> No failure to exercise and no delay in exercising, on the part of any party hereto, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. Subject to Section 14, the rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law. No waiver of any provision hereunder shall be effective unless it is in writing and is signed by the party asserted to have granted such waiver.
- 24. <u>Headings</u>. The headings of the sections of this Agreement have been inserted for convenience of reference only and shall not be deemed part of this Agreement.
- 25. <u>Counterparts.</u> This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original as against any party whose signature appears thereon, and all of which shall together constitute one and the same instrument. This Agreement shall become binding when one or more counterparts of this Agreement, individually or taken together, shall bear the signatures of all of the parties reflected hereon as the signatories.
- 26. <u>Severability</u>. Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 27. <u>Gender.</u> Words used herein regardless of the number and gender specifically used, shall be deemed and construed to include any other number, singular or plural, and any other gender, masculine, feminine or neuter, as the context requires.
- 28. <u>Attorneys' Fees.</u> Should any action or other proceeding be necessary to enforce any of the provisions of this Agreement or the various transactions contemplated hereby, the prevailing party or parties will be entitled to recover it or their actual reasonable attorneys' fees and expenses from the non-prevailing party or parties whose actions, or failure or omission to act, gave rise to such action or other proceeding.
- 29. Amendments. This Agreement may not be amended, modified or changed (in whole or in part), except by a formal, definitive written agreement expressly referring to this Agreement, which agreement shall be executed by all the parties affected by such agreement. If the REIT is such a party affected by any such agreement amending modifying or changing this Agreement, such agreement must be, if and when any of the stock of the REIT becomes publicly traded, approved by the Board of Directors. The parties hereto expressly acknowledge that no consent or approval of the REIT's stockholders is required in connection with any amendment, modification or change to this Agreement.
- 30. Non-Disparagement. (i) The Sub-Manager hereby agrees to refrain, and to use reasonable efforts to cause its members, from making any defamatory or derogatory statements concerning the REIT, the Manager and the Members, and (ii) the REIT, the Manager and the Members hereby agree to refrain from making any defamatory or derogatory statements concerning the Sub-Manager and its members. This Section 29 shall survive termination of this Agreement for a period of five years. Notwithstanding the foregoing, nothing herein shall be deemed to prohibit any individual or entity from making any truthful statements in response to a subpoena, in connection with a legal proceeding or as otherwise required by law or legal process.
- 31. <u>Authority</u>. Each signatory to this Agreement with respect to the Manager, Sub-Manager, or REIT warrants and represents that such signatory is authorized to sign this Agreement on behalf of and to bind the party on whose behalf such signatory is signing this Agreement.

32. <u>Section 409A</u>.

(a) <u>General</u>. It is the intention of all the parties to this Agreement (the <u>Parties</u>") that the benefits and rights to which the Sub-Manager is entitled pursuant to this Agreement comply with Code Section 409A, to the extent that the requirements of Code Section 409A are applicable thereto, and the provisions of this Agreement shall be construed in a manner consistent with that intention. If the Parties believes, at any time, that any such benefit or right that is subject to

Code Section 409A does not so comply, it shall promptly advise the other and shall negotiate reasonably and in good faith to amend the terms of such benefits and rights such that they comply with Code Section 409A (with the most limited possible economic effect on the Parties).

- (b) <u>Distributions On Account Of Separation from Service</u>. To the extent required to comply with Code Section 409A, any payment or benefit required to be paid under this Agreement on account of termination of the Agreement (or any other similar term) shall be made only in connection with a "separation from service" with respect to the Sub-Manager within the meaning of Code Section 409A.
- (c) No Acceleration of Payments. The Parties, neither individually or in combination, may accelerate any payment or benefit that is subject to Code Section 409A, except in compliance with Code Section 409A and the provisions of this Agreement, and no amount that is subject to Code Section 409A shall be paid prior to the earliest date on which it may be paid without violating Code Section 409A.
- (d) Six Month Delay for Specified Employees. In the event that any payment under the Agreement is to be made to a "specified employee" (as described in Code Section 409A), then the Parties shall cooperate in good faith to undertake any actions that would cause such payment or benefit not to constitute deferred compensation under Code Section 409A. In the event that, following such efforts, the REIT or the Manager determine (after consultation with its counsel) that such payment or benefit is still subject to the six-month delay requirement described in Code Section 409A(2)(b) in order for such payment or benefit to comply with the requirements of Code Section 409A, then no such payment or benefit shall be made before the date that is six months after the specified employee's "separation from service" (as described in Code Section 409A) (or, if earlier, the date of the specified employee's death). Any payment or benefit delayed by reason of the prior sentence shall be paid out or provided in a single lump sum at the end of such required delay period in order to catch up to the original payment schedule.
- (e) <u>Treatment of Each Installment as a Separate Payment</u> For purposes of applying the provisions of Code Section 409A to this Agreement, each separately identified amount to which the Sub-Manager is entitled under this Agreement shall be treated as a separate payment. In addition, to the extent permissible under Code Section 409A, any series of installment payments under this Agreement shall be treated as a right to a series of separate payments.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first written above.

ARMOUR RESIDENTIAL MANAGEMENT, LLC

By: /s/ Jeffrey J. Zimmer
Name: <u>Jeffrey J. Zimmer</u>
Title: Managing Partner
STATON BELL BLANK CHECK LLC
By: /s/ Daniel C. Staton
Name: Daniel C. Staton
Title: Managing Member
Title. Managing Memoer
ARMOUR RESIDENTIAL REIT, INC., solely with respect to Sections 1,
6(a), 9(b), 11(a), 11(b), 11(e), 14(a), 14(b), 15, and 18 through 32
By: /s/ Jeffrey J. Zimmer
Name: Jeffrey J. Zimmer
Title: Chairman, Co-CEO & President
JEFFREY J. ZIMMER, solely with respect to Sections 1, 9, 11(a), 14(a),
15(a), 18 through 30, and 32
By: /s/ Jeffrey J. Zimmer
Name: Jeffrey J. Zimmer
CCOTT I III II I
SCOTT J. ULM, solely with respect to Sections 1, 9, 11(a), 14(a), 15(a), 18
through 30, and 32
unrough 50, and 52
By: /s/ Scott J. Ulm
o /

[Signature page to Sub-Management Agreement]

AMENDED AND RESTATED MANAGEMENT AGREEMENT

This **MANAGEMENT AGREEMENT** is entered into as of November 6, 2009, by and between (i) ARMOUR RESIDENTIAL REIT, INC., a Maryland corporation (the "<u>REIT</u>"), and (ii) ARMOUR RESIDENTIAL MANAGEMENT LLC, a Delaware limited liability company (the "<u>Manager</u>").

RECITALS

WHEREAS, the REIT and Manager are parties to that certain Management Agreement dated as of November 6, 2009 (the "Management Agreement"); and

WHEREAS, the parties desire to amend and restate the Management Agreement upon the terms and conditions provided herein.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- 1. <u>Definitions.</u> Capitalized terms used but not otherwise defined in this Agreement shall have the respective meanings assigned to them below:
- 1.1 "Affiliate" means, with respect to any specified Person, any other Person that directly or indirectly controls, is controlled by, or is under common control with, that specified Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlling", "controlled by" and "under common control with"), with respect to any specified Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of that specified Person, whether by contract, through the ownership of voting securities or other equity interests (including partnership or membership interests), or otherwise.
 - 1.2 "Agreement" means this Management Agreement, as the same may be amended from time to time.
 - 1.3 "Annual Minimum Fee" means \$900,000 for each fiscal year of this Agreement.
 - 1.4 "Base Management Fee" shall have the meaning set forth in Section 6.1 of this Agreement.
- 1.5 "Board of Directors" means the member(s) of the Board of Directors of the REIT, applicable if and when any of the stock of the REIT becomes publicly traded.
- 1.6 "Business Day" means a day on which the banks are opened for business (Saturdays, Sundays, statutory and civic holidays excluded) in New York, New York, United States.
- 1.7 "Cause" means, for purposes of a termination of this Agreement by the REIT without penalty or payment of a Termination Fee, a final determination by a court of competent jurisdiction (a) that the Manager has materially breached this Agreement that has a material adverse effect on the REIT and such material breach has continued for a period of 30 days after receipt by the Manager of written notice thereof specifying such breach and requesting that the same be remedied in such 30-day period, (b) that an action taken or omitted to be taken by the Manager in connection with this Agreement constitutes willful misconduct or gross negligence that results in material harm to the REIT and such willful misconduct or gross negligence has not been cured within a period of 30 days after receipt by the Manager of written notice thereof specifying such willful misconduct or gross negligence and requesting that the same be remedied in such 30-day period, or (c) that an action taken or omitted to be taken by the Manager in connection with this Agreement constitutes fraud that results in material harm to the REIT.
 - 1.8 "Code" means the Internal Revenue Code of 1986, as amended.
 - 1.9 "<u>Effective Date</u>" means the date of the consummation of the Merger.
- 1.10 "Governing Instruments" means the articles of incorporation or charter, as the case may be, and the bylaws of the REIT and its subsidiaries, as those documents may be amended from time to time.

- 1.11 "Gross Equity Raised" means an amount in dollars calculated as of the date of determination that is equal to (a) the initial equity capital of the REIT following the consummation of the Merger, plus (b) equity capital raised in public or private issuances of the REIT's equity securities (calculated before underwriting fees and distribution expenses, if any), less (c) capital returned to the stockholders of the REIT, as adjusted to exclude (d) one-time charges pursuant to changes in GAAP and certain non-cash charges after discussion between the Manager and the Board of Directors and approved by a majority of the Board of Directors, if and when any of the stock of the REIT becomes publicly traded.
- 1.12 "Independent Directors" means the members of the Board of Directors who are not officers or employees of the Manager or any Person directly or indirectly controlling or controlled by the Manager, and who are otherwise "independent" in accordance with the REIT's Governing Instruments and policies and, if applicable, the rules of any national securities exchange on which the REIT's common stock is listed.
 - 1.13 "Initial Term" shall have the meaning set forth in Section 10.1 of this Agreement.
 - 1.14 "Investment Company Act" shall mean the Investment Company Act of 1940, as amended.
- 1.15 "Manager" shall have the meaning set forth in the Preamble of this Agreement and shall include any successor thereto (subject to the provisions of Section 13).
- 1.16 "Manager Obligations" shall have the meaning set forth in Section 2.4.2 of this Agreement and may be limited from time to time in the REIT's discretion.
 - 1.17 "Manager Shareholders" shall have the meaning set forth in Section 2.5 of this Agreement.
 - 1.18 "Merger" means the merger contemplated pursuant to the Merger Agreement.
- 1.19 "Merger Agreement" means that Agreement and Plan of Merger, dated as of July 29, 2009, among the REIT, ARMOUR Merger Sub Corp., a Delaware corporation, and Enterprise Acquisition Corp., a Delaware corporation.
- 1.20 "Mortgage Assets" means the following assets types of the REIT which the REIT may determine from time to time shall be solely managed by the Manager:
 - (i) mortgage securities (or interests therein), including (a) adjustable-rate, hybrid adjustable-rate and pass-through certificates (including GNMA certificates, FNMA certificates and FHLMC certificates), collateralized mortgage obligations, (c) securities representing interests in, or secured by, agency wrapped mortgages on real property other than pass-through certificates and CMOs, (d) agency mortgage derivative securities and other agency mortgage-backed and mortgage collateralized obligations, and (e) mortgage derivative securities;
 - (ii) U.S. government issued bills, notes and bonds including general obligations of the agencies of the U.S. government (including, but not limited to GNMA, FNMA and FHLMC); and
 - (iii) short-term investments, including short-term bank certificates of deposit, short-term U.S. Treasury securities, short-term U.S. government agency securities, commercial paper, repurchase agreements, short-term CMOs, short-term asset backed securities and other similar types of short-term investment instruments, all of which will have maturities or average lives of less than one (1) year.
 - 1.21 "Non-Renewal Notice" shall have the meaning set forth in Section 10.1 of this Agreement.
 - 1.22 "Notice of Proposal to Negotiate" shall have the meaning set Forth in Section 10.5 of this Agreement.
- 1.23 "Person" means any individual, corporation, partnership, joint venture, limited liability company, estate, trust, unincorporated association, any federal, state, county or municipal government or any bureau, department or agency thereof and any fiduciary acting in such capacity on behalf of any of the foregoing.
 - 1.24 "Real Estate Investment Trust" means a "real estate investment trust" as defined under the Code.
- 1.25 "REIT" shall have the meaning set forth in the Preamble of this Agreement and shall include any subsidiary and any successor thereto.

- 1.26 "REIT Provisions of the Code" means Sections 856 through 860 of the Code.
- 1.27 "Renewal Term" shall have the meaning set forth in Section 10.1 of this Agreement.
- 1.28 "Staton Bell" shall have the meaning set forth in Section 2.5 of this Agreement.
- 1.29 "Sub-Management Agreement" shall have the meaning set forth in Section 2.5 of this Agreement.
- 1.30 "Termination Fee" means an amount equal to three (3) times the Base Management Fee paid to the Manager in the preceding full twelve (12) months, calculated as of the effective date of the termination of this Agreement pursuant to Section 10.2.

2. <u>General Duties of the Manager</u>.

- 2.1 Services. Until any of the stock of the REIT becomes publicly traded, all services performed by the Manager under this Agreement shall be under the direction of the Manager. If and when any of the stock of the REIT becomes publicly traded, all services performed by the Manager under this Agreement shall be subject to the direction and oversight of the Board of Directors. As may be limited from time to time by the REIT in its discretion, the Manager shall (i) manage the day-to-day operations of the REIT and perform the services and other activities described below, and (ii) to the extent directed by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), perform similar management and services for any subsidiary of the REIT; provided, however, that nothing herein shall give the Manager the right (or obligate the Manager) to supervise any other manager engaged by the REIT (each such other manager, an "Other Manager"), or to manage or otherwise participate in any way in any securitization transaction undertaken by the REIT or any joint venture formed by the REIT. Subject to the REIT's right to retain Other Managers and the REIT's right to limit the following duties in its discretion from time to time to the Mortgage Assets which the REIT determines from time to time shall be solely managed by the Manager, the Manager shall perform the following services from time to time as may be required for the management of the REIT and its assets (other than any such assets solely being managed by an Other Manager):
 - 2.1.1 serving as a consultant to the REIT with respect to the formulation of investment criteria for assets managed by the Manager and the preparation of policy guidelines by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) for such assets;
 - 2.1.2 assisting the REIT in developing criteria for Mortgage Asset purchase commitments that are consistent with the REIT's long-term investment objectives and making available to the REIT its knowledge and experience with respect to Mortgage Assets managed by the Manager;
 - 2.1.3 representing the REIT in connection with certain of the REIT's purchases, sales and commitments to purchase or sell Mortgage Assets managed by the Manager that meet in all material respects the REIT's investment criteria, including without limitation by providing repurchase agreement and similar portfolio management expertise as appropriate in connection therewith:
 - 2.1.4 managing the REIT's Mortgage Assets (other than any Mortgage Assets managed solely by Other Managers);
 - 2.1.5 advising the REIT and negotiating the REIT's agreements with third-party lenders for borrowings by the REIT;
 - 2.1.6 making available to the REIT statistical and economic research and analysis regarding the REIT's activities managed by the Manager and the services performed for the REIT by the Manager;
 - 2.1.7 monitoring and providing to the Board of Director, if and when any of the stock of the REIT becomes publicly traded, from time to time price information and other data obtained from certain nationally-recognized dealers that maintain markets in mortgage assets identified by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) from time to time, and providing data and advice to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) in connection with the identification of such dealers, in each case with respect to assets managed by the Manager;

- 2.1.8 investing or reinvesting money of the REIT, which the REIT determines from time to time shall be solely managed by the Manager, in accordance with the REIT's policies and procedures;
- 2.1.9 providing executive and administrative personnel, office space and other appropriate services required in rendering services to the REIT, in accordance with and subject to the terms of this Agreement;
- 2.1.10 administering the day-to-day operations of the REIT and performing and supervising the performance of such other administrative functions necessary to the management of the REIT as may be agreed upon by the Manager and the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), including, without limitation, the collection of revenues and the payment of the REIT's debts and obligations from the REIT's accounts (in each case in respect of assets managed by the Manager), and the maintenance of appropriate computer systems and related information technology to perform such administrative and management functions;
- 2.1.11 advising the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) in connection with certain policy decisions (other than any such decisions solely relating to Other Managers);
- 2.1.12 evaluating and recommending hedging strategies to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) and, upon approval by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), engaging in hedging activities on behalf of the REIT consistent with the REIT's status as a Real Estate Investment Trust, in each case in respect of assets managed by the Manager;
- 2.1.13 supervising compliance by the REIT with the REIT Provisions of the Code and maintenance of its status as a Real Estate Investment Trust (other than in respect of any assets not managed by the Manager);
- 2.1.14 qualifying and causing the REIT to qualify to do business in all applicable jurisdictions and obtaining and maintaining all appropriate licenses (other than in respect of any activities not managed by the Manager);
- 2.1.15 assisting the REIT to retain qualified accountants and tax experts to assist in developing and monitoring appropriate accounting procedures and testing systems and to conduct quarterly compliance reviews as the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) may deem necessary or advisable (other than any such procedures or reviews relating solely to Other Managers);
- 2.1.16 assisting the REIT in its compliance with all federal (including, without limitation, the Sarbanes-Oxley Act of 2002), state and local regulatory requirements applicable to the REIT in respect of its business activities, including preparing or causing to be prepared all financial statements required under applicable regulations and contractual undertakings and all reports, documents and filings, if any, required under the Securities Exchange Act of 1934, as amended, or other federal or state laws;
- 2.1.17 assisting the REIT in its compliance with federal, state and local tax filings and reports, and generally enable the REIT to maintain its status as a Real Estate Investment Trust, including soliciting stockholders, as defined below, for required information to the extent provided in the REIT Provisions of the Code;
- 2.1.18 assisting the REIT in its maintenance of an exemption from the Investment Company Act and monitoring compliance with the requirements for maintaining an exemption from the Investment Company Act;
- 2.1.19 advising the REIT as to its capital structure and capital raising activities (other than in respect of capital not to be managed by the Manager);
- 2.1.20 handling and resolving all claims, disputes or controversies (including all litigation, arbitration, settlement or other proceedings or negotiations) in which the REIT may be involved or to which the REIT may be subject arising out of the REIT's day-to-day operations, subject to the approval of the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) and excluding any such proceedings or negotiations solely involving Other Managers;
- 2.1.21 engaging and supervising, on behalf of the REIT at the REIT's request and at the REIT's expense, the following, without limitation: independent contractors to provide investment banking services, leasing services, mortgage brokerage services, securities brokerage services, other financial services and such other services as may

be deemed by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) to be necessary or advisable from time to time (other than Other Managers, or any of the foregoing to be utilized in connection with activities being solely conducted by Other Managers);

- 2.1.22 so long as the Manager does not incur additional costs or expenses, and the REIT does not incur additional costs or expenses which are not specifically approved in writing by the REIT, performing such other services as may be necessary or advisable from time to time for management and other activities relating to the assets of the REIT as the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) shall reasonably request or the Manager shall deem appropriate under the particular circumstances; and
- 2.1.23 assisting the REIT, upon the REIT's request therefor, in evaluating the advantages and disadvantages of the REIT internalizing the functions of the Manager or of any merger and acquisition transaction that the REIT may elect to pursue, which also may be subject to approval by the shareholders of the REIT.

2.2 <u>Obligations of the Manager</u>.

- 2.2.1 <u>Verify Conformity with Acquisition Criteria</u>. At all times (and, if and when any of the stock of the REIT becomes publicly traded, subject to the direction of the Board of Directors), the Manager shall use commercially reasonable efforts to provide that each Mortgage Asset acquired by the Manager for the REIT conforms in all material respects to the acquisition criteria of the REIT and shall seek to cause each seller or transferor of such Mortgage Assets to the REIT to make such representations and warranties regarding such Mortgage Assets as may, in the reasonable judgment of the Manager, be necessary and appropriate, subject to market custom. In addition, the Manager shall take such other action as it deems reasonably necessary or appropriate in seeking to protect the REIT's investments to the extent consistent with its duties under this Agreement.
- 2.2.2 Conduct Activities in Conformity with REIT Status and All Applicable Restrictions. At all times (and, if and when any of the stock of the REIT becomes publicly traded, subject to the direction of the Board of Directors) and with reasonable advance notice from the REIT of any pertinent information relating to any activities of the REIT as may then be conducted by Other Managers, the Manager shall refrain from any action which would adversely affect the status of the REIT or, if applicable, any subsidiary of the REIT as a Real Estate Investment Trust or (i) which would violate any material law, rule or regulation of any governmental body or agency having jurisdiction over the REIT or any such subsidiary or (ii) which would otherwise not be permitted by the REIT's or such subsidiary's Governing Instruments, any material operating policies adopted by the REIT, or any agreements actually known by the Manager, except in each of clauses (i) and (ii) as could not reasonably be expected to have a material adverse effect on the REIT. If the Manager is directed to take any such action by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), the Manager's judgment that such action would adversely affect such status or cause such violation or not be permitted as aforesaid.
- 2.2.3 Reports. If and when any of the stock of the REIT becomes publicly traded and upon the request of the Board of Directors and at the sole cost and expense of the REIT, the Manager shall cause an annual compliance report of the REIT to be prepared by a firm independent of the Manager and its Affiliates and having the proper expertise to determine compliance with the REIT Provisions of the Code and related matters. In addition, the Manager shall prepare regular reports for the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) that will review the REIT's acquisitions of Mortgage Assets, portfolio composition and characteristics, credit quality (if applicable), performance and compliance with the REIT's investment policies and policies that enable the REIT to maintain its qualification as a Real Estate Investment Trust and to maintain its exemption from being deemed an "investment company" under the Investment Company Act; provided that such reports shall only relate to assets the REIT has determined shall be managed by the Manager.
- 2.2.4 <u>Portfolio Transactions.</u> In placing portfolio transactions and selecting brokers or dealers, the Manager shall seek to obtain on behalf of the REIT commercially reasonable terms. In assessing commercially reasonable terms for any transaction, the Manager shall consider all factors it deems relevant, including, without limitation, the breadth of the market for the security, the price of the security, the financial condition and execution capability of the broker or dealer, and the reasonableness of the commission, if any, both for the specific transaction and on a continuing basis.

2.3 Cooperation of the REIT. The REIT (and, if and when any of the stock of the REIT becomes publicly traded, the Board of Directors) shall take such actions as may reasonably be required to permit and enable the Manager to carry out its duties and obligations under this Agreement, including, without limitation, the steps reasonably necessary to allow the Manager to file any registration statement on behalf of the REIT in a timely manner if the REIT requests that the Manager do so. The REIT further agrees to use commercially reasonable efforts to make available to the Manager reasonably available resources, information and materials reasonably requested by the Manager to enable the Manager to satisfy its obligations hereunder, including its obligations to deliver financial statements and any other information or reports with respect to the REIT. If the Manager is not able to provide a service, or in the reasonable judgment of the Manager it is not prudent to provide a service, without the approval of the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), then the Manager shall be excused from providing such service (and shall not be in breach of this Agreement) until the applicable approval has been obtained; *provided*, *however*, that the Manager shall have promptly advised the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) in writing that the Manager is awaiting such approval.

2.4 Engagement of Third Parties.

- 2.4.1 <u>Securities Dealers.</u> Subject to the REIT's right to retain Other Managers and the REIT's right to limit the Manager's authorizations in the REIT's discretion from time to time, the Manager is authorized, for and on behalf, and at the sole cost and expense of the REIT, to employ such securities dealers (including Affiliates of the Manager) for the purchase and sale of the REIT's Mortgage Assets managed by the Manager as may, in the reasonable judgment of the Manager, be necessary to obtain the best commercially available net results taking into account such factors as the policies of the REIT, price, dealer spread, the size, type and difficulty of the transaction involved, the firm's general execution and operational facilities and the firm's risk in positioning the securities involved. Consistent with this policy, and subject to the foregoing caveats with respect to the REIT's rights, the Manager is authorized to direct the execution of the REIT's portfolio transactions to dealers and brokers furnishing statistical information or research deemed by the Manager to be reasonably necessary to the performance of its investment advisory functions for the REIT.
- 2.4.2 Other Third Parties. The Manager is authorized to retain, for and on behalf of the REIT, the services of third parties (including Affiliates of the Manager), including, without limitation, accountants, legal counsel, appraisers, insurers, brokers, dealers, transfer agents, registrars, developers, investment banks, financial advisors, banks and other lenders and others as the Manager deems reasonably necessary or advisable in connection with the management and operations of the REIT. The costs and expenses related to the retention of third parties shall be the sole cost and expense of the REIT except to the extent (i) the third party is retained to make decisions to invest in and dispose of Mortgage Assets, provide administrative, data processing or clerical services, prepare the financial records of the REIT or prepare a report summarizing the REIT's acquisitions of Mortgage Assets, portfolio compensation and c haracteristics, credit quality (if applicable) or performance of the portfolio, in each case with respect to assets the REIT has determined shall be managed by the Manager, in which case it shall be at the sole cost and expense of the Manager unless otherwise approved by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) or (ii) the costs and expenses are not reimbursable pursuant to Section 7.3 of this Agreement (collectively, the "Manager Obligations"). Notwithstanding anything in this Agreement to the contrary, in no event shall the Manager be responsible for any costs or expenses related to or incurred by any Other Manager.
- 2.4.3 <u>Affiliates.</u> Notwithstanding anything contained in this Agreement to the contrary, the Manager shall have the right to cause any of its services under this Agreement to be rendered by the Manager's employees or Affiliates of the Manager. The REIT shall pay or reimburse the Manager or its Affiliates (subject to the foregoing approval) for the reasonable and actually incurred cost and expense of performing such services by the Affiliate, including, without limitation, back office support services specifically requested by the REIT if the costs and expenses of such Affiliate would have been reimbursable under this Agreement if such Affiliate were an unaffiliated third party, or if such service had been performed by the Manager itself.
- 2.5 <u>Sub-Management Agreement</u>. The REIT and the Manager expressly acknowledge and agree that, concurrent with this Agreement, the Manager is entering into the Sub-Management Agreement, dated as of even date herewith, by and among the Manager, Staton Bell Blank Check LLC ("<u>Staton Bell</u>"), and Jeffrey J. Zimmer and Scott J. Ulm (Messrs. Zimmer and Ulm, together, the "<u>Manager Shareholders</u>") (such agreement, the "<u>Sub-Management Agreement</u>"), and

nothing to the contrary contained in this Agreement shall limit the ability of the Manager, Staton Bell, or the Manager Shareholders to enter into and perform their respective obligations under such Sub-Management Agreement or otherwise limit the effectiveness of such Sub-Management Agreement. The REIT represents and warrants that the Sub-Management Agreement has been duly authorized and approved by all necessary action of the REIT.

Additional Activities.

- Other Activities of the Manager. Nothing in this Agreement shall (i) prevent the Manager or its Affiliates, officers, directors or employees, from engaging in other businesses or from rendering services of any kind to any other person or entity, including, without limitation, investing in, or rendering advisory service to others investing in, any type of mortgage assets or other real estate investments (including, without limitation, investments that meet the principal investment objectives of the REIT), whether or not the investment objectives or policies of any such other person or entity are similar to those of the REIT, or (ii) in any way bind or restrict the Manager or its Affiliates, officers, directors or employees from buying, selling or trading any securities or commodities for their own accounts or for the account of others for whom the Manager or its Affiliates, officers, directors or employees may be acting. The REIT ackno wledges that the Manager will base allocation decisions on the procedures the Manager and the REIT reasonably and in good faith consider fair and equitable, including, without limitation, such considerations as investment objectives, restrictions and time horizon, availability of cash and the amount of existing holdings. While information and recommendations supplied to the REIT shall, in the Manager's reasonable and good faith judgment, be appropriate under the circumstances and in light of the investment objectives and policies of the REIT, they may be different from the information and recommendations supplied by the Manager or any Affiliate of the Manager to other investment companies, funds and advisory accounts. The REIT shall be entitled to equitable treatment under the circumstances in receiving information, recommendations and any other services. However, the REIT recognizes that it is not entitled to receive preferential treatment as compared with the treatment given by the Manager or any Affi liate of the Manager to any investment company, fund or advisory account other than any fund or advisory account which contains only funds invested by the Manager (and not of any of its clients or customers) or its officers and directors.
- 3.2 Other Activities of the REIT. Except to the extent expressly set forth in this Agreement or any other written agreement between the REIT and the Manager, neither this Agreement nor the relationship between the REIT and the Manager shall be deemed (i) to limit or restrict the activities of the REIT, its officers, its employees, or members of its Board of Directors (if and when any of the stock of the REIT becomes publicly traded), or (ii) impose a fee or other penalty on the REIT, its officers, its employees, or members of its Board of Directors (if and when any of the stock of the REIT becomes publicly traded) for pursuing any such other activities.
- 3.3 Service to the REIT; Execution of Documents Directors, officers, employees and agents of the Manager and its Affiliates may serve as trustees, directors, officers, employees, agents, nominees or signatories for the REIT or any subsidiary of the REIT, to the extent permitted by the Governing Instruments, as from time to time amended, or by any resolutions duly adopted by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) pursuant to the Governing Instruments. When executing documents or otherwise acting in such capacities for the REIT, such persons shall use their respective titles in the REIT.
- 4. Bank Accounts. The Manager may establish and maintain one or more bank accounts in the name of the REIT or any subsidiary of the REIT, and may collect and deposit into any such account or accounts, and disburse funds from any such account or accounts in a manner consistent with this Agreement, including, without limitation, the following: (a) the payment of the Base Management Fee, (b) the payment (or advance) of reimbursable costs and expenses, and (c) such other amounts. The Manager shall from time to time render appropriate accountings of such collections and payments to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) and, upon request (whether or not the REIT is publicly traded), to the auditors of the REIT or any subsidiary of the REIT. One or more of the obligations of the Manager hereunder may be revoked in whole or in part by the REIT from time to time in its sole discretion.
- 5. Records; Confidentiality. The Manager shall maintain appropriate and accurate books of account and records relating to services performed under this Agreement, and such books of account and records shall be accessible for inspection by representatives (including the auditors) of the REIT or any subsidiary of the REIT at any time during normal business hours. Except in the ordinary course of business of the REIT, the Manager shall, and shall use commercially reasonable efforts to cause each of its Affiliates to, keep confidential any and all information they (or such Affiliates) may obtain from time to time in connection with the services they (or such Affiliates) render under this Agreement.

- 6. Compensation of the Manager.
- Base Management Fee. For services rendered under this Agreement, commencing after the end of the first month of business, the REIT shall pay to the Manager each month in arrears (by wire transfer of immediately available funds) compensation equal to 1/12th of the sum of (a) 1.5% of the Gross Equity Raised up to \$1 billion plus (b) 0.75% of the Gross Equity Raised in excess of \$1 billion (the "Base Management Fee") within one (1) Business Day after the end of such month; provided, however, that the Base Management Fee shall not ever be less than 1/12th of the Annual Minimum Fee. In the event of a termination of this Agreement during a calendar month, the Base Management Fee shall be pro-rated based upon the number of days elapsed in such calendar month prior to the effective date of such termination. Notwithstanding the foregoing, for such period of time from the commencement of this Agreement until the date tha t Gross Equity Raised is \$50.0 million or more, the monthly Base Management Fee shall be equal to 1/12th of 1% of the Gross Equity Raised without regard to the Annual Minimum Fee.
 - 6.2 No Incentive Management Compensation. The Manager shall not receive any incentive-based compensation.
 - 7. Expenses of the Manager and the REIT.
 - 7.1 <u>Expenses of the Manager</u>. The Manager shall be responsible for the following expenses:
 - 7.1.1 employment expenses of the personnel employed by the Manager, including, without limitation, salaries (base and bonuses alike), wages, payroll taxes and the cost of employee benefit plans of such personnel (but excluding any stock of the REIT that the Board of Directors, if and when any of the stock of the REIT becomes publicly traded, may determine to grant to such personnel, which stock shall not reduce employment expenses otherwise payable by the Manager pursuant to this Section 7.1.1 or cause the Manager or the REIT to pay any payroll taxes in respect thereof); and
 - 7.1.2 rent, telephone, utilities, office furniture, equipment, machinery and other office, internal and overhead expenses of the Manager required for the REIT's day-to-day operations, including, bookkeeping, clerical and back-office services provided by the Manager, *provided*, *however*, that the REIT shall pay for supplies applicable to operations (paper, software, presentation materials, etc.).
- 7.2 Expenses of the REIT. The REIT shall pay all of the costs and expenses of the REIT and the Manager incurred solely on behalf of the REIT or any subsidiary or in connection with this Agreement, other than (i) those expenses that are specifically the responsibility of the Manager pursuant to Section 7.1 of this Agreement, and (ii) any costs or expenses incurred by the Manager which the REIT is not required to reimburse pursuant to the provisions of Section 7.3 below. Without limiting the generality of the foregoing, it is specifically agreed that the following costs and expenses of the REIT or any subsidiary of the REIT shall be paid by the REIT and shall not be paid by the Manager and/or the Affiliates of the Manager (except to the extent of any costs or expenses which the REIT is not required to reimburse pursuant to the provisions of Section 7.3 below):
 - 7.2.1 all costs and expenses associated with the formation and capital raising activities of the REIT and its subsidiaries, including, without limitation, the costs and expenses of the preparation of the REIT's registration statements, and any and all costs and expenses of any public offering of the REIT, any subsequent offerings and any filing fees and costs of being a public company, including, without limitation, filings with the Securities and Exchange Commission, the Financial Industry Regulatory Authority, the New York Stock Exchange (and any other exchange or over-the-counter market), among other such entities:
 - 7.2.2 all costs and expenses of the REIT in connection with the acquisition, disposition, financing, hedging, administration and ownership of the REIT's or any subsidiary's investment assets (including, without limitation, the Mortgage Assets) and, including, without limitation, costs and expenses incurred in contracting with third parties, including Affiliates of the Manager (as may be approved by the REIT pursuant to the terms of this Agreement), to provide such services, such as legal fees, accounting fees, consulting fees, trustee fees, appraisal fees, insurance premiums, commitment fees, brokerage fees, guaranty fees, ad valorem taxes, costs of foreclosure, maintenance, repair and improvement of property and premiums for insurance on property owned by the REIT or any subsidiary of the REIT;

- 7.2.3 all costs and expenses relating to the acquisition of, and maintenance and upgrades to, the REIT's portfolio analytics and accounting systems (including, but not limited to Bloomberg);
- 7.2.4 all costs and expenses of money borrowed by the REIT or its subsidiaries, including, without limitation, principal, interest and the costs associated with the establishment and maintenance of any credit facilities, warehouse loans and other indebtedness of the REIT and its subsidiaries (including commitment fees, legal fees, closing and other costs);
- 7.2.5 all taxes and license fees applicable to the REIT or any subsidiary of the REIT, including interest and penalties thereon;
- 7.2.6 all legal, audit, accounting, underwriting, brokerage, listing, filing, rating agency, registration and other fees, printing, engraving, clerical, personnel and other expenses and taxes of the REIT incurred in connection with the issuance, distribution, transfer, registration and stock exchange listing of the REIT's or any subsidiary's equity securities or debt securities;
- 7.2.7 other than for the Manager Obligations, all fees paid to and expenses of third-party advisors and independent contractors, consultants, managers and other agents (other than the Manager) engaged by the REIT or any subsidiary of the REIT or by the Manager for the account of the REIT or any subsidiary of the REIT (other than the Manager) and all employment expenses of the personnel employed by the REIT or any subsidiary of the REIT, including, without limitation, the salaries (base and bonuses alike), wages, equity based compensation of such personnel, and payroll taxes;
- 7.2.8 all insurance costs incurred by the REIT or any subsidiary of the REIT and including, but not limited to, insurance paid for by the REIT to insure the Manager for liabilities as a result of being the manager for the REIT;
 - 7.2.9 all custodian, transfer agent and registrar fees and charges incurred by the REIT;
- 7.2.10 all compensation and fees paid to directors of the REIT or any subsidiary of the REIT, all expenses of directors of the REIT or any subsidiary of the REIT (including those directors who are also employees of the Manager), the cost of directors and officers liability insurance and premiums for errors and omissions insurance, and any other insurance deemed necessary or advisable by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) for the benefit of the REIT and its directors and officers (including those directors who are also employees of the Manager), the cost of all meetings of the REIT's Board of Directors (if and when any of the stock of the REIT becomes publicly traded), and the cost of travel, hotel accommodations, food and entertainment for all participants in the meetings of the REIT's Board of Directors (if and when any of the stock of the REIT becomes publicly traded);
- 7.2.11 all third-party legal, accounting and auditing fees and expenses and other similar services relating to the REIT's or any subsidiary's operations (including, without limitation, all quarterly and annual audit or tax fees and expenses);
- 7.2.12 all legal, expert and other fees and expenses relating to any actions, proceedings, lawsuits, demands, causes of action and claims, whether actual or threatened, made by or against the REIT, or which the REIT is authorized or obligated to pay under applicable law or its Governing Instruments or by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded);
- 7.2.13 any judgment or settlement of pending or threatened proceedings (whether civil, criminal or otherwise) against the REIT or any subsidiary of the REIT, or against any trustee, director or officer of the REIT or any subsidiary of the REIT or any subsidiary of the REIT is required to indemnify such trustee, director or officer by any court or governmental agency, or settlement of pending or threatened proceedings;

- 7.2.14 at all times all travel and related expenses of directors, officers and employees of the REIT and the Manager incurred in connection with meetings related to the business of the REIT, attending meetings of the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) or holders of securities of the REIT or any subsidiary of the REIT or performing other business activities that relate to the REIT or any subsidiary of the REIT, including, without limitation, travel and expenses incurred in connection with the purchase, financing, refinancing, sale or other disposition of Mortgage Assets or other investments of the REIT; provided, however, that the REIT shall only be responsible for a proportionate share of such expenses, as reasonably determined by the Manager in good faith after full disclosure to the REIT, in instances in which such expenses were not incurred solely for the benefit of the REIT;
- 7.2.15 all expenses of organizing, modifying or dissolving the REIT or any subsidiary of the REIT, costs preparatory to entering into a business or activity, and costs of winding up or disposing of a business or activity of the REIT or its subsidiaries;
- 7.2.16 all expenses relating to payments of dividends or interest or distributions in cash or any other form made or caused to be made by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) to or on account of holders of the securities of the REIT or any subsidiary of the REIT, including, without limitation, in connection with any dividend reinvestment plan;
- 7.2.17 all expenses of third parties relating to communications to holders of equity securities or debt securities issued by the REIT or any subsidiary of the REIT and the other bookkeeping and clerical work necessary in maintaining relations with holders of such securities and in complying with the continuous reporting and other requirements of governmental bodies or agencies, including any costs of computer services in connection with this function, the cost of printing and mailing certificates for such securities and proxy solicitation materials and reports to holders of the REIT's or any subsidiary's securities and reports to third parties required under any indenture to which the REIT or any subsidiary of the REIT is a party;
- 7.2.18 subject to Section 7.1, all expenses relating to any office or office facilities maintained by the REIT or any subsidiary of the REIT (exclusive of the office of the Manager and/or Affiliates of the Manager), including, without limitation, rent, telephone, utilities, office furniture, equipment, machinery and other office expenses for the REIT's chief financial officer and any other persons the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) authorizes the REIT to hire;
- 7.2.19 all costs and expenses related to the design and maintenance of the REIT's web site or sites and associated with any computer software or hardware that is used solely for the REIT;
- 7.2.20 other than for the Manager Obligations, all other costs and expenses relating to the REIT's business and investment operations, including, without limitation, the costs and expenses of acquiring, owning, protecting, maintaining, developing and disposing of Mortgage Assets, including, without limitation, appraisal, reporting, audit and legal fees;
- 7.2.21 other than for the Manager Obligations, and subject to a line item budget approved in advance by the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), all other expenses actually incurred by the Manager, its Affiliates (as may be approved by the REIT pursuant to the terms of this Agreement) or their respective officers, employees, representatives or agents, or any Affiliates thereof (as may be approved by the REIT pursuant to the terms of this Agreement) which are reasonably necessary for the performance by the Manager of its duties and functions under this Agreement (including, without limitation, any fees or expenses relating to the REIT's compliance with all governmental and regulatory matters); and
- 7.2.22 all other expenses of the REIT or any subsidiary of the REIT that are not the responsibility of the Manager under Section 7.1 of this Agreement.
- 7.3 Expense Reimbursement to the Manager. Costs and expenses incurred by the Manager on behalf of the REIT or its subsidiaries shall be reimbursed in cash monthly to the Manager within five (5) Business Days of receipt by the REIT from the Manager of a statement of such costs and expenses. Cost and expense reimbursement to the Manager shall be subject to adjustment at the end of each calendar year in connection with the annual audit of the REIT.

8. Limits of Manager Responsibility: Indemnity.

- 8.1 <u>Limits of Manager Responsibility.</u> The Manager shall have the responsibility under this Agreement to render the services specifically called for under this Agreement and shall not be responsible for any action of the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) in following or declining to follow any advice or recommendations of the Manager, including, without limitation, as set forth in Section 2.2.2 of this Agreement. The Manager and its Affiliates, directors, officers, stockholders, equity holders, employees, representatives and agents, and any Affiliates thereof, shall not be liable to the REIT (including, without limitation, any stockholder thereof), any issuer of mortgage securities, any subsidiary of the REIT, its subsidiary's stockholders, the Board of Directors (if and when any of the stock of the REIT becomes publicly traded), any credit-party, any counter-party under any agreement to rany other person whatsoever for any acts or omissions, errors of judgment or mistakes of law by the Manager or its Affiliates, directors, officers, employees, representatives or agents, or any Affiliates thereof, under or in connection with this Agreement, except in the event that the Manager was grossly negligent, acted with reckless disregard or engaged in willful misconduct or fraud while discharging its duties under this Agreement.
- 8.2 Indemnification. The REIT and its subsidiaries shall reimburse, indemnify and hold harmless the Manager and its Affiliates, directors, officers, stockholders, equity holders, employees, representatives and agents, and any Affiliates thereof from and against any and all expenses, losses, costs, damages, liabilities, demands, charges and claims of any nature whatsoever, actual or threatened (including, without limitation, reasonable attorneys' fees), arising from or in respect of any acts or omissions, errors of judgment or mistakes of law) performed or made while acting in any capacity contemplated under this Agreement or pursuant to any underwriting agreement or similar agreement to which Manager is a party that is related to the REIT's activities. Notwithstanding the foregoing, the REIT shall have no indemnification obligation under this Section 8.2 in the event that the Manager was grossly negligent, acted with reckless disregard or engaged in willful misconduct or fraud while discharging its duties under this Agreement.
- 9. <u>No Joint Venture</u>. The REIT and the Manager are not partners or joint venturers with each other, and nothing in this Agreement shall be construed to make them such partners or joint venturers or impose any liability as such on any of them. The Manager is an independent contractor and, except as expressly provided or authorized in this Agreement, shall have no authority to act for or represent the REIT.

10. <u>Effectiveness; Termination</u>.

10.1 Effectiveness. This Agreement shall commence on the Effective Date and shall continue in effect thereafter for an initial term of five (5) years (the "Initial Term"). Following the Initial Term, this Agreement shall automatically extend for successive one (1)-year terms (each, a "Renewal Term"), unless either party gives 180 days' written notice prior to the expiration of the Initial Term or any Renewal Term to the respective other party of such first party's intent not to renew the then-current term (any such notice, a "Non-Renewal Notice"); provided, however, that if the REIT pays the Final Payment (as such term is defined in the Sub-Management Agreement) to Staton Bell pursuant to the terms of the Sub-Management Agreement, the then-current Renewal Term shall automatically be extended as necessary so that it expires one (1) year from the date on which such Fi nal Payment was made; provided, further, that the REIT may give a Non-Renewal Notice to the Manager only if, if and when any of the stock of the REIT becomes publicly traded, at least two-thirds of all of the Independent Directors or the holders of a majority of the outstanding shares of common stock of the REIT (other than those shares held by the Manager or its Affiliates) agree that (i) there has been unsatisfactory performance by the Manager that is materially detrimental to the REIT and its subsidiaries or (ii) the compensation payable to the Manager hereunder is unfair; provided further, however, that in the event that the REIT gives a Non-Renewal Notice to the Manager under clause (ii) above, such Non-Renewal Notice, and its effectiveness, shall be subject to Section 10.5. This Agreement may be terminated during the Initial Term or any Renewal Term only in accordance with the provisions of Sections 10.2, 10.3 and 10.4 or 13.1 (as applicable).

10.2 <u>Early Termination without Cause</u>.

10.2.1 The REIT may not terminate the Agreement during the Initial Term, except for Cause. After the Initial Term, the REIT may terminate the agreement without Cause upon 180 days' prior written notice to the Manager and subject to payment of the Termination Fee pursuant to Section 10.4 (except as otherwise provided in Section 13.1).

- 10.2.2 The Manager may terminate the agreement at any time and for any reason upon 180 days' prior written notice to the REIT.
- 10.3 <u>Early Termination for Cause.</u> Notwithstanding the provisions of Section 10.2.1, or any other provision of this Agreement to the contrary, the REIT may terminate the agreement for Cause at any time and without paying any Termination Fee, effective immediately upon written notice.

10.4 Payments In Connection With Termination.

- 10.4.1 Payments By the REIT. Following any termination of this Agreement by the REIT or the Manager, the REIT shall pay the following amounts to the Manager (by wire transfer of immediately available funds to such bank account as is designated by the Manager to the REIT in writing) not later than five (5) Business Days after the effective date of such termination:
 - (i) all reimbursable costs and expenses permitted under the Agreement (to the extent not previously reimbursed to the Manager), if any, as of the date of the effectiveness of such termination of this Agreement; and
 - (ii) either (a) if this Agreement was terminated by the REIT for Cause pursuant to Section 10.3, any Base Management Fee due and not yet paid to the Manager, (as pro-rated pursuant to Section 6.1 through the date of the effectiveness of such termination of this Agreement) or (b) if this Agreement was terminated by the REIT without Cause pursuant to Section 10.2.1, and subject to the provisions of Section 13.1, the Termination Fee (as calculated through the effective date of such termination of the Agreement).
- 10.4.2 <u>Payments By the Manager</u>. For the avoidance of doubt, following any termination of this Agreement by the Manager, no fees or other payment shall be due from the Manager to the REIT except as otherwise expressly provided in this Agreement.
- Renegotiation of Compensation. In the event that a Non-Renewal Notice is given by the REIT to the Manager in connection with a determination pursuant to clause (b)(ii) of Section 10.1 that the compensation payable to the Manager is unfair, the Manager shall have the right to renegotiate such compensation by delivering to the REIT, no fewer than 45 days prior to the prospective expiration of the Initial Term or Renewal Term then in effect, as applicable, written notice (any such notice, a "Notice of Proposal to Negotiate") of its intent to renegotiate its compensation under this Agreement. Thereupon, the REIT (represented by the Independent Directors, if and when any of the stock of the REIT becomes publicly traded) and the Manager shall endeavor to negotiate the revised compensation payable to the Manager under this Agreement. In the event that the Manager and the REIT, including, if and when any of the stock of the R EIT becomes publicly traded, at least two-thirds of all of the Independent Directors, agree to the terms of the revised compensation to be payable to the Manager within 45 days following the receipt of the Notice of Proposal to Negotiate, the Non-Renewal Notice shall be deemed of no force and effect and this Agreement shall continue in full force and effect on the terms stated in this Agreement, except that the compensation payable to the Manager hereunder shall be the revised compensation then agreed upon by the parties to this Agreement. The REIT and the Manager agree to execute and deliver an amendment to this Agreement setting forth such revised compensation promptly upon reaching an agreement regarding same. In the event that the REIT and the Manager are unable to agree to the terms of the revised compensation to be payable to the Manager during such 45-day period, this Agreement shall terminate, such termination to be effective on the expiration of the Initial Term or Renewal Term then in effect, as ap plicable.
 - 11. Action Upon Termination. In connection with any termination of this Agreement, the Manager shall promptly:
 - 11.1.1 pay over to the REIT or any subsidiary of the REIT all money collected and held for the account of the REIT or any subsidiary of the REIT by the Manager pursuant to this Agreement;
 - 11.1.2 deliver to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) an accounting, including a statement showing all payments collected by it and a statement of all money held by it, covering the period following the date of the last accounting furnished to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) with respect to the REIT or any subsidiary of the REIT;

- 11.1.3 deliver to the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) all property and documents of the REIT or any subsidiary of the REIT then in the custody of the Manager;
- 11.1.4 assign to the REIT any authorized agreements the Manager executed in its name on behalf of the REIT (and obtain the counter-parties' consent thereto); and
- 11.1.5 assign to the REIT all proprietary information with respect to the REIT, including, without limitation, software, models, intellectual property, licenses, tradenames and trademarks (but subject to the limitations set forth in Section 28 hereof).
- 12. <u>Survival of Obligations</u>. The REIT's obligation to make payments hereunder and the limitations set forth herein shall survive the termination of this Agreement. The covenants and agreements of the Manager contained herein (for expenses through the effective date of termination) shall survive the termination of this Agreement.

13. Assignments.

- 13.1 Assignment by the Manager. This Agreement shall terminate automatically in the event that the Manager assigns all or any part of this Agreement (including, without limitation, any transfer or assignment by operation of law), unless such assignment is consented to in advance in writing by the REIT, including, if and when any of the stock of the REIT becomes publicly traded, the Board of Directors. In the event an assignment by the Manager is consented to by the REIT, including, if and when any of the stock of the REIT becomes publicly traded, the Board of Directors in accordance with this Section 13.1, such assignment shall bind the assignee under this Agreement in the same manner as the Manager is bound, and the Manager shall be released from all of its obligations, duties and responsibilities under this Agreement and all liability therefore and in respect hereof accruing on or after that date. In addition, the assignee shall execute a nd deliver to the REIT a counterpart of this Agreement naming such assignee as Manager, and the REIT shall deliver to the assigning Manager a duly executed instrument evidencing the release of the assigning Manager from such obligations, duties and responsibilities as aforesaid. Notwithstanding the provisions of Section 10.2.1, or any other provision of this Agreement to the contrary, in the event that the REIT terminates this Agreement, whether for Cause or without Cause, following its assignment by the Manager to a successor Manager, the REIT shall not have any payment obligations to such successor Manager other than to pay unpaid reimbursable costs and expenses pursuant to Section 10.4.1(i) and earned but unpaid Base Management Fee payments pursuant to Section 10.4.1(ii)(a).
- 14. Release of Money or Other Property Upon Written Request. The Manager agrees that any money or other property of the REIT or any subsidiary of the REIT held by the Manager under this Agreement shall be held by the Manager as custodian for the REIT or such subsidiary, and the Manager's records shall be appropriately marked clearly to reflect the ownership of such money or other property by the REIT or such subsidiary.
- 14.1 <u>Procedures.</u> Upon the receipt by the Manager of a written request signed by a duly authorized officer of the REIT or an authorized member of the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) requesting the Manager to release to the REIT or any subsidiary of the REIT any money or other property then held by the Manager for the account of the REIT or any subsidiary of the REIT under this Agreement, the Manager shall release such money or other property to the REIT or such subsidiary of the REIT within a reasonable period of time, but in no event later than the earlier to occur of (i) thirty (30) days following such request, or (ii) the date of the termination of this Agreement.
- 14.2 <u>Limitations</u>. The Manager and its Affiliates, directors, officers, stockholders, equity holders, employees, representatives and agents, and any Affiliates thereof, shall not be liable to the REIT, any subsidiaries of the REIT, the Board of Directors (if and when any of the stock of the REIT becomes publicly traded) or the REIT's or its subsidiaries' stockholders for any acts performed or omissions to act by the REIT or any subsidiary of the REIT in connection with the money or other property released to the REIT or any subsidiary of the REIT in accordance with this Section 14, except in the event that the Manager was grossly negligent, acted with reckless disregard or engaged in willful misconduct or fraud while discharging its duties under this Agreement.
- 14.3 <u>Indemnification</u>. The REIT and any subsidiary of the REIT shall indemnify the Manager and its Affiliates, directors, officers, stockholders, equity holders, employees, representatives and agents, and any Affiliates thereof, against any and all expenses, costs, losses, damages, liabilities, demands, charges and claims of any nature whatsoever, which arise in connection with the Manager's release of such money or other property to the REIT or any subsidiary of the REIT in accordance with the terms of this Section 14, except in the event that the Manager was grossly negligent, acted with reckless disregard or engaged in willful misconduct or fraud while discharging its duties under this Agreement. Indemnification pursuant to this provision shall be in addition to any right of the Manager and its Affiliates, directors, officers, stockholders, equity holders, employees, representatives and agents, and any Affiliates thereof, to indemni fication under Section 8 of this Agreement.

- 15. Representations, Warranties and Covenants.
- 15.1 REIT in Favor of the Manager. The REIT hereby represents and warrants to the Manager as follows:
- 15.1.1 <u>Due Formation</u>. The REIT is duly organized, validly existing and in good standing under the laws of Maryland, has the power to own its assets and to transact the business in which it is now engaged and is duly qualified to do business and is in good standing under the laws of each jurisdiction where its ownership or lease of property or the conduct of its business requires such qualification, except for failures to be so qualified, authorized or licensed that could not in the aggregate have a material adverse effect on the business operations, assets or financial condition of the REIT and its subsidiaries, taken as a whole. The REIT does not do business under any fictitious business name.
- 15.1.2 Power and Authority. The REIT has the power and authority to execute, deliver and perform this Agreement and all obligations required under this Agreement and has taken all necessary action to authorize this Agreement on the terms and conditions hereof and the execution, delivery and performance of this Agreement and all obligations required under this Agreement. Except as shall have been obtained, no consent of any other person, including, without limitation, stockholders and creditors of the REIT, and no license, permit, approval or authorization of, exemption by, notice or report to, or registration, filing or declaration with, any governmental authority is required by the REIT in connection with this Agreement or the execution, delivery, performance, validity or enforceability of this Agreement and all obligations required under this Agreement. This Agreement has been, and each instrument or document required under this Agreement will be, executed and delivered by a duly authorized officer of the REIT, and this Agreement constitutes, and each instrument or document required under this Agreement when executed and delivered under this Agreement will constitute, the legally valid and binding obligation of the REIT enforceable against the REIT in accordance with its terms.
- 15.1.3 Execution, Delivery and Performance. The execution, delivery and performance of this Agreement and the documents or instruments required under this Agreement will not violate any provision of any existing law or regulation binding on the REIT, or any order, judgment, award or decree of any court, arbitrator or governmental authority binding on the REIT, or the Governing Instruments of, or any securities issued by, the REIT or of any mortgage, indenture, lease, contract or other agreement, instrument or undertaking to which the REIT is a party or by which the REIT or any of its assets may be bound, the violation of which would have a material adverse effect on the business operations, assets or financial condition of the REIT and its subsidiaries, taken as a whole, and will not result in, or require, the creation or imposition of any lien on any of its property, assets or revenues pursuant to the provisions of any su ch mortgage, indenture, lease, contract or other agreement, instrument or undertaking (other than the pledge of amounts payable to the Manager under this Agreement to secure the Manager's obligations to its lenders).
- 15.2 <u>Manager in Favor of the REIT</u>. The Manager hereby represents and warrants to the REIT as follows:
- 15.2.1 <u>Due Formation</u>. The Manager is duly organized, validly existing and in good standing under the laws of Delaware, has the power to own its assets and to transact the business in which it is now engaged and is duly qualified to do business and is in good standing under the laws of each jurisdiction where its ownership or lease of property or the conduct of its business requires such qualification, except for failures to be so qualified, authorized or licensed that could not in the aggregate have a material adverse effect on the business operations, assets or financial condition of the Manager and its subsidiaries, taken as a whole. The Manager does not do business under any fictitious business name.
- 15.2.2 Power and Authority. The Manager has the power and authority to execute, deliver and perform this Agreement and all obligations required under this Agreement and has taken all necessary corporate action to authorize this Agreement on the terms and conditions hereof and the execution, delivery and performance of this Agreement and all obligations required under this Agreement. Except as shall have been obtained, no consent of any other person including, without limitation, stockholders and creditors of the Manager, and no license, permit, approval or authorization of, exemption by, notice or report to, or registration, filing or declaration with, any governmental authority is required by the Manager in connection with this Agreement or the execution, delivery, performance, validity or enforceability of this Agreement and all obligations required under this Agreement. This Agreement has been and each instrument or doc ument required under this Agreement will be executed and delivered by a duly authorized officer of the Manager, and this Agreement constitutes, and each instrument or document required under this Agreement when executed and delivered under this Agreement will constitute, the legally valid and binding obligation of the Manager enforceable against the Manager in accordance with its terms.

- 15.2.3 Execution, Delivery and Performance. The execution, delivery and performance of this Agreement and the documents or instruments required under this Agreement will not violate any provision of any existing law or regulation binding on the Manager, or any order, judgment, award or decree of any court, arbitrator or governmental authority binding on the Manager, or the governing instruments of, or any securities issued by, the Manager or of any mortgage, indenture, lease, contract or other agreement, instrument or undertaking to which the Manager is a party or by which the Manager or any of its assets may be bound, the violation of which would have a material adverse effect on the business operations, assets or financial condition of the Manager and its subsidiaries, taken as a whole, and will not result in, or require, the creation or imposition of any lien on any of its property, assets or revenues pursuant to the provisions of any such mortgage indenture, lease, contract or other agreement, instrument or undertaking.
- 15.2.4 <u>No Limitations</u>. The personnel of the Manager providing services to the REIT on the Manager's behalf pursuant to this Agreement will be free of legal and contractual impediments to their provision of services pursuant to the terms of this Agreement.
- 16. Notices. Unless expressly provided otherwise in this Agreement, all notices, requests, demands and other communications required or permitted under this Agreement shall be in writing and shall be deemed to have been duly given, made and received when (1) delivered by hand, (2) otherwise delivered by reputable overnight courier against receipt therefor, or (3) upon actual receipt of registered or certified mail, postage prepaid, return receipt requested. The parties may deliver to each other notice by electronically transmitted facsimile copies or electronically transmitted mail (i.e., e-mail), provided that such facsimile or e-mail notice is followed within 24 hours by any type of notice otherwise provided for in this Section 16. Any party may alter the address or other contact information to which communications or copies are to be sent by giving notice of such change of address or other contact information in conformity with the provisions of this Section 16 for the giving of notice. Any notice shall be duly addressed to the parties as follows:

16.1 If to the REIT:

Jeffrey Zimmer ARMOUR Residential REIT, Inc. 3005 Hammock Way Vero Beach, FL 32963 Telecopy: (772) 388-4758 E-mail: jz@armourreit.com

with a copy given in the manner prescribed above, to:

Akerman Senterfitt One Southeast Third Avenue, 25th Floor SunTrust International Center Miami, FL 33131 Telecopy: (305) 374-5095

Attn.: Martin Burkett, Esq., and Bradley Houser, Esq.

E-mail: martin.burkett@akerman.com and bradley.houser@akerman.com

16.2 If to the Manager:

Jeffrey Zimmer
ARMOUR Residential Management, LLC
3005 Hammock Way
Vero Beach, FL 32963
Telecopy: (772) 388-4758
E-mail: jz@armourreit.com

with a copy given in the manner prescribed above, to:

Cahill Wink LLP

5 Penn Plaza, 23rd Floor New York, NY 10001 Telecopy: (518) 584-1962 Attn: David G. Nichols, Jr. PLLC E-mail: david.nichols@cahillwink.com

- 17. <u>Binding Nature of Agreement: Successors and Assigns</u> This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, personal representatives, successors and assigns as provided in this Agreement.
- 18. Entire Agreement. This Agreement and the Sub-Management Agreement contain the entire agreement and understanding among the parties hereto with respect to the subject matter of this Agreement and the Sub-Management Agreement, and supersede all prior and contemporaneous agreements, understandings, inducements and conditions, express or implied, oral or written, of any nature whatsoever with respect to the subject matter of this Agreement and the Sub-Management Agreement. The express terms of this Agreement and the Sub-Management Agreement control and supersede any course of performance and/or usage of the trade inconsistent with any of the terms of this Agreement or the Sub-Management Agreement. This Agreement may not be modified or amended other than in accordance with Section 27.
- 19. GOVERNING LAW. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF FLORIDA WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES TO THE CONTRARY.
- 20. Jurisdiction; Waiver of Jury Trial Any proceeding or action based upon, arising out of or related to this Agreement or the transactions contemplated hereby shall be brought in any state court of the State of Florida or, in the case of claims to which the federal courts have subject matter jurisdiction, any federal court of the United States of America, in either case, located in the State of Florida, and each of the parties irrevocably submits to the exclusive jurisdiction of each such court in any such proceeding or action, waives any objection it may now or hereafter have to personal jurisdiction, venue or to convenience of forum, agrees that all claims in respect of the proceeding or action shall be heard and determined only in any such court, and agrees not to bring any proceeding or action arising out of or relating to this Agreement or the transactions contemplated hereby in any other court. Nothing herein contained shall be de emed to affect the right of any party to serve process in any manner permitted by law or to commence legal proceedings or otherwise proceed against any other party in any other jurisdiction, in each case, to enforce judgments obtained in any action, suit or proceeding brought pursuant to this Section 20. EACH PARTY HERETO HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT THAT SUCH PARTY MAY HAVE TO A TRIAL BY JURY OF ANY CLAIM OR CAUSE OF ACTION DIRECTLY OR INDIRECTLY BASED UPON OR ARISING OUT OF THIS AGREEMENT.
- 21. No Waiver; Cumulative Remedies. No failure to exercise and no delay in exercising, on the part of any party hereto, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. Except as otherwise provided in this Agreement, the rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law. No waiver of any provision hereunder shall be effective unless it is in writing and is signed by the party asserted to have granted such waiver.
- 22. <u>Headings</u>. The headings of the sections of this Agreement have been inserted for convenience of reference only and shall not be deemed part of this Agreement.
- 23. <u>Counterparts.</u> This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original as against any party whose signature appears thereon, and all of which shall together constitute one and the same instrument. This Agreement shall become binding when one or more counterparts of this Agreement, individually or taken together, shall bear the signatures of all of the parties reflected hereon as the signatories.
- 24. <u>Severability.</u> Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 25. <u>Gender.</u> Words used herein regardless of the number and gender specifically used shall be deemed and construed to include any other number, singular or plural, and any other gender, masculine, feminine or neuter, as the context requires.

- 26. <u>Attorneys' Fees.</u> Should any action or other proceeding be necessary to enforce any of the provisions of this Agreement or the various transactions contemplated hereby, the prevailing party will be entitled to recover its actual reasonable attorneys' fees and expenses from the non-prevailing party.
- 27. Amendments. This Agreement may not be amended, modified or changed (in whole or in part), except by a formal, definitive written agreement expressly referring to this Agreement, which agreement is executed by all of the parties and, in the case of the REIT, if and when any of the stock of the REIT becomes publicly traded, approved by the Board of Directors. The parties hereto expressly acknowledge that no consent or approval of the REIT's stockholders is required in connection with any amendment, modification or change to this Agreement.
- 28. <u>Authority</u>. Each signatory to this Agreement warrants and represents that such signatory is authorized to sign this Agreement on behalf of and to bind the party on whose behalf such signatory is signing this Agreement.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first written above.

"REIT"

ARMOUR RESIDENTIAL REIT, INC.,

a Maryland corporation

By: <u>/s/ Jeffrey J. Zimmer</u> Name: Jeffrey J. Zimmer

Title: Chairman, CEO & President

"MANAGER"

ARMOUR RESIDENTIAL MANAGEMENT LLC,

a Delaware limited liability company

By: <u>/s/ Jeffrey J. Zimmer</u> Name: Jeffrey J. Zimmer Title: Managing Partner

[Signature page to Management Agreement]

ENTERPRISE ACQUISITION CORP. AND ARMOUR RESIDENTIAL REIT, INC. ANNOUNCE CLOSING OF MERGER TRANSACTION

BOCA RATON, Florida, November 6, 2009 – Enterprise Acquisition Corp. ("Enterprise") (NYSE Amex: EST; EST.U; EST.WS) and ARMOUR Residential REIT, Inc. ("ARMOUR") today announced the completion of their merger transaction. ARMOUR will immediately begin conducting business as a REIT, primarily investing in hybrid adjustable-rate, adjustable-rate and fixed-rate residential mortgage-backed securities issued or guaranteed by a U.S. Government-chartered entity.

"We are excited to announce the completion of this transaction," said Scott J. Ulm, Co-Chief Executive Officer, Chief Investment Officer, Head of Risk Management and Vice Chairman of ARMOUR. "On behalf of the stockholders of ARMOUR, we look forward to pursuing opportunities in the residential mortgage-backed securities markets."

ARMOUR projects a \$9.25 book value for each of the approximately 2,304,054 shares of common stock outstanding as of the close of the merger. ARMOUR is externally managed by ARMOUR Residential Management LLC ("ARRM"). ARRM's annual management fee will be 1% of initial gross equity until gross equity raised exceeds \$50 million; when the fee will revert to the schedule in ARRM's Management Agreement.

ARRM provides the day-to-day management of ARMOUR's operations and will perform services and activities relating to ARMOUR's assets and operations. Mr. Ulm and Jeffrey J. Zimmer, the Co-Chief Executive Officers of ARMOUR, are also principals of ARRM. Mr. Ulm has 23 years of structured finance and debt capital markets experience, including mortgage backed securities, and Mr. Zimmer has worked in the mortgage securities market for 25 years.

Ladenburg Thalmann acted as the primary financial advisor to Enterprise. Enterprise's counsel in the transaction was Akerman Senterfitt and ARMOUR's counsel was Cahill Wink LLP.

ARMOUR Residential REIT, Inc.

ARMOUR is Maryland corporation focused on investing in residential mortgage-backed securities. ARMOUR is externally managed and advised by ARRM. ARMOUR intends to elect and qualify to be taxed as a real estate investment trust ("REIT") for U.S. federal income tax purposes, commencing with ARMOUR's taxable year ending December 31, 2009.

Enterprise Acquisition Corp.

Enterprise (www.enterpriseacq.com) was a blank check company formed for the purpose of effecting a merger, capital stock exchange, asset acquisition or other similar business combination with one or more operating businesses. Upon completion of the transactions described above, Enterprise became a wholly-owned subsidiary of ARMOUR.

Safe Harbor

This press release includes "forward-looking statements" within the meaning of the safe harbor provisions of the United States Private Securities Litigation Reform Act of 1995. Actual results may differ from expectations, estimates and projections and, consequently, you should not rely on these forward looking statements as predictions of future events. Words such as "expect," "estimate," "project," "budget," "forecast," "anticipate," "intend," "plan," "may," "will," "could," "should," "believes," "predicts," "potential," "continue," and similar expressions are intended to identify such forward-looking statements. These forward-looking statements involve significant risks and uncertainties that could cause the actual results to differ materially from the expected results.

Additional information concerning these and other risk factors is contained in Enterprise's and ARMOUR's most recent filings with the Securities and Exchange Commission ("SEC"). All subsequent written and oral forward-looking statements concerning Enterprise and ARMOUR, the merger, the related transactions or other matters attributable to Enterprise and ARMOUR or any person acting on their behalf are expressly qualified in their entirety by the cautionary statements above. Enterprise and ARMOUR caution readers not to place undue reliance upon any forward-looking statements, which speak only as of the date made. Enterprise and ARMOUR do not undertake or accept any obligation or undertaking to release publicly any updates or revisions to any forward-looking statement to reflect any change in their expectations or any change in events, conditions or circumstances on which any such statement is based.

Additional Information and Where to Find It

Investors, security holders and other interested persons may find additional information regarding the companies at the SEC's Internet site at http://www.sec.gov/ or by directing requests to: ARMOUR Residential REIT, Inc., 3005 Hammock Way, Vero Beach, Florida 32963 Attention: Investor Relations.

Investor Contact:

Jeffrey Zimmer Co-Chief Executive Officer, President and Vice Chairman ARMOUR Residential REIT, Inc. (772) 388 9498